CWI Tracts

Managing Editors

K.R. Apt (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.M. Schumacher (CWI, Amsterdam) N.M. Temme (CWI, Amsterdam)

Executive Editor

M. Bakker (CWI Amsterdam, e-mail: Miente.Bakker@cwi.nl)

Editorial Board

W. Albers (Enschede)
M.S. Keane (Amsterdam)
J.K. Lenstra (Eindhoven)
P.W.H. Lemmens (Utrecht)
M. van der Put (Groningen)
A.J. van der Schaft (Enschede)
H.J. Sips (Delft, Amsterdam)
M.N. Spijker (Leiden)
H.C. Tijms (Amsterdam)

CWI

P.O. Box 94079, 1090 GB Amsterdam, The Netherlands Telephone + 31 - 20 592 9333 Telefax + 31 - 20 592 4199 URL http://www.cwi.nl

 $\mathsf{CWI} \text{ is the nationally funded } \mathsf{Dutch institute for research in Mathematics and Computer Science}.$

Probabilistic and analytical aspects of the umbral calculus

A. Di Bucchianico

1991 Mathematics Subject Classification: 05A40, 05A19, 33CXX, 41A36, 46J05, 60F05, 60E07, 60K05.

ISBN 90 6196 471 7 NUGI-code: 811

Copyright C1997, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

د دیگر معرف در چ انداز انداز انداز انداز انداز انداز انداز انداز ا

Acknowledgements

First of all I wish to thank Aart Stam for his excellent guidance during the time I did my research for this tract and for his continuing interest after that time. I also record with pleasure the cooperation¹ with Daniel Loeb and Gian-Carlo Rota which has led to interesting results on the Umbral Calculus, some of which can be found in this tract. Furthermore I wish to thank (in alphabetical order) several people who contributed one way or another to this tract: Piet Bruinsma, Philip Feinsilver, Robbert Fokkink, Mourad Ismail, Gérard Letac, Jan van Mill, Rense Posthumus, Henk de Snoo, Fred Steutel, and Erik Thomas. Last but certainly not least I am very grateful to my wife Francis for all those evenings that I spent with this tract instead with her.

Alessandro Di Bucchianico

Eindhoven, The Netherlands December 18, 1996

 $^{^1\}mathrm{This}$ cooperation was made possible by NATO grant CRG 930554.

Contents

	Ac	knowle	dgements	4			
1	Introduction						
2	Um	Umbral Calculus					
	2.1	A conv	volution equation	12			
	2.2	Basic I	polynomials and delta operators	19			
	2.3	Explici	it formulas for polynomials of convolution type	27			
	2.4	Sheffer	sequences	35			
	2.5	Cross s	sequences and Steffensen sequences	42			
3	Арр	Applications of the Umbral Calculus					
	3.1	Polyno	mials with $q_n(1) = c$ for $n \ge 1$.	50			
	3.2	Applic	ations to combinatorial identities	51			
	3.3	Discret	te probability distributions	55			
	3.4	Orthog	gonal Sheffer polynomials	60			
	3.5	Momen	nt systems	63			
	3.6	Shift-in	nvariant operators and integral operators	70			
	3.7	Expon	ential families	73			
		3.7.1	Natural exponential families	74			
		3.7.2	Natural exponential families and approximation				
			theory	77			
		3.7.3	Quadratic variance functions	78			
4	Ban	ach alg	gebras	81			
	4.1	Genera	al Banach algebra techniques	82			
	4.2	Algebr	as with contractible maximal ideal space	aximal ideal space 87			
		4.2.1	Algebras of summable sequences	88			
		4.2.2	Algebras of continuous functions	89			
		4.2.3	Algebras of holomorphic functions	91			
	4.3	Algebr	as on the unit circle	92			
	4.4	Applications to polynomials of convolution type					
	4.5	Two-si	ded sequences of functions of convolution type \ldots .	102			

5	Central limit theorems and infinite divisibility				
	5.1	Preliminaries	110		
	5.2	Asymptotics when g converges on its circle of convergence	112		
	5.3	Renewal theory	116		
	5.4	Logarithmic singularities	120		
	5.5	Infinitely divisible probability measures on $\mathbb N$ $\hfill\hfill$	126		
	Bibliography				
	Ind	ex	149		

Chapter 1 Introduction

This introduction consists of two parts. The first part is a historical introduction to the subject of the tract. The second part briefly describes the contents of the chapters of this tract (more elaborate descriptions of the chapters can be found in the introductions of the chapters). The aim of this tract is to study the probabilistic and analytic aspects of the Umbral Calculus. Therefore, the contents of this tract have little overlap with the existing books on Umbral Calculus ([60, 161, 163, 134, 209, 202]), which mainly stress the combinatorial and algebraic aspects of the Umbral Calculus. An interesting book with the same emphasis as the present tract is [91].

I have tried to make this tract as self-contained as possible. I have added Mathematical Reviews references to the items in bibliography at the end of this tract.

Historical introduction

There are quite a number of well-known sequences of polynomials, e.g. those attached to the names of Hermite, Legendre, Laguerre and many others. These sequences can be described in several ways. E.g., they can be described by generating functions, as solutions to differential equations, by orthogonality relations or by recurrence relations. The subject of this tract is a class of sequences of polynomials $(q_n)_{n \in \mathbb{N}}$ defined by the following functional equations

$$q_n(x+y) = \sum_{k=0}^n q_k(x) q_{n-k}(y) \qquad (n=0,1,\dots)$$
(1.1)

A sequence of polynomials that satisfies (1.1) is called a sequence of polynomials of convolution type. These sequences are closely related to the sequences of polynomials of binomial type introduced by Rota (see [162] and [210]), i.e.

sequences of polynomials $(p_n)_{n \in \mathbb{N}}$ satisfying

$$p_n(x+y) = \sum_{k=0}^n \binom{n}{k} p_k(x) p_{n-k}(y) \qquad (n=0,1,\dots)$$
(1.2)

The sequence $(x^n)_{n\in\mathbb{N}}$ is of binomial type by the Binomial Theorem, which explains the name binomial type. Obviously, $(q_n)_{n\in\mathbb{N}}$ is of convolution type if and only if $(n! q_n)_{n\in\mathbb{N}}$ is of binomial type. Thus these two types are essentially the same. I have chosen to work with sequences of polynomials of convolution type instead of sequences of polynomials of binomial type because convolution is a fundamental operation in analysis and probability theory. The binomial convolution appearing in (1.2) has advantages when dealing with certain combinatorial problems (see [162]).

An extension of the class of sequences of polynomials of binomial/convolution type is the class of Sheffer sequences $(s_n)_{n \in \mathbb{N}}$, whose convolution type version is defined by

$$s_n(x+y) = \sum_{k=0}^n s_k(x) q_{n-k}(y) \qquad (n=0,1,\dots)$$
(1.3)

for some fixed sequence $(q_n)_{n \in \mathbb{N}}$ of convolution type. The class of Sheffer sequences includes (amongst others) the Hermite, Bernoulli and Laguerre polynomials (more examples can be found in [29, 202, 235]).

The history of Sheffer sequences goes back to 1880 when Appell studied sequences $(a_n)_{n \in \mathbb{N}}$ of polynomials satisfying $Da_n = n a_{n-1}$ (D is the differentiation operator). Appell showed that these sequences satisfy

$$a_n(x+y) = \sum_{k=0}^n \binom{n}{k} a_k(x) y^{n-k} \qquad (n=0,1,\dots)$$
(1.4)

These sequences are called Appell sequences nowadays (see [32, Chapter 6], [72], [202, Chapter 4], [210, Section 13] or [215]). The Hermite polynomials form an Appell sequence.

The next major step was taken by Sheffer, whose work on difference equations led him in 1939 to generalize the Appell polynomials (see [217] or [72, p. 25]). Sheffer called his generalization polynomial sets of type zero; they are the Sheffer sequences defined by (1.3). The same class of polynomials was introduced in 1941 by Steffensen [227] (see also [220, 219, 227, 228, 229, 226]). There do exist even more general classes of polynomials such as Brenke sequences. These classes will not be considered in this tract; the interested reader is referred to the papers [8, 7, 20, 28, 29, 33, 36, 39, 40, 44, 49, 56, 57, 107, 108, 121, 122, 176]. A very elegant theory of Sheffer sequences is due to Rota and co-authors (see [162, 210]). These two papers are part of a series of papers on combinatorics, namely: [208, 69, 162, 109, 9, 81, 80, 210, 82, 30]. The Rota theory uses linear operators on the vector space of polynomials (cf. [209, Foreword]) and is therefore of a purely algebraic nature. It also provides a rigorous foundation for the Umbral Calculus (also called Blissard Calculus, see e.g. [23, 113, 185]). The systematic nature of the Rota theory easily yields numerous identities for special polynomials (see [202, Chapter 4] or [210]). The Rota theory of operators rests on earlier work by Pincherle, Steffensen, Toscano and Curry [70, 72, 180, 246]).

An extended and polished form of the Rota theory can be found in [202]. There have been several attempts to generalize the Rota theory. In particular, Roman has extended the Rota theory to include many sequences of polynomials that are not Sheffer sequences, e.g. Jacobi polynomials (see [201]) or q-polynomials (see [202, Section 6.4] or [203]). Roman remarks in [202, Section 6.1] that his ideas go back to 1936 ([246]), but he forgets to mention the work of Viskov (see [242]). Viskov has even an extension of the Rota theory including all sequences of polynomials (see [243, 244], cf. [139, 138, 154]). Cholewinski has adapted the Rota theory to Bessel functions (see [111, 112]). An application of operator calculus to hypergeometric functions can be found in [240].

For other generalizations of the Rota theory, see [13, 15, 16, 17, 18, 19, 34, 36, 37, 39, 40, 53, 54, 63, 51, 52, 103, 104, 119, 133, 136, 135, 149, 145, 147, 148, 146, 150, 158, 177, 178, 186, 197, 196, 204, 238, 239, 247, 248].

There is a wide range of applications of the Rota theory, e.g. statistics ([164, 168, 170]), combinatorics ([79, 114, 137, 162, 167, 169, 174, 173, 175, 189, 193, 194, 195, 198, 208, 223, 249, 250]), approximation theory ([123, 129, 241, 159, 219, 234]), recurrence relations ([41, 166, 165, 171, 206, 205]), physics ([25, 26, 89, 88, 91, 115, 116, 117, 252]), algebraic topology ([187, 188, 191, 190, 192]) and stochastic processes ([50, 222, 224]).

A survey of the Umbral Calculus with over 400 references can be obtained in electronic form through the Electronic Journal of Combinatorics:

http://ejc.math.gatech.edu:8080/Journal/Surveys/index.html

Contents of this tract

Chapter 2 is an introduction to the Rota Umbral Calculus. It is shown that if $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type, then $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ for some sequence $(g_n)_{n\in\mathbb{N}}$ with $g_0 = 0$. This aspect is only implicitly present in the work of Rota. However, it will be shown in Chapter 2 that the coefficient sequence $(g_n)_{n\in\mathbb{N}}$ is important for the theory of polynomials of convolution type. The section on cross sequences and Steffensen sequences brings together several results scattered in the literature.

Chapter 3 contains a miscellany of applications of the Umbral Calculus. Topics covered include finite probability distributions, combinatorial identities, exponential families, approximation operators, orthogonal polynomials, semigroups of probability measures, and integral representations of shift-invariant operators. These sections are partly based on [75, 77].

Chapter 4 starts with some general Banach algebra theory. This theory is used to obtain a new, unified treatment of existence problems for logarithms. This treatment is applied to polynomials of convolution type and yields analytic results on the generating function of sequences of polynomials of convolution type. Moreover, a two-sided analogue of polynomials of convolution type is introduced and studied. This section extends the results of [73].

The first sections of Chapter 5 consider central limit theorems for the coefficients of polynomials of convolution type as in [48, 55, 96, 97, 222, 224]. Results by Stam ([222, 224] are extended to the case of non-negative coefficients. The last section of Chapter 5 concerns an application of the theory of Chapter 4 to the theory of infinitely divisible probability measures on \mathbb{N} . It is shown that the Banach algebra techniques used in the literature (in particular, those by Chover, Ney and Wainger [61]) can be simplified considerably. Moreover, we give a simple proof of a result by Embrechts and Hawkes [87] on subexponential sequences.

Chapter 2

Umbral Calculus

This chapter is an introduction to Rota's Umbral Calculus as presented in [210]. For reasons explained in the introduction, we use polynomials of convolution type instead of binomial type.

Rota and his co-authors used operators together with formal power series in their papers [162, 210]. Although a rigorous foundation of formal power series exists (see e.g. [172]), we prefer to use operators only to develop the basic theory (see Section 2.2). More important, however, is our emphasis in this chapter on the coefficient sequence (Definition 2.1.11). The coefficient sequence is only implicitly present in [210]. Another feature of our approach is the use of elementary operator methods.

In Section 2.1 we study the system of convolution equations that defines the polynomials of convolution type. Sections 2.2 and 2.3 are an introduction to delta operators and polynomials of convolution type. The concept of polynomials of convolution type is generalized to Sheffer polynomials in Section 2.4 and to cross and Steffensen sequences in Section 2.5. Examples are included to illustrate the theory; systematic presentations of examples can be found in [202, Chapter 4] and [235].

Contents of Chapter 2

- 2.1 A convolution equation.
- 2.2 Basic polynomials and delta operators.
- 2.3 Explicit formulas for polynomials of convolution type.
- 2.4 Sheffer sequences.
- 2.5 Cross sequences and Steffensen sequences.

Notation and conventions

The degree of a polynomial, notation: $\deg p$, is defined as usual, however, the degree of a nonzero constant is defined to be zero and the degree of the zero polynomial is defined to be -1.

 \mathbb{N} is defined to be the set $\{0, 1, 2, \dots\}$.

The vector space of polynomials with coefficients in some fixed commutative ring, is denoted by \mathcal{P} . The commutative rings that we will use are the reals, the integers and the complex numbers.

2.1 A convolution equation

In this section we study the following system of equations:

$$f_n(x+y) = \sum_{k=0}^n f_k(x) f_{n-k}(y) \qquad (n=0,1,\ldots), \qquad (2.1)$$

where each f_n $(n \in \mathbb{N})$ is defined on a semigroup S and takes values in a commutative ring \mathcal{R} . This general setting enables us to prove the necessary results for all semigroups of interest to us (the reals, the positive reals, the natural numbers, etc.) at the same time.

These equations come up at several places:

- transition probabilities of stochastic processes ([126]): let $N(t)_{t\geq 0}$ be a stationary stochastic process with independent increments. If $f_n(x) = P(N(x) = n)$, then (2.1) follows by conditioning N(x + y) on N(x). For further references, see [1, pp. 111-116] and [3, Chapter 12])
- semigroups of convolution operators on sequence spaces: let $(T_t)_{t>0}$ be a semigroup of convolution operators on some Banach space X of onesided sequences. Then $(T_t x)_n = \sum_{k=0}^n f_k(t) x_{n-k}$ for some sequence of functions $(f_n)_{n\in\mathbb{N}}$ and (2.1) follows from the semigroup property.
- combinatorics: let $f_n(x)$ denote the number of functions with some specified property (e.g., injectivity) from an *n*-element set to an *x*-element set. Then (2.1) follows by partitioning an x + y-element set into two disjoint sets (see [162]).

All bounded solutions for $S = \mathcal{R} = (0, \infty)$ have been determined in [126], where (2.1) is related to transition probabilities of a stationary stochastic process with independent increments. It turns out that the solutions are given by so-called compound Poisson processes. A general study of the system of equations (2.1) was undertaken by Aczél and collaborators (see e.g. [2], [3, Chapter 12], [4]). We give a less general self-contained treatment which suffices for our purposes.

Definition 2.1.1 A sequence $(f_n)_{n \in \mathbb{N}}$ of functions on a semigroup S and taking values in a commutative ring \mathcal{R} is a sequence of functions of convolution type if $(f_n)_{n \in \mathbb{N}}$ satisfies (2.1) for all $n \in \mathbb{N}$ and all $x, y \in S$. If each f_n is a polynomial, then $(f_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type.

The interpretation of (2.1) in terms of transition probabilities of compound Poisson processes suggests that f_n must be of the form $e^{-\lambda x} \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$. Before we continue to determine the general solution of (2.1), we define the numbers g_n^{k*} and give some properties.

Definition 2.1.2 Let $\alpha = (\alpha_n)_{n \in \mathbb{N}}$ and $\beta = (\beta_n)_{n \in \mathbb{N}}$ be sequences in a commutative ring \mathcal{R} . The convolution $\alpha * \beta$ is the sequence defined by $(\alpha * \beta)_n := \sum_{k=0}^n \alpha_k \beta_{n-k}$.

If $k \in \mathbb{N}$, then α^{k*} is defined recursively as follows: $\alpha^{0*} := (\delta_{0n})_{n \in \mathbb{N}}$ (δ_{0n} is the Kronecker delta) and $\alpha^{(k+1)*} := \alpha^{k*} * \alpha$.

For sake of brevity, we will write α_n^{k*} instead of $(\alpha^{k*})_n$.

Remarks 2.1.3 Let $\alpha = (\alpha_n)_{n \in \mathbb{N}}$ be a sequence in a commutative ring \mathcal{R} . a) If $\sum_{n=0}^{\infty} \alpha_n z^n$ is a formal power series, then α_n^{k*} is the coefficient of z^n in $(\sum_{n=0}^{\infty} \alpha_n z^n)^k$. In other words,

$$\alpha_n^{k*} = \sum_{i_1 + \dots + i_k = n} \alpha_{i_1} \dots \alpha_{i_k}.$$

b) It follows directly from Definition 2.1.2 that $\alpha^{1*} = \alpha$ and $\alpha^{2*} = \alpha * \alpha$. c) Note that the convolution operation is commutative and associative. Associativity implies $\alpha^{i*} * \alpha^{j*} = \alpha^{(i+j)*}$ for all $i, j \in \mathbb{N}$. In particular, taking j = k - i, we obtain

$$\alpha_n^{k*} = \sum_{n=0}^{\infty} \alpha_m^{i*} \alpha_{n-m}^{(k-i)*}, \ (0 \le i \le k).$$
(2.2)

d) Let us prove the useful fact that α_n^{k*} is a polynomial in $\alpha_0, \ldots, \alpha_n$ for all $k \geq 1$ and all $n \in \mathbb{N}$. We proceed by induction on k. The statement holds for k = 1. Suppose by induction that the statement is true at k. Then $\alpha_n^{(k+1)*} = \sum_{n=0}^{\infty} \alpha_m \alpha_{n-m}^{k*}$ is a polynomial in $\alpha_0, \ldots, \alpha_n$. This completes the proof.

It follows that, if $\beta_0, \beta_1, \ldots, \beta_N$ is a finite sequence in \mathcal{R} , then β_n^{k*} is well-defined for $n \leq N$ and all $k \in \mathbb{N}$.

e) Formula (2.1) is a system of convolution equations, because it says that $(f_n(x+y))_{n\in\mathbb{N}}$ is the convolution of the sequences $(f_n(x))_{n\in\mathbb{N}}$ and $(f_n(y))_{n\in\mathbb{N}}$. If $(a_n)_{n\in\mathbb{N}}$ and $(b_n)_{n\in\mathbb{N}}$ are sequences of functions of convolution type, then $(f_n)_{n\in\mathbb{N}}$, defined by $f_n(x) := \sum_{k=0}^n a_k(x) b_{n-k}(x)$ for all $x \in S$, is also a sequence of functions of convolution type, since convolution is an associative and commutative operation.

Remark 2.1.4 If $(\alpha_n)_{n\in\mathbb{N}}$ is a sequence of non-negative real numbers such that $\sum_{n=0}^{\infty} \alpha_n = 1$, then α_n^{k*} has the following probabilistic interpretation. Let $X_i, i = 1, 2, \ldots$ be independent identically distributed random variables on some probability space $(\Omega, \mathcal{P}, \mathcal{F})$ with $P(X_i = n) = \alpha_n$ for $i = 1, 2, \ldots$ and $n \in \mathbb{N}$. Define $S_k := X_1 + \cdots + X_k$ $(k = 1, 2, \ldots)$. Using Remark 2.1.3a, it is easy to see that $P(S_k = n) = \alpha_n^{k*}$. Suppose $\alpha_0 = 0$. It follows from $P(S_k = k) = P(X_1 = X_2 = \ldots = X_k = 1)$ that $\alpha_k^{k*} = (\alpha_1)^k$. If k > n, then $P(S_k = n) = 0$, since $P(X_i \ge 1) = 1$ for $i = 1, 2, \ldots$. Hence, $\alpha_n^{k*} = 0$ if k > n.

Formula (2.2) can be interpreted as conditioning on S_i , i.e.

$$P(S_k = n) = \sum_{n=0}^{\infty} P(S_k = n \cap S_i = m) = \sum_{n=0}^{\infty} P(S_i = m) P(S_{k-i} = n - m).$$

The following lemma shows that the two properties mentioned in Remark 2.1.4 are also true under a more general condition.

Lemma 2.1.5 Let $(\alpha_n)_{n \in \mathbb{N}}$ be a sequence in a commutative ring such that $\alpha_0 = 0$. Then:

- a) $\alpha_n^{k*} = 0 \text{ if } k > n \ (k, n \in \mathbb{N}).$
- b) $\alpha_n^{n*} = (\alpha_1)^n$ for all $n \in \mathbb{N}$.
- c) α_n^{k*} is a polynomial in $\alpha_1, \ldots, \alpha_{n-1}$ for $2 \le k \le n$ $(k, n \in \mathbb{N})$.

Proof: a) We apply induction on k. The statement is true for k = 0. Suppose by induction that the statement is true at k. Then $\alpha_{n-m}^{k*} = 0$ for k+1 > nand $m \ge 1$. Hence, $\alpha_n^{(k+1)*} = \sum_{n=0}^{\infty} \alpha_m \alpha_{n-m}^{k*} = \alpha_0 \alpha_n^{k*} = 0$ since $\alpha_0 = 0$. b) We apply induction on n. The statement is true for n = 0, since $\alpha_0^{0*} = 1$ by definition. Suppose by induction that $\alpha_n^{n*} = (\alpha_1)^n$. It follows from a) and $\alpha_0 = 0$ that $\alpha_{n+1}^{(n+1)*} = \sum_{m=0}^{n+1} \alpha_m \alpha_{n+1-m}^{n*} = \alpha_1 \alpha_n^{n*} = (\alpha_1)^{n+1}$. c) We proceed by induction on k. The statement is true for k = 2, since $\alpha_n^{2*} = \sum_{n=0}^{n} \alpha_n \alpha_{n+1-m}^{n*} = \alpha_n \alpha_n^{n*} = (\alpha_n)^{n+1}$.

C) we proceed by induction on k. The statement is true for k = 2, since $\alpha_n = \sum_{i=0}^{n} \alpha_i \alpha_{n-i}$. Suppose by induction that the statement is true at k. Then Formula (2.2) yields $\alpha_n^{(k+1)*} = \sum_{n=0}^{\infty} \alpha_m \alpha_{n-m}^{k*}$, which equals $\sum_{m=1}^{n-k} \alpha_m \alpha_{n-m}^{k*}$ since $\alpha_0 = 0$ and $\alpha_{n-m}^{k*} = 0$ for m > n-k by a).

We are now ready to derive the general form of sequences of functions of convolution type (Definition 2.1).

Lemma 2.1.6 Let g_0, g_1, \ldots, g_N be elements of a commutative ring \mathcal{R} . Let h_0, h_1, \ldots, h_N be functions from a semigroup S to \mathcal{R} such that for fixed $x, y \in S$,

$$h_n(x+y) = \sum_{k=0}^n h_k(x) h_{n-k}(y)$$

for all n = 0, 1, ..., N. Define $f_n(x) := \sum_{k=0}^n g_n^{k*} h_k(x)$ (n = 0, 1, ..., N). Then

$$f_n(x+y) = \sum_{k=0}^n f_k(x) f_{n-k}(y)$$

for all $0 \le n \le N$.

Proof: This follows from direct substitution and the following form of (2.2):

$$\sum_{k=0}^{n} g_k^{i*} g_{n-k}^{j*} = g_n^{(i+j)*}.$$

Lemma 2.1.7 Let $(f_n)_{n \in \mathbb{N}}$ be a sequence of functions of convolution type from a semigroup S to a commutative ring \mathcal{R} .

- a) If $f_0 = 0$, then $f_n = 0$ for all $n \in \mathbb{N}$.
- b) If $S = (0, \infty)$ and $f_0(x) = 0$ for some x, then $f_n = 0$ for all $n \in \mathbb{N}$.

Proof: a) Follows directly by induction on n.

b) We apply induction on n. It follows immediately from $f_0(x+y) = f_0(x) f_0(y)$ that $f_0(t) = 0$ for all $t \ge x$ and that $f_0(x/2) = 0$. Iterating this argument yields $f_0(t) = 0$ for all t > 0. Suppose by induction that we proved that $f_m = 0$ for all m < n. Then $f_n(t) = f_n(\frac{1}{2}t + \frac{1}{2}t) = \sum_{k=0}^n f_k(\frac{1}{2}t) f_{n-k}(\frac{1}{2}t) = 0$ for all t > 0.

Theorem 2.1.8 Let $S \subset \mathbb{C}$ be a semigroup and let \mathcal{R} be a subset of \mathbb{C} that is closed under addition and a group with respect to multiplication. Suppose \mathcal{A} is an algebra of functions $S \to \mathcal{R}$ such that

- the only non-zero solutions in A to the equation f(x + y) = f(x) + f(y)are f(x) = cx with $c \in \mathcal{R}$.
- the only non-zero solutions in \mathcal{A} to the equation f(x+y) = f(x) f(y) are $f(x) = e^{ax}$ with $a \in \mathcal{R}$.

Then $(f_n)_{n \in \mathbb{N}}$ is a sequence of functions of convolution type in \mathcal{A} if and only if there exist $a \in \mathcal{R}$ and a sequence $(g_n)_{n \in \mathbb{N}}$ in \mathcal{R} with $g_0 = 0$, such that

$$f_n(x) = e^{ax} \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$$

for all $n \in \mathbb{N}$ and all $x \in S$.

Proof: ' \Leftarrow ' This follows from Lemma 2.1.6 with $h_n(x) = x^n/n!$. ' \Rightarrow ' If $f_0 = 0$, then $f_n = 0$ for all $n \in \mathbb{N}$ by Lemma 2.1.7a and the theorem holds with $g_n = 0$ for all $n \in \mathbb{N}$. We therefore assume that $f_0 \neq 0$.

Since $f_0(x+y) = f_0(x) f_0(y)$ for all $x, y \in S$ and $f_0 \in A$, there exists $a \in \mathcal{R}$ such that $f_0(x) = e^{ax}$. Define $(p_n)_{n \in \mathbb{N}}$ by $p_n(x) := e^{-ax} f_n(x)$. Note that $(p_n)_{n \in \mathbb{N}}$ is a sequence of functions of convolution type in \mathcal{A} too.

We now use induction on n in order to show that there exists a sequence $(g_n)_{n \in \mathbb{N}}$ in \mathcal{R} such that $p_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ for all $n \in \mathbb{N}$. Note that convolutions of finite sequences are well-defined by Remark 2.1.3d. If n = 0, then $p_0(x) = 1 = g_0^{0*} \frac{x^0}{0!}$ for all x. Since $p_1 \in \mathcal{A}$ and $p_1(x+y) = p_0(x) p_1(y) + p_1(x) p_0(y) = p_1(x) + p_1(y)$, it follows that $p_1(x) = p_1(1) x$. So

 $p_1(x) = g_1^{0*} \frac{x^0}{0!} + g_1^{1*} \frac{x^1}{1!} = g_1 x$, if g_1 is defined to be $p_1(1)$. Suppose that we have $g_0, g_1, \ldots, g_{n-1}$ (n > 1) in \mathcal{R} such that $g_0 = 0$ and

 $p_m(x) = \sum_{k=0}^m g_m^{k*} \frac{x^k}{k!}$ for m < n. It follows from (2.1) that p_n is a solution of the following linear functional equation in p:

$$p(x+y) - p(x) - p(y) = \sum_{k=1}^{n-1} p_k(x) p_{n-k}(y)$$

for all $x, y \in \mathcal{S}$.

It follows from Lemma 2.1.6 that p, defined by $p(x) := \sum_{k=0}^{n} g_n^{k*} \frac{x^k}{k!}$, is a solution in \mathcal{A} of this functional equation. Thus $(p_n - p)(x + y) = (p_n - p)(x) + (p_n - p)(x)$ $(p_n-p)(y)$ for all $x, y \in S$. Hence, there exists $c \in \mathcal{R}$ such that $(p_n-p)(x) = cx$ for all $x \in S$. Set $g_n := c$. Since g_n^{k*} $(1 < k \le n)$ can be expressed in terms of $g_1, g_2, \ldots, g_{n-1}$ by Remark 2.1.3d, we have $p_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$.

Theorem 2.1.9 The assumptions of Theorem 2.1.8 are satisfied in the following cases:

- $\mathcal{A} = measurable functions, S = \mathbb{N}, \mathbb{Z}, \mathbb{R}, (0, \infty) \text{ or } \mathbb{C}, and \mathcal{R} = \mathbb{R}, (0, \infty)$ or $\mathbb C$
- \mathcal{A} = locally bounded functions, $\mathcal{S} = \mathbb{N}, \mathbb{Z}, \mathbb{R}, (0, \infty)$ or \mathbb{C} , and \mathcal{R} = $\mathbb{R}, (0,\infty) \text{ or } \mathbb{C}$
- $\mathcal{A} = polynomials, S = \mathbb{N}, \mathbb{Z}, \mathbb{R}, (0, \infty) \text{ or } \mathbb{C}, and \mathcal{R} = \mathbb{R}, (0, \infty), or \mathbb{C}.$

Proof: It follows from [3, Remark below Theorem 4, p. 56], [3, Proposition 1, p. 53 and remark below Theorem 4, p. 56] and [3, Proposition 1, p. 53 and remark below Theorem 4,p. 56] that the assumptions of Theorem 2.1.8 are satisfied in these cases.

Remarks 2.1.10 a) Note that the proof of Theorem 2.1.8 shows that Theorem 2.1.8 also holds for finite sequences of functions of convolution type (cf. Remark 2.1.3d).

b) It follows from Theorem 2.1.8 that if $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type, then $g_n = q'_n(0)$. Thus $(q_n)_{n \in \mathbb{N}}$ determines $(g_n)_{n \in \mathbb{N}}$. Conversely, $(g_n)_{n \in \mathbb{N}}$ determines $(q_n)_{n \in \mathbb{N}}$ by Lemma 2.1.6. Hence, there is a one-to-one correspondence between sequences $(q_n)_{n\in\mathbb{N}}$ of polynomials of convolution type and sequences $(g_n)_{n \in \mathbb{N}}$ with $g_0 = 0$.

c) It follows from Theorem 2.1.8 that the convolution property of coefficients of polynomials of binomial type mentioned in [207, Proposition 4.3]) is nothing else than formula (2.2) in disguise.

d) It is possible to give a purely algebraic proof of Theorem 2.1.8 in case $(f_n)_{n \in \mathbb{N}}$ is a sequence of polynomials. The method of proof is similar to the idea employed in the proof of Theorem 2.1.12c.

We now present some general properties of sequences of polynomials of convolution type.

Definition 2.1.11 Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type. The coefficient sequence of $(q_n)_{n\in\mathbb{N}}$ is the sequence $(g_n)_{n\in\mathbb{N}}$ such that $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$.

The following theorem describes the interplay between a sequence of polynomials of convolution type and its coefficient sequence. Other results have been obtained by Niederhausen, see [164, 167].

Theorem 2.1.12 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n \in \mathbb{N}}$. Then:

- a) deg $q_n \leq n$ for all $n \in \mathbb{N}$.
- b) deg $q_n = n$ for all $n \in \mathbb{N}$ if and only if $g_1 \neq 0$.
- c) if $g_1 = 0$, then $\deg q_n \leq [n/2]$ for all $n \in \mathbb{N}$. In this case, $\deg q_n = [n/2]$ for all $n \in \mathbb{N}$ if and only if $g_2 \neq 0$ and $g_3 \neq 0$.
- d) the following formal generating function relation holds:

$$\sum_{n=0}^{\infty} q_n(x) t^n = e^{x g(t)}, \qquad (2.3)$$

where $g(t) = \sum_{n=0}^{\infty} g_n t^n$.

e)
$$q_0 = 1$$
 and $q_n(0) = 0$ for $n \ge 1$.

Proof: a) This follows directly from Theorem 2.1.8.

b) This follows directly from Theorem 2.1.8 and Lemma 2.1.5b.

c) We apply induction on n. It follows from Theorem 2.1.8 that the statement is true for n = 0 and n = 1.

Suppose by induction that $\deg q_m \leq [m/2]$ for all $m < n \ (n \geq 2)$. If $\deg q_n > [n/2]$, then $q_n(x) = \sum_{k=0}^{N} a_k x^k \ (N > [n/2], a_N \neq 0)$. Moreover, (2.1) yields

$$\sum_{k=0}^{N} a_k (2x)^k = q_n (2x) = \sum_{k=0}^{n} q_k (x) q_{n-k} (x).$$

Using the induction hypothesis and the inequality $\left[\frac{k}{2}\right] + \left[\frac{n-k}{2}\right] \leq \left[\frac{n}{2}\right] (0 \leq k \leq n)$, we see that the coefficient of x^N on the left-hand side equals $a_N 2^N$, whereas the coefficient of x^N on the right-hand side equals $2a_N$. This leads to N = 1 since $a_N \neq 0$, which contradicts $N > [n/2] \geq 1$. We conclude that deg $q_n \leq [n/2]$. This proves the first assertion.

For the second assertion, we note that the last line of Remark 2.1.3a easily yields $g_{2k}^{k*} = (g_2)^k$ and $g_{2k+1}^{k*} = k (g_2)^{k-1} g_3$, since $g_0 = g_1 = 0$. Hence, deg $q_n = [n/2]$ for all $n \in \mathbb{N}$ if and only if $g_2 \neq 0$ and $g_3 \neq 0$.

d) Using Theorem 2.1.8 and Lemma 2.1.5a we have

$$\sum_{n=0}^{\infty} q_n(x) t^n = \sum_{n=0}^{\infty} \sum_{k=0}^{\infty} g_n^{k*} \frac{x^k}{k!} t^n = \sum_{k=0}^{\infty} \left(\sum_{n=0}^{\infty} g_n^{k*} t^n \right) \frac{x^k}{k!} = e^{x g(t)}.$$

e) By definition, $g_0^{0*} = 1$. Thus Theorem 2.1.8 yields $q_0 = 1$. It follows from Lemma 2.1.5a that $g_n^{0*} = 0$ for $n \ge 1$. Hence, $q_n(0) = 0$ for $n \ge 1$ by Theorem 2.1.8.

Remarks 2.1.13

a) Theorem 2.1.12c yields the following extension of Lemma 2.1.5a: if $g_0 = g_1 = 0$, then $g_n^{k*} = 0$ for k > [n/2].

b) An example of a sequence of polynomials of convolution type with deg $q_n = [n/2]$ is the sequence of polynomials defined by the generating function

$$\sum_{n=0}^{\infty} q_n(x) \, z^n = e^{x (\log(1+z) - z)}.$$

These polynomials appear in combinatorics (see [199, p. 73]). It follows from Theorem 2.1.12c that deg $q_n = [n/2]$ for all $n \in \mathbb{N}$.

c) It is possible to extend Theorem 2.1.12c to the case $g_0 = \ldots = g_k = 0$. As an illustration, let us consider the following example due to Daniel Loeb. Fix $k \in \mathbb{N}$. Take $g_n = \delta_{nk}$ for all $n \in \mathbb{N}$. Then $q_n(x) = \frac{x^{n/k}}{n/k!}$ if k divides n, and 0 otherwise.

We conclude this section with an extension of a theorem due to Markowsky (see [154, Theorem 4.4]).

Theorem 2.1.14 Let $(\alpha_n)_{n \in \mathbb{N}}$ be a sequence in a commutative ring \mathcal{R} such that $\alpha_0 = 1$. Then for each sequence $(x_n)_{n \in \mathbb{N}}$ in $\mathcal{R} \setminus \{0\}$ there exists a unique sequence of polynomials of convolution type $(q_n)_{n \in \mathbb{N}}$ such that $q_n(x_n) = \alpha_n$ for all $n \in \mathbb{N}$.

Proof: Uniqueness is clear, since $q_n(x_n) = \alpha_n$ determines the values of $q_n(kx_n)$ for all $k \in \mathbb{N}$ (use (2.1)). Existence can be shown inductively as follows. By Theorem 2.1.8, it suffices to find a sequence $(g_n)_{n \in \mathbb{N}}$ in \mathcal{R} with $g_0 = 0$ such that $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ for all $n \in \mathbb{N}$. Suppose g_k has been found for each k < n. By Lemma 2.1.5c, g_n^{k*} is a polynomial in g_1, \ldots, g_{n-1} for $2 \le k \le n$. This means that we can choose g_n such that $\sum_{k=0}^n g_n^{k*} \frac{x_n^k}{k!} = \alpha_n$. This proves existence.

2.2 Basic polynomials and delta operators

In this section we link polynomials of convolution type to a certain kind of linear operators on polynomials, following and extending the exposition of the Umbral Calculus in [162, 210]. Our emphasis on operator methods yields simpler proofs than in [162] and [210]. The key theorems of this section are Theorems 2.2.15, 2.2.17 and 2.2.19 which describe the relation between delta operators and polynomials of convolution type.

Most proofs are new or modifications of existing proofs.

Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type with deg $q_n = n$ for all $n \in \mathbb{N}$. Then $(q_n)_{n\in\mathbb{N}}$ is a basis for the vector space \mathcal{P} of polynomials with coefficients in some field \mathcal{K} of characteristic zero. Hence there exists a unique linear operator Q on \mathcal{P} with $Qq_n = q_{n-1}$, $n \geq 1$ and $Qq_0 = 0$. The Rota Umbral Calculus is based on this operator. It turns out that this operator is shift-invariant (see Definition 2.2.2 and Theorem 2.2.19).

Definition 2.2.1 The shift-operator E^a is defined by $(E^a p)(x) := p(x + a)$ $(p \in \mathcal{P}).$

Definition 2.2.2 An operator T on \mathcal{P} is called shift-invariant if $E^aT = TE^a$ for all a.

Examples 2.2.3 Examples of shift-invariant operators include:

- a) the identity operator I.
- b) the differentiation operator D.
- c) the operators E^a of Definition 2.2.1.
- d) the forward difference operator $E^1 I$.
- e) the backward difference operator $I E^{-1}$.
- f) the Abel operators DE^a .
- g) the Laguerre operator L, defined by

$$(Lp)(x) := -\int_0^\infty e^{-t} p'(x+t) dt$$

h) the Bernoulli operator J, defined by

$$(Jp)(x) := \int_{x}^{x+1} p(t) dt$$

Remark 2.2.4 If S is an invertible shift-invariant operator on \mathcal{P} , then its inverse S^{-1} is also shift-invariant, since $S^{-1}E^a = S^{-1}E^aSS^{-1} = S^{-1}SE^aS^{-1} = E^aS^{-1}$ for all a.

Definition 2.2.5 A linear operator Q on \mathcal{P} is called a **delta operator** if Q is shift-invariant and Qx is a nonzero constant.

Examples 2.2.6 Examples of delta operators include b, d, e, f and g from Examples 2.2.3, but not a, c and h.

It is a remarkable fact that every linear shift-invariant operator has a Taylorlike expansion in terms of an arbitrary delta operator (see Theorem 2.2.22). We start by proving this expansion theorem for the differentiation operator D, because this yields simple proofs for properties of shift-invariant operators.

Theorem 2.2.7 ([210, Theorem 2]) Let D be the differentiation operator and define $q_n(x) = \frac{x^n}{n!}$ for all $n \in \mathbb{N}$. Then T is a linear shift-invariant operator on \mathcal{P} if and only if

$$T = \sum_{k=0}^{\infty} \left(Tq_k \right)(0) D^k.$$

Proof: ' \Leftarrow ' Note that the infinite sum is in fact a finite sum when applied to a polynomial and thus is a well-defined operator on \mathcal{P} . Shift-invariance of T follows from shift-invariance of D.

'⇒' Since $(q_n)_{n \in \mathbb{N}}$ is a basis for \mathcal{P} , it suffices to verify the result for Tq_n for all $n \in \mathbb{N}$. Using the Binomial Theorem, we obtain $(Tq_n)(a) = (E^aTq_n)(0) = (TE^aq_n)(0) = (\sum_{k=0}^n q_{n-k}(a)Tq_k)(0) = (\sum_{k=0}^\infty (Tq_k)(0)D^kq_n)(a)$ for all $n \in \mathbb{N}$ and all a.

Examples 2.2.8

a) Consider the shift-invariant operator E^a . Theorem 2.2.7 yields

$$E^a = \sum_{k=0}^\infty \, \frac{a^k}{k!} \, D^k = e^{aD}$$

Hence, $p(x + a) = (E^a p)(x) = \sum_{k=0}^{\infty} (D^k p)(x) \frac{a^k}{k!}$ for all $p \in \mathcal{P}$, which is Taylor's Formula.

b) Consider the Laguerre operator of Example 2.2.3e. Since for $k \ge 1$ we have

$$L\left(\frac{x^{k}}{k!}\right)(0) = -\int_{0}^{\infty} e^{-t} \frac{t^{k-1}}{(k-1)!} dt = -1,$$

it follows that $L = -\sum_{k=0}^{\infty} D^k = D(D-I)^{-1}$.

c) Consider the Bernoulli operator of Example 2.2.3f. Since

$$J\left(\frac{x^{k}}{k!}\right)(0) = \int_{0}^{1} \frac{t^{k}}{k!} dt = \frac{1}{(k+1)!},$$

it follows that $J = \sum_{k=0}^{\infty} \frac{D^k}{(k+1)!}$.

We now derive some corollaries from Theorem 2.2.7. The first corollary is an extension of [210, Propositions 1 and 2, p. 687]. Recall that the degree of a nonzero constant is defined to be zero and that the degree of the zero polynomial is defined to be -1.

Corollary 2.2.9 a) If T is a linear shift-invariant operator on \mathcal{P} , then there exists a non-negative integer n(T) such that deg $Tp = \max\{-1, \deg p - n(T)\}$ for all $p \in \mathcal{P}$. The null space of T equals the set of polynomials with degree less than n(T).

b) If Q is a delta operator, then $\deg Qp = \max\{-1, \deg(p) - 1\}$ and the null space of Q equals the set of constant polynomials.

Proof: a) By Theorem 2.2.7, we have $T = \sum_{k=0}^{\infty} a_k D^k$ for some sequence $(a_n)_{n \in \mathbb{N}}$. It follows from deg $D^k p = \max\{-1, \deg(p) - k\}$ that if we set $n(T) := \min\{k \in \mathbb{N} : a_k \neq 0\}$, then deg $Tp = \max\{-1, \deg(p) - n(T)\}$ for all $p \in \mathcal{P}$. Thus Tp = 0 if and only if deg p < n(T).

b) By definition, Qx is a nonzero constant. Thus a) implies that deg $Qp = \max\{-1, \deg(p) - 1\}$ for all polynomials $p \in \mathcal{P}$.

Remarks 2.2.10 a) The converse of Corollary 2.2.9a is not true. Fix $m \in \mathbb{N}$. We construct a linear, non shift-invariant operator T on \mathcal{P} such that deg $Tp = \max\{-1, \deg(p) - m\}$ for all $p \in \mathcal{P}$.

Define a linear operator T on \mathcal{P} by $Tx^k := 0$ if k < m, $Tx^m := 1$, $Tx^{m+1} := \frac{1}{2}x$ and $Tx^k := x^{k-m}$ if $k \ge m+2$. Clearly deg $Tp = \max\{-1, \deg(p) - m\}$ for all $p \in \mathcal{P}$. Then $(TE^1)x^{m+1} = T(x+1)^{m+1} = \sum_{k=0}^{m+1} {m+1 \choose k} Tx^k = T(x^{m+1} + (m+1)x^m) = \frac{1}{2}x + m + 1$ and $(E^1T)x^{m+1} = \frac{1}{2}(x+1)$. Thus $m = -\frac{1}{2}$, which is impossible since $m \in \mathbb{N}$. We conclude that T is not shiftinvariant.

For more information on the structure of linear shift-invariant operators on \mathcal{P} , see Remark 2.2.22.

b) Erik Thomas has pointed out to me that Corollary 2.2.9 can be used to find all translation-invariant linear subspaces of \mathcal{P} . A linear subspace L of \mathcal{P} is translation-invariant if $E^a L \subset L$ for all a. Translation-invariant linear subspaces are important in harmonic analysis.

The following result is somewhat stronger: if L is a linear subspace and $E^b L \subset L$ for some $b \neq 0$, then L is either one of the trivial subspaces $\{0\}$ or \mathcal{P} , or there exists $n \in \mathbb{N}$ such that $L = \mathcal{P}_n$, where \mathcal{P}_n is the set of all polynomials with degree not exceeding n. The proof runs as follows: let $p \in L$ be arbitrary and let m be the degree of p. Consider the linear shift-invariant operator $E^b - I$. Since $(E^b - I)x = b \neq 0$, it follows that $E^b - I$ is a delta operator. It follows from Corollary 2.2.9b that deg $((E^b - I)^k)p = m - k$ for $0 \leq k \leq m$. By linearity, $\mathcal{P}_m \subset L$. Suppose $L \neq \{0\}$ and define $n := \sup\{\deg p | p \in L\}$. The above argument yields that $L = \mathcal{P}_n$ if $n < \infty$ and that $L = \mathcal{P}$ if $n = \infty$. Conversely, \mathcal{P}_n is translation-invariant for each $n \in \mathbb{N}$.

Corollary 2.2.11 ([210, Corollary 1]) Let T be a linear shift-invariant operator on \mathcal{P} . Then the following are equivalent:

- a) T is invertible.
- b) $T1 \neq 0$.
- c) deg $p = \deg T p$ for all $p \in \mathcal{P}$.

Proof: 'a \Rightarrow b' The null space of an invertible linear operator consists of 0 only, so $T1 \neq 0$.

'b \Rightarrow c' Since $T1 \neq 0$, it follows from Corollary 2.2.9a that deg $p = \deg Tp$ for all $p \in \mathcal{P}$.

'c \Rightarrow a' It suffices to prove that T is injective and surjective. If $p, q \in \mathcal{P}$ and $p \neq q$, then $T(p-q) \neq 0$ since $\deg(p-q) \geq 0$. Moreover, $\deg p = \deg Tp$ implies that $(Tx^n)_{n \in N}$ is a basis for \mathcal{P} . Hence, T is surjective. \Box

Corollary 2.2.12 ([210, Corollary 4]) Any two linear shift-invariant operators on \mathcal{P} commute.

Proof: All linear shift-invariant operators can be represented as a formal power series in the differentiation operator D by Theorem 2.2.7. Since the action of these operators on a polynomial only involves finitely many terms of their expansions, the result follows.

The polynomials $q_n(x) = \frac{x^n}{n!}$ appeared in the proof of Theorem 2.2.7. These polynomials have the properties $q_0 = 1$, $Dq_n = q_{n-1}$ and $q_n(0) = 0$ for n > 0. Moreover, they are of convolution type by the Binomial Formula. We will now show that for every delta operator there exists a sequence of polynomials with analogous properties (see Theorems 2.2.15, 2.2.17 and 2.2.19).

Definition 2.2.13 Let Q be a delta operator. A sequence $(q_n)_{n \in \mathbb{N}}$ of polynomials is a basic sequence for Q if:

- 1. $q_0 = 1$
- 2. $q_n(0) = 0$ if $n \ge 1$
- 3. $Qq_n = q_{n-1}$ if $n \ge 1$.

Remarks 2.2.14 a) It follows from (1), (3) and Corollary 2.2.9b that deg $q_n = n$ for all $n \in \mathbb{N}$.

b) Note that properties (1) and (2) of Definition 2.2.13 are satisfied by each sequence $(q_n)_{n\in\mathbb{N}}$ of polynomials of convolution type by Theorem 2.1.12e. c) If $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type with deg $q_1 = 1$ and T is a linear operator on \mathcal{P} such that $Tq_n = q_{n-1}$ for $n \geq 1$, then T is shift-invariant since $TE^yq_n = T(\sum_{k=0}^n q_{n-k}(y)q_k) = \sum_{k=1}^n q_{n-k}(y)q_{k-1} = \sum_{h=0}^{n-1} q_{n-1-h}(y)q_h = E^yq_{n-1} = E^yTq_n$. Hence, by linearity, $TE^y = E^yT$.

Theorem 2.2.15 ([210, Proposition 3]) There is a unique basic sequence for every delta operator.

Proof: Let Q be an arbitrary delta operator. It follows from Theorem 2.2.7 and Corollary 2.2.9b that there exists a sequence $(\alpha_n)_{n\in\mathbb{N}}$ with $\alpha_1 \neq 0$ such that $Q = \sum_{k=1}^{\infty} \alpha_k D^k$. By Remark 2.2.14a, we must construct polynomials q_n of degree n. By (1) of Definition 2.2.13, $q_0 = 1$. Suppose by induction that $q_{n-1} = \sum_{k=0}^{n-1} a_{n-1,k} x^k$ has been constructed. Since deg $q_n = n$, q_n must be of the form $\sum_{k=0}^{n} a_{n,k} x^k$. Because $q_n(0) = 0$ by (3) of Definition 2.2.13, $a_{n,0}$ must be zero. Substitution of $Q = \sum_{k=1}^{\infty} \alpha_k D^k$ into $Qq_n = q_{n-1}$ and comparing coefficients yields the following system of equations:

$$a_{n-1,n-1} = \alpha_1 n a_{n,n}$$

$$a_{n-1,n-2} = \alpha_1 (n-1) a_{n,n-1} + \alpha_2 n (n-1) a_{n,n}$$

$$\vdots \qquad \vdots$$

$$a_{n-1,1} = \alpha_1 . 2 a_{n,2} + \alpha_2 . 2 . 3 a_{n,3} + \dots + \alpha_{n-1} n! a_{n,n}$$

Because $\alpha_1 \neq 0$ this system of equations has a unique solution. This proves uniqueness and existence.

Explicit formulas for the calculation of basic sequences will be discussed in Section 2.3.

Examples 2.2.16

a) The differentiation operator D has basic sequence $\left(\frac{x^n}{n!}\right)_{n\in\mathbb{N}}$.

b) The forward difference operator $E^1 - I$ has basic sequence $\binom{x}{n}_{n \in \mathbb{N}}$, where

$$\begin{pmatrix} x \\ n \end{pmatrix}$$
 := $rac{x(x-1)\dots(x-n+1)}{n!}$

are the lower factorials.

c) The backward difference operator $I - E^{-1}$ has basic sequence $\binom{x+n-1}{n}_{n \in \mathbb{N}}$, where

$$\binom{x+n-1}{n} := \frac{x(x+1)\dots(x+n-1)}{n!}$$

are the upper factorials .

d) The Abel operator DE^a has basic sequence $\left(\frac{x(x-na)^{n-1}}{n!}\right)_{n\in\mathbb{N}}$, the Abel polynomials.

Theorem 2.2.17 ([210, Theorem 1]) The basic sequence of a delta operator is a sequence of polynomials of convolution type.

Proof: Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. According to Definition 2.1.1 we have to prove

$$q_n(x+y) = \sum_{k=0}^n q_k(x) q_{n-k}(y)$$
(2.4)

for all $n \in \mathbb{N}$ and all x, y (cf. Remark 2.1.10b). We proceed by induction on n. The case n = 0 is trivial because $q_0 = 1$.

Suppose by induction that (2.4) has been proved for m < n. Fix y. It follows from Definition 2.2.5 that $QE^yq_n = E^yQq_n = E^yq_{n-1}$. Hence,

$$Q\left(E^{y}q_{n}-\sum_{j=0}^{n}q_{j}q_{n-j}(y)\right)=E^{y}q_{n-1}-\sum_{j=1}^{n}q_{j-1}q_{n-j}(y)=$$
$$E^{y}q_{n-1}-\sum_{k=0}^{n}q_{k}q_{n-1-k}(y)=0.$$

Corollary 2.2.9b implies that $E^y q_n - \sum_{k=0}^n q_k q_{n-k}(y)$ is a constant. So $q_n(x + y) = c + q_k(x) q_{n-k}(y)$. Taking x = 0 yields c = 0, since $q_n(0) = 1$ for $n \ge 1$. Because y was arbitrary, we obtain $q_n(x+y) = \sum_{k=0}^n q_k(x) q_{n-k}(y)$ for all x, y. \Box

Remark 2.2.18 Theorem 2.2.17 shows that the polynomials appearing in Examples 2.2.16 are of convolution type. This yields the following formulas:

a)

$$(x+y)^n = \sum_{k=0}^n \binom{n}{k} x^k y^{n-k}$$

(the well-known Binomial Formula).

b)

$$\binom{x+y}{n} = \sum_{k=0}^{n} \binom{x}{k} \binom{y}{n-k}$$

(this is the Vandermonde convolution formula, see e.g. [200, p. 8]).

c)

$$\binom{x+y+n-1}{n} = \sum_{k=0}^{n} \binom{x+n-1}{k} \binom{y+n-k-1}{n-k}$$

This formula is equivalent to the Vandermonde convolution formula, since $\binom{x+k-1}{k} = (-1)^k \binom{-x}{k}$.

d)

$$(x+y)(x+y-na)^{n-1} = \sum_{k=0}^{n} \binom{n}{k} x (x-ka)^{k-1} y (y-(n-k)a)^{n-k-1}$$

(this is the Abel generalization of the Binomial Formula, see e.g. [200, p. 18]).

The following theorem is a converse to Theorems 2.2.15 and 2.2.17.

Theorem 2.2.19 ([210, Theorem 1]) Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type such that deg $q_1 = 1$. Then there exists a unique delta operator Q with basic sequence $(q_n)_{n \in \mathbb{N}}$.

Proof: By Theorem 2.1.8, there exists a sequence $(g_n)_{n\in\mathbb{N}}$ such that $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ for all $n \in \mathbb{N}$. Thus $g_1 \neq 0$, because deg $q_1 = 1$ and $q_1(x) = g_1 x$. By Theorem 2.1.12b, deg $q_n = n$ for all $n \in \mathbb{N}$. Therefore $(q_n)_{n\in\mathbb{N}}$ is a basis for \mathcal{P} . Since $(q_n)_{n\in\mathbb{N}}$ is a basis for \mathcal{P} , there exists a unique linear operator Q on \mathcal{P} such that $Qq_n = q_{n-1}$ $(n \geq 1)$ and $Qq_0 = 0$. Since deg $q_1 = 1$, it follows that Qx is a nonzero constant. Shift-invariance of Q follows from Remark 2.2.14c. \Box

Remarks 2.2.20 a) It is essential in Theorem 2.2.19b that deg $q_1 = 1$. If deg $q_1 \neq 1$, then $q_1 = 0$ by Theorem 2.1.8 and no delta operator Q with $Qq_1 = q_0$ can exist by Corollary 2.2.9b since $q_0 = 1$.

b) Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type. As remarked in the beginning of this chapter, the Rota theory of polynomials of convolution type depends on the delta operator Q that maps q_n to q_{n-1} . Yang remarks in [253] that the linear operator on \mathcal{P} that maps q_n to q_{n+1} (the so-called Roman shift) is in some cases more useful than the delta operator. For more information on the Roman shift, see [202, Section 3.6].

We conclude this section with the general Expansion Theorem (cf. Theorem 2.2.7).

Theorem 2.2.21 (Polynomial Expansion Theorem) Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. Then

$$p = \sum_{k=0}^{\infty} \left(Q^k p \right)(0) \, q_k$$

for all $p \in \mathcal{P}$.

Proof: Let $p \in \mathcal{P}$ be arbitrary and let n be the degree of p. By Remark 2.2.14a, there exist constants c_k such that $p = \sum_{k=0}^{n} c_k q_k$. It follows that $Q^r p = \sum_{k=r}^{n} c_k q_{k-r}$ for $0 \leq r \leq n$. Evaluating at zero yields $c_r = (Q^r p)(0)$ since $q_k(0) = 0$ for $k \geq 1$. Hence, $p = \sum_{k=0}^{\infty} (Q^k p)(0) q_k$.

Theorem 2.2.22 (Operator Expansion Theorem, [210, Theorem 2]) Let T be a linear shift-invariant operator on \mathcal{P} and let Q be a delta operator with basic sequence $(q_n)_{n\in\mathbb{N}}$. Let $(g_n)_{n\in\mathbb{N}}$ be the coefficient sequence of $(q_n)_{n\in\mathbb{N}}$. Then:

$$a) T = \sum_{k=0}^{\infty} (Tq_k)(0)Q^k$$

b) In particular, if $(g_n)_{n \in \mathbb{N}}$ is the coefficient sequence of $(q_n)_{n \in \mathbb{N}}$, then $D = \sum_{n=0}^{\infty} g_n Q^n$ and $Q = \sum_{n=0}^{\infty} \bar{g}_n D^n$ where $\sum_{n=0}^{\infty} \bar{g}_n t^n$ is the composition inverse of the formal power series $\sum_{n=0}^{\infty} g_n t^n$.

Proof: a) Let $p \in \mathcal{P}$ be arbitrary with degree *n*. Applying Lemma 2.2.21 to $E^{y}p$, we obtain $TE^{y}p = \sum_{k=0}^{n} (Q^{k}E^{y}p)(0) Tq_{k} = \sum_{k=0}^{n} (Q^{k}p)(y) Tq_{k}$. Hence, $(Tp)(y) = (E^{y}Tp)(0) = (TE^{y}p)(0) = \sum_{k=0}^{n} (Tq_{k})(0) (Q^{k}p) (y) = \sum_{k=0}^{\infty} (Tq_{k})(0) (Q^{k}p) (y)$ for all y. This completes the proof, since p is arbitrary.

b) It follows from $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ that $(Dq_n)(0) = g_n$ for all $n \in \mathbb{N}$. Thus a) yields $D = \sum_{n=0}^{\infty} (Dq_n)(0)Q^n = \sum_{n=0}^{\infty} g_nQ^n$. Since $g_0 = 0$, the formal power series $\sum_{n=0}^{\infty} g_n t^n$ has a compositional inverse (see e.g. [172]).

Remarks 2.2.23 a) We implicitly used the Isomorphism Theorem 2.3.1 in the proof of Theorem 2.2.22b.

b) Fix an arbitrary delta operator Q with basic sequence $(q_n)_{n \in \mathbb{N}}$. We know from Remark 2.2.14a that $(q_n)_{n \in \mathbb{N}}$ is a basis for \mathcal{P} . Let T be an arbitrary linear shift-invariant operator on \mathcal{P} . Consider the infinite matrix $(a_{ij})_{i,j}$ with entries a_{ij} , where $Tq_j = \sum_{i=0}^{\infty} a_{ij} q_i$. Theorem 2.2.22a yields

$$Tq_j = \sum_{n=0}^{\infty} (Tq_n)(0)Q^n q_j = \sum_{n=0}^{j} (Tq_n)(0)q_{j-n} = \sum_{i=0}^{j} (Tq_{j-i})(0)q_i.$$

Hence, $a_{ij} = (Tq_{j-i})(0)$. Thus $a_{i,j} = a_{i+k,j+k}$ for all $k \in \mathbb{N}$, i.e. T is a Toeplitz operator on \mathcal{P} .

c) There also exists Operator Expansion Theorems for more general operators than shift-invariant operators. The coefficients of these expansions are polynomials in x rather than constants (see [76, 139]).

Examples 2.2.24 a) We want to expand the differentiation operator D in powers of the forward difference operator $E^1 - I$. The basic sequence of $E^1 - I$ is $\binom{x}{n}_{n \in \mathbb{N}}$, so

$$D = \sum_{k=0}^{\infty} \left(D\binom{x}{k} \right) (0) \left(E^1 - I \right)^k = \sum_{k=0}^{\infty} \frac{(-1)^{k-1}}{k} (E^1 - I)^k.$$

This is a classical formula for numerical differentiation.

b) Consider the shift operator E^a . Expanding E^a in powers of $E^1 - I$ yields

$$E^{a} = \sum_{k=0}^{\infty} {a \choose k} (E^{1} - I)^{k}.$$

This is Newton's forward difference interpolation formula.

We conclude this section with a few remarks on linear functionals. In [202, 207] the Umbral Calculus is presented in terms of linear functionals instead of linear operators as in this section. The following theorem describes the relationship between linear functionals and operators on \mathcal{P} .

Theorem 2.2.25 a) The map $p \to (Tp)(0)$ is a linear functional on \mathcal{P} for every linear operator T on \mathcal{P} .

b) If Λ is a linear functional on \mathcal{P} , then there exists a linear shift-invariant operator Q on \mathcal{P} such that $\Lambda p = (Qp)(0)$.

Proof: a) This follows directly from the linearity of T. b) Let Q be the shift-invariant operator defined by

$$Qp := \sum_{k=0}^{\infty} \Lambda\left(rac{x^k}{k!}
ight) D^k.$$

An easy calculation shows that $\Lambda x^n = (Q x^n)(0)$ for all $n \in \mathbb{N}$. Since the powers x^n span \mathcal{P} , this completes the proof.

2.3 Explicit formulas for polynomials of convolution type

In this section we derive some explicit formulas for polynomials of convolution type and we discuss the problem of connection coefficients.

Whereas we avoided the use of formal power series in Section 2.2, we can hardly do so in this section. The reason is that formal power series makes computations considerably easier.

We start with showing that the ring of formal power series and the ring of linear shift-invariant operators on \mathcal{P} are isomorphic.

Theorem 2.3.1 (Isomorphism Theorem = [210, Theorem 3]) Let Q be any delta operator on \mathcal{P} . The map Λ_Q defined by

$$\Lambda_Q\left(\sum_{k=0}^\infty a_k t^k\right) := \sum_{k=0}^\infty a_k Q^k$$

is an isomorphism between the ring of formal power series and the ring of linear shift-invariant operators on \mathcal{P} .

Proof: It is clear from Theorem 2.2.22a that Λ_Q is linear and injective. It follows from the Expansion Theorem 2.2.22a that Λ_Q is surjective. Hence, we only need to show that $\Lambda_Q(fg) = \Lambda_Q(f) \Lambda_Q(g)$ for all formal power series f and g. Let $f(t) = \sum_{k=0}^{\infty} a_k t^k$ and $g(t) = \sum_{k=0}^{\infty} b_k t^k$ be arbitrary formal power series. Let $(q_n)_{n\in\mathbb{N}}$ be the basic sequence of Q (Theorem 2.2.15). By Remark 2.2.14, $(q_n)_{n\in\mathbb{N}}$ is a basis for \mathcal{P} , hence it suffices to prove $\Lambda_Q(fg) q_n = \Lambda_Q(f) \Lambda_Q(g) q_n$ for all $n \in \mathbb{N}$. Since $(c_k Q^k) q_n = \sum_{k=0}^n c_k q_{n-k}$, we have $\Lambda_Q(fg) q_n = ((a*b)*q)_n$ and $\Lambda_Q(f) \Lambda_Q(g) q_n = (a*(b*q))_n$, where $a = (a_n)_{n\in\mathbb{N}}$, $b = (b_n)_{n\in\mathbb{N}}$ and $q = (q_n)_{n\in\mathbb{N}}$. Thus $\Lambda_Q(fg) q_n = \Lambda_Q(f) \Lambda_Q(g) q_n$ follows from associativity of the convolution operation (Remark 2.1.3c).

The proof of the next corollary shows once more that Theorem 2.3.1 is useful (cf. Remark 2.2.23a). Corollary 2.3.2 will be used in Section 3.3 for computing moments of discrete distributions.

Corollary 2.3.2 Let Q be a delta operator with basic sequence $(q_n)_{n\in\mathbb{N}}$ and let $(g_n)_{n\in\mathbb{N}}$ be the coefficient sequence of $(q_n)_{n\in\mathbb{N}}$. Let g be the formal power series defined by $g(t) := \sum_{n=0}^{\infty} g_n t^n$. Then $\sum_{k=0}^{\infty} k q_k(\alpha) Q^k = \alpha E^{\alpha} g'(Q) Q$.

Proof: Fix an arbitrary α . Using the formal generating function of Theorem 2.1.12d we obtain $\sum_{k=0}^{\infty} k q_k(\alpha) t^k = t \frac{d}{dt} \left(\sum_{k=0}^{\infty} q_k(\alpha) t^k \right) = t \frac{d}{dt} e^{\alpha} g(t) = \alpha e^{\alpha} g(t) g'(t) t$. Theorem 2.3.1 now yields

$$\sum_{k=0}^{\infty} k q_k(\alpha) Q^k = \alpha e^{\alpha g(Q)} g'(Q) Q = \alpha E^{\alpha} g'(Q) Q$$

(the last equality follows from Theorem 2.2.22b and Example 2.2.8a). \Box

We now present explicit formulas for basic sequences of delta operators. Formulas a) through d) of Theorem 2.3.6 were already known to Steffensen (see [227, Sections 2 and 3]; see also [210, Theorem 4]).

Definition 2.3.3 If T is a linear operator on \mathcal{P} , then its Pincherle derivative T' is defined by $T' := T \underline{\mathbf{x}} - \underline{\mathbf{x}} T$ where the linear operator $\underline{\mathbf{x}}$ is defined by $(\underline{\mathbf{x}} p)(x) := x p(x)$ for all x and all polynomials $p \in \mathcal{P}$.

The Pincherle derivative was introduced by Pincherle in [179, Section 56].

We now derive some elementary properties of the Pincherle derivative.

Lemma 2.3.4 a) If $T = \sum_{k=0}^{\infty} a_k D^k$, then $T' = \sum_{k=0}^{\infty} k a_k D^{k-1}$.

- b) The Pincherle derivative of a linear shift-invariant operator on \mathcal{P} is a linear shift-invariant operator on \mathcal{P} .
- c) The Pincherle derivative of a delta operator is an invertible shift-invariant operator on \mathcal{P} .
- d) If T and S are linear shift-invariant operators on \mathcal{P} , then (TS)' = T'S + TS'.

Proof: a) Since $\underline{\mathbf{x}}$ is a linear operator on \mathcal{P} , it suffices to prove a) for the polynomials $\frac{x^n}{n!}$. We have

$$T' \, \frac{x^n}{n!} := (T \, \underline{\mathbf{x}} - \underline{\mathbf{x}} \, T) \, \frac{x^n}{n!} = (n+1) \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^{n+1}}{(n+1)!} - x \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, a_k \, D^k \, \frac{x^n}{n!} = \frac{1}{2} \sum_{k=0}^{\infty} \, \frac{x^n}{n!} = \frac{1}$$

$$(n+1)\sum_{k=0}^{n+1} \frac{x^{n+1-k}}{(n+1-k)!} - \sum_{k=0}^{n} (n+1-k) a_k \frac{x^{n+1-k}}{(n+1-k)!} = \sum_{k=0}^{n+1} k a_k \frac{x^{n+1-k}}{(n+1-k)!} = \sum_{i=0}^{n} (i+1) a_{i+1} D^i \frac{x^n}{n!}.$$

Hence, $T' = \sum_{i=0}^{\infty} (i+1) a_{i+1} D^i$, since $\underline{\mathbf{x}}$ is a linear operator on \mathcal{P} . b) This follows directly from a) and Theorem 2.2.7.

c) By Theorem 2.2.7 and Corollary 2.2.9b we have $Q = \sum_{k=0}^{\infty} b_k D^k$ with $b_1 \neq 0$. We get from a) that $Q' = \sum_{i=0}^{\infty} (i+1) b_{i+1} D^i$. Hence, Q' is invertible by Corollary 2.2.11.

d) This follows from $(TS)' = TS \underline{\mathbf{x}} - \underline{\mathbf{x}}TS = (TS \underline{\mathbf{x}} - T\underline{\mathbf{x}}S) + (T\underline{\mathbf{x}}S - \underline{\mathbf{x}}TS) = TS' + T'S.$

Lemma 2.3.5 ([210, Proposition 4]) For every delta operator Q there exists a unique invertible shift-invariant operator U on \mathcal{P} such that Q = DU.

Proof: By Theorem 2.2.7 and Corollary 2.2.9b, we have $Q = \sum_{k=1}^{\infty} b_k D^k$ with $b_1 \neq 0$. Define U by $U := \sum_{k=0}^{\infty} b_{k+1} D^k$, so Q = DU. The invertibility of U follows from Corollary 2.2.11b, since $b_1 \neq 0$. Uniqueness of U follows from the expansion of Q and U in powers of D.

The operator U that appears in the statement of Theorem 2.3.6 is the operator whose existence is assured by Lemma 2.3.5.

Theorem 2.3.6 Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$ and let $(g_n)_{n \in \mathbb{N}}$ be the coefficient sequence of $(q_n)_{n \in \mathbb{N}}$. Let U be the unique invertible shift-invariant operator such that Q = DU. Then the following formulas hold for $n \geq 1$:

- a) $n! q_n = (Q' U^{-n-1}) (x^n)$
- b) $n! q_n = (U^{-n}) (x^n) (U^{-n})' (x^{n-1})$
- c) $n! q_n = (\underline{\mathbf{x}} U^{-n}) (x^{n-1})$
- d) $n q_n = (x(Q')^{-1}) q_{n-1}$ (Rodrigues Formula)
- e) $n q_n(x) = x \sum_{k=0}^n k g_k q_{n-k}(x)$.

Proof: Since D' = I, we have $Q' U^{-n-1} x^n = (D U)' U^{-n-1} x^n = ((D' U + DU') U^{-n-1}) x^n = ((U + DU') U^{-n-1}) x^n = (U^{-n} + DU' U^{-n-1}) x^n = U^{-n} x^n + U' U^{-n-1} D x^n = U^{-n} x^n - (U^{-n})' x^{n-1} = U^{-n} x^n - (U^{-n} \underline{\mathbf{x}} - \underline{\mathbf{x}} U^{-n}) x^{n-1} = (\underline{\mathbf{x}} U^{-n}) x^{n-1}$, so the right-hand sides of a), b) and c) are identical. Since Q has a unique basic sequence by Theorem 2.2.15, it suffices to note that $(\underline{\mathbf{x}} U^{-n} x^{n-1}) (0) = 0$ and $(Q Q' U^{-n-1}) \frac{x^n}{n!} = (D U Q' U^{-n-1}) \frac{x^n}{n!} = (Q' U^{-n}) \frac{x^{n-1}}{(n-1)!}$ for $n \ge 1$. This proves a), b) and c).

By Lemma 2.3.4c, Q' is invertible. Thus it follows from a) that $\frac{x^{n-1}}{(n-1)!} =$ $((Q')^{-1} U^n) q_{n-1}(x)$ for $n \ge 2$. By c),

$$nq_n(x) = \left(\underline{\mathbf{x}} U^{-n}\right) \frac{x^{n-1}}{(n-1)!} = \left(\underline{\mathbf{x}} U^{-n} (Q')^{-1} U^n\right) q_{n-1}(x) = \left(\underline{\mathbf{x}} (Q')^{-1}\right) q_{n-1}(x)$$

for $n \geq 2$. This proves d), since the case n = 1 follows from Lemma 2.3.4a and Theorem 2.2.22b.

In order to prove e) we write $n \frac{q_n(x)}{x} = \sum_{k=0}^{n-1} c_k q_k(x)$. Using d) and Lemma 2.2.21 we obtain

$$c_{k} = \left(Q^{k} n \frac{q_{n}(x)}{x}\right) (0) = \left(Q^{k} (Q')^{-1} q_{n-1}\right)(0) = ((Q')^{-1} q_{n-1-k})(0) = (n-k) \left(\frac{q_{n-1}(x)}{x}\right) (0) = (n-k) g_{n-k}.$$

is completes the proof.

This completes the proof.

Remark 2.3.7 It follows from the Rodrigues Formula that the Roman shift, i.e. the linear operator that takes q_n to q_{n+1} can be explicitly expressed as $(n+1) x (Q')^{-1}$ (cf. Remark 2.2.20b). The name Rodrigues Formula comes from the theory of orthogonal polynomials (see e.g. [58, 186]). An example of a classical Rodrigues Formula can be found in Example 2.3.8e.

Examples 2.3.8 We consider the delta operators of Examples 2.2.6 and use Theorem 2.3.6 to calculate the corresponding basic sequences (cf. Examples 2.2.16).

a) Consider the differentiation operator D. It is clear that D' = I and that U = I, since D = DI. Thus Theorem 2.3.6a yields $q_n(x) = \frac{x^n}{n!}$.

b) Consider the forward difference operator $E^1 - I$. Then $(E^1 - I)' = (E^1)' =$ $\left(e^{D}\right)'$ (use Theorem 2.2.22a) = e^{D} (use Lemma 2.3.4a) = E^{1} . Thus Theo-

rem 2.3.6d yields $q_n(x) = \frac{x}{n} E^{-1} q_{n-1}(x)$. Since $q_0 = 1$, induction on n yields $q_n(x) = \binom{x}{n} := \frac{x(x-1)\dots(x-n+1)}{n!}$.

(a) $(n) = (n)^{n!}$ (b) Consider the backward difference operator $I - E^{-1}$. In the same way as in (c) Consider the backward difference operator $I - E^{-1}$. In the same way as in (c) Consider the Abel operator DE^a for some fixed a. Obviously $U = E^a$, so (c) $U = E^a$, so

 $U^{-n} = E^{-na}$ for all $n \in \mathbb{N}$. Thus Theorem 2.3.6c yields $q_n(x) = \frac{x(x-na)^{n-1}}{n!}$.

e) Consider the Laguerre operator L of Example 2.2.3g. We will show that the basic sequence of the Laguerre operator is the sequence of Laguerre polynomials $L_n^{(-1)}$. We know from Example 2.2.8b that $L = -\sum_{k=0}^{\infty} D^k = D (D-I)^{-1}$, hence $U = (D-I)^{-1}$ in this case. Thus Theorem 2.3.6c yields $q_n(x) = \frac{x^n}{n!} (D-I)^n x^{n-1} = \sum_{k=1}^n (-1)^k {\binom{n-1}{k-1}} \frac{x^k}{k!}$. Since $e^x D \left(e^{-x} p\right) = C$ $e^{x}\left(e^{-x}p'-e^{-x}p\right)=\left(D-I
ight)\left(p
ight),$ we may write

$$q_n(x) = \frac{x^n}{n!} e^x D^n \left(e^{-x} x^{n-1} \right)$$

which is the classical Rodrigues formula for the Laguerre polynomials $L_n^{(-1)}$. The formula $Lq_n = q_{n-1}$ is the recurrence formula $q'_n = q'_{n-1} - q_{n-1}$, since $L = D(D-I)^{-1}$. Since $L' = -(D-I)^{-2}$, Theorem 2.3.6d yields $nq_n(x) =$ $-x(D-I)^2 q_{n-1}(x).$

We conclude this section with a discussion of umbral operators. Umbral operators play an important role in the connection-constant problem which will be discussed below.

Umbral operators were introduced by Rota to give a rigorous foundation to the so-called classical Umbral Calculus (also called Symbolic Calculus or Blissard Calculus). For more information on umbral operators, see [210, pp. 705-706], [102], [130], [202]. For more information on the classical Umbral Calculus we refer to [23, 113].

Definition 2.3.9 An umbral operator T is a linear operator on \mathcal{P} such that there exist basic sequences $(r_n)_{n \in \mathbb{N}}$ and $(v_n)_{n \in \mathbb{N}}$ with $T r_n = v_n$ for all $n \in \mathbb{N}$.

It is important to have basic sequences in Definition 2.3.9, since this implies deg $r_n = \deg v_n = n$ for all $n \in \mathbb{N}$ by Remark 2.2.14. Hence, both $(r_n)_{n \in \mathbb{N}}$ and $(v_n)_{n \in \mathbb{N}}$ are bases for \mathcal{P} .

Some important properties of umbral operators are listed in Theorem 2.3.11, which is an extension of [210, Proposition 1]). The following theorem is important for our proof of Theorem 2.3.11.

Theorem 2.3.10 Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$ and let the sequence $(p_n)_{n \in \mathbb{N}}$ be given by $p_n = \sum_{k=0}^n a_{n,k} q_k$. Then $(p_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type if and only if there exists a sequence $(\gamma_n)_{n \in \mathbb{N}}$ such that $\gamma_0 = 0$ and $a_{n,k} = \gamma_n^{k*}$ for all k and n. Moreover, $\gamma_n = (\gamma_n)_{n \in \mathbb{N}}$ such that $\gamma_0 = 0$ and $a_{n,k} = \gamma_n^{k*}$ for all k and n. $(Q p_n)(0).$

Proof: ' \Leftarrow ' This follows from Lemma 2.1.6.

' \Rightarrow ' We construct the sequence $(\gamma_n)_{n\in\mathbb{N}}$ by induction. Set $\gamma_0 = 0$. It follows from $p_1(0) = 0$ and Theorem 2.1.12b that either $p_1 = 0$ or deg $p_1 = \deg q_1 = 1$. Hence, there is a unique γ_1 such that $p_1 = \gamma_1 q_1$. Suppose by induction that γ_k has been constructed for k < n such that $p_m = \gamma_m^{k*} q_k$ for m < n. Since $\gamma_0 = 0$, Lemma 2.1.5c yields that γ_n^{k*} is a polynomial in $\gamma_1, \ldots, \gamma_{n-1}$ for $2 \leq k \leq n$. Thus we can choose γ_n such that $p_n(1) = \sum_{k=0}^n \gamma_n^{k*} q_k(1)$. It follows from Lemma 2.1.6 that $p_n(m) = \sum_{k=0}^n \gamma_n^{k*} q_k(m)$ for all $m \in \mathbb{N}$. Thus $p_n = \sum_{k=0}^n \gamma_n^{k*} q_n$, since p_n and $\sum_{k=0}^n \gamma_n^{k*} q_k$ are polynomials. The last statement follows from $Q q_n = q_{n-1}$ and $q_n(0) = 0$ for $n \geq 1$.

In [103], Garsia and Joni study equivalence classes whose elements are sequences of polynomials $(p_n)_{n\in\mathbb{N}}$ of the form $p_n = \sum_{k=0}^n \gamma_n^{k*} q_n$, where $(q_n)_{n\in\mathbb{N}}$ is an arbitrary fixed sequence of polynomials. Representations of the form $q_n = \sum_{k=0}^n \gamma_n^{k*} {x+k-1 \choose k}$ are used in Section 5.3 in the context of renewal theory. This idea is due to Stam ([222]). The following theorem describes the basic properties of umbral operators. As an introduction to parts d), e) and g), we let Q and P be delta operators with basic sequence $(q_n)_{n \in \mathbb{N}}$, $(p_n)_{n \in \mathbb{N}}$ respectively. Let T be the umbral operator that maps r_n to q_n for all $n \in \mathbb{N}$. This leads to the following commutative diagram.

$$\begin{array}{ccc} q_n & \stackrel{Q}{\longrightarrow} & q_{n-1} \\ \uparrow T^{-1} & & \downarrow T \\ p_n & \stackrel{P}{\longrightarrow} & p_{n-1} \end{array}$$

We immediately read off that $P = T Q T^{-1}$.

Theorem 2.3.11 Let T be an umbral operator. Then:

- a) T is invertible.
- b) T is shift-invariant if and only if T = I.
- c) If $(p_n)_{n \in \mathbb{N}}$ is an arbitrary sequence of polynomials of convolution type, then $(T p_n)_{n \in \mathbb{N}}$ is also of convolution type.
- d) If $(q_n)_{n \in \mathbb{N}}$ is the basic sequence of the delta operator Q, then $(T q_n)_{n \in \mathbb{N}}$ is the basic sequence of the delta operator $T Q T^{-1}$.
- e) If Q is a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$, then $TQ^nT^{-1} = P^n$, where P is the delta operator of the basic sequence $(Tq_n)_{n \in \mathbb{N}}$.
- f) The map $S \longrightarrow T S T^{-1}$ is an automorphism of the sequence of linear shift-invariant operators on \mathcal{P} .
- g) The map $Q \longrightarrow TQT^{-1}$ is an automorphism of the sequence of delta operators on \mathcal{P} .

Proof: Let $(r_n)_{n\in\mathbb{N}}$ and $(v_n)_{n\in\mathbb{N}}$ be basic sequences such that $T r_n = v_n$. Let R and V be the delta operators of $(r_n)_{n\in\mathbb{N}}$, $(v_n)_{n\in\mathbb{N}}$ respectively.

a) Since deg $r_n = \deg v_n = n$ for all $n \in \mathbb{N}$, T is invertible by Corollary 2.2.11. b) If T is shift-invariant, then Corollary 2.2.12 yields $R v_n = R T r_n = T R r_n = T r_{n-1} = v_{n-1}$ for $n \ge 1$. Hence, $r_n = v_n$ for all $n \in \mathbb{N}$, since both $(r_n)_{n \in \mathbb{N}}$ and $(v_n)_{n \in \mathbb{N}}$ are basic sequences for R.

c) By Theorem 2.3.10, there exists a sequence $(\gamma_n)_{n \in \mathbb{N}}$ such that $\gamma_0 = 0$ and $p_n = \sum_{k=0}^n \gamma_n^{k*} r_k$. Thus $T p_n = \sum_{k=0}^n \gamma_n^{k*} v_k$ and Theorem 2.3.10 implies that $(T p_n)_{n \in \mathbb{N}}$ is of convolution type.

d) We know from c) that $(Tq_n)_{n\in\mathbb{N}}$ is of convolution type. Since $q_1 = g_1 x$, we have deg $(Tq_1) = 1$. Thus $(Tq_n)_{n\in\mathbb{N}}$ is a basic sequence by Theorem 2.2.19b. Because $(TQT^{-1})(Tq_n) = Tq_{n-1}$ for $n \ge 1$, TQT^{-1} is a linear shift-invariant operator on \mathcal{P} by Theorem 2.2.19a. Moreover, $TQT^{-1}x$ is a nonzero constant, since deg $(Tq_1) = 1$. Hence, TQT^{-1} is a delta operator. e) This follows from d) and $TQT^{-1} = TQ^nT^{-1}$. f) Let S be an arbitrary linear shift-invariant operator on \mathcal{P} . By Theorem 2.2.22a, there exist constants a_n such that $S = \sum_{n=0}^{\infty} a_n D^n$. Then e) yields $TST^{-1} = \sum_{n=0}^{\infty} a_n D^n$. Hence, TST^{-1} is a linear shift-invariant operator on \mathcal{P} . Injectivity of $S \longrightarrow TST^{-1}$ follows from a). We now want to prove surjectivity. Let W be an arbitrary shift-invariant operator on \mathcal{P} . Since T^{-1} is also an umbral operator, it follows that $S := T^{-1}WT$ is a linear shift-invariant operator that satisfies $TST^{-1} = W$.

g) It follows from a) and d) that $Q \longrightarrow T Q T^{-1}$ is an injective homomorphism of the sequence of delta operators in itself. Surjectivity follows as in the proof of f).

For a probabilistic interpretation of umbral operators we refer to Section 3.5.

Now that we know how to calculate basic sequences, we are ready to discuss the problem of connection coefficients. The problem of connection coefficients consists of finding numbers $a_{n,k}$ such that $p_n = \sum_{k=0}^n a_{n,k} q_k$ where $(p_n)_{n \in \mathbb{N}}$ and $(q_n)_{n \in \mathbb{N}}$ are sequences of polynomials with deg $p_n = \deg q_n = n$ for all $n \in \mathbb{N}$. Note that the connection coefficients are the coefficients of the basis change $(p_n)_{n \in \mathbb{N}}$ to $(q_n)_{n \in \mathbb{N}}$ (cf. [94]).

If $(q_n)_{n \in \mathbb{N}}$ is a basic sequence and p is an arbitrary polynomial, then the connection coefficients can be calculated with Lemma 2.2.21.

Example 2.3.12 Consider the polynomials $\binom{nx}{n}$ which are not of convolution type (e.g, the convolution identity of Definition 2.1.1 is not satisfied for n = 2 and $y = \frac{1}{2}$). Since

$$\left((E^1-I)^k \binom{n\,x}{n}\right)(0) = \sum_{j=0}^k \binom{k}{j} (-1)^{k-j} \binom{n\,j}{n},$$

Lemma 2.2.21 yields the following expansion in terms of the basic polynomials $\binom{x}{n}$ of Example 2.2.16b:

$$\binom{n x}{n} = \sum_{k=0}^{n} \left[\sum_{j=0}^{k} \binom{k}{j} (-1)^{k-j} \binom{n j}{n} \right] \binom{x}{k}$$

If both $(p_n)_{n \in \mathbb{N}}$ and $(q_n)_{n \in \mathbb{N}}$ are sequences of polynomials of convolution type, then the Rota theory gives the following elegant answer (cf. [162, p. 202]).

Theorem 2.3.13 Let P and Q be delta operators with basic sequences $(p_n)_{n\in\mathbb{N}}$, $(q_n)_{n\in\mathbb{N}}$ respectively. Let T be the umbral operator defined by $Tq_n := \frac{x^n}{n!}$ for all $n \in \mathbb{N}$. Then the constants $a_{n,k}$ $(k, n \in \mathbb{N})$, defined by $p_n := \sum_{k=0}^n a_{n,k} q_k$, are uniquely determined as follows. The polynomials r_n , defined by $r_n(x) := \sum_{k=0}^n a_{n,k} \frac{x^k}{k!}$, are the basic polynomials of the delta operator TPT^{-1} . Moreover, if $P = \sum_{i=1}^{\infty} a_i Q^i$, then $TPT^{-1} = \sum_{i=1}^{\infty} a_i D^i$.

Proof: It follows from Theorem 2.3.11d that $T P T^{-1}$ is a delta operator with basic sequence $(T p_n)_{n \in \mathbb{N}}$. Since $r_n = T p_n$ for all $n \in \mathbb{N}$, $(r_n)_{n \in \mathbb{N}}$ is the basic sequence of $T P T^{-1}$. For the last statement, note that $T P T^{-1} = D$ by Theorem 2.3.11d, since $T q_n = \frac{x^n}{n!}$. The last statement now follows from Theorem 2.3.11e.

A different description of the connection coefficients is given in Theorem 2.3.10. Other descriptions of connection coefficients can be found in [99].

Examples 2.3.14 a) We want to express the lower factorials in terms of upper factorials of Example 2.2.16c, i.e. we want to calculate coefficients $a_{n,k}$ such that $\binom{x}{n} = \sum_{k=0}^{n} a_{n,k} \binom{x+k-1}{k}$. We apply Theorem 2.3.13 with $P = E^1 - I$, $Q = I - E^{-1}$ (of course, we could also apply Lemma 2.2.21). Let T be the umbral operator defined by $T\binom{x+n-1}{n} = \frac{x^n}{n!}$ for all $n \in \mathbb{N}$. Theorem 2.2.22a yields $P = \sum_{n=0}^{\infty} \left(P\binom{x+n-1}{n} \right) (0) Q^n = \sum_{n=1}^{\infty} Q^n$. Hence, it follows from Theorems 2.3.11d and 2.3.11e that

$$T P T^{-1} = \sum_{n=1}^{\infty} (T Q T^{-1})^n = \sum_{n=1}^{\infty} D^n = D (I - D)^{-1}.$$

Thus the coefficients $a_{n,k}$ are the coefficients of the polynomials $q_n(-x)$, where $(q_n)_{n \in \mathbb{N}}$ are the Laguerre polynomials of Example 2.3.8e.

Another relation between these polynomials is $\binom{x}{n} = (-1)^n \binom{-x+n-1}{n}$.

b) We want to derive duplication formulas for the Laguerre polynomials q_n of Example 2.3.8e. Fix α and define polynomials p_n by $p_n(x) := q_n(\alpha x)$ for all x. Let W be the umbral operator defined by $Wx^n := \alpha^n x^n$. Note that $Wq_n = p_n$. It follows from Theorem 2.3.11d that $(p_n)_{n\in\mathbb{N}}$ is the basic sequence of the delta operator P, defined by $P := WLW^{-1} = \alpha^{-1}D(\alpha^{-1}D-I)^{-1}$. Theorem 2.3.13 yields that the connection coefficients of $(p_n)_{n\in\mathbb{N}}$ and $(q_n)_{n\in\mathbb{N}}$ are the coefficients of the basic sequence of the delta operator TPT^{-1} , where T is the umbral operator defined by $Tq_n := \frac{x^n}{n!}$ for all $n \in \mathbb{N}$. By Theorem 2.2.22a,

$$D = \sum_{k=0}^{\infty} (D q_k(0)) L^k = \sum_{k=1}^{\infty} (-1)^k L^k = L (L-I)^{-1},$$

since $q_n(x) = \sum_{k=1}^n (-1)^k {\binom{n-1}{k-1}} \frac{x^k}{k!}$ (see Example 2.3.8e). Hence,

$$P = \alpha^{-1} L \left(I - (1 - \alpha^{-1}) L \right)^{-1}$$

and the last statement of Theorem 2.3.13 yields

$$T P T^{-1} = \alpha^{-1} \sum_{n=1}^{\infty} (1 - \alpha^{-1})^{n-1} D^n = D (\alpha I + (1 - \alpha) D)^{-1}.$$
It follows from Theorem 2.3.6c that the basic sequence $(r_n)_{n \in \mathbb{N}}$ of $T P T^{-1}$ is given by

$$\frac{x}{n} (\alpha I + (1-\alpha)D)^n \frac{x^{n-1}}{n-1!} = \frac{x}{n} \sum_{k=0}^n \binom{n}{k} \alpha^k (1-\alpha)^{n-k} D^{n-k} \frac{x^{n-1}}{n-1!} = \sum_{k=1}^n \binom{n}{k} \alpha^k (1-\alpha)^{n-k} \frac{x^{k-1}}{k-1!} = \sum_{k=1}^n \binom{n-1}{k-1} \alpha^k (1-\alpha)^{n-k} \frac{x^k}{k!}.$$

Putting everything together yields the following duplication formula for the Laguerre polynomials of Example 2.3.8e:

$$q_n(\alpha x) = \sum_{k=1}^n {\binom{n-1}{k-1}} \alpha^k (1-\alpha)^{n-k} q_k(x) \qquad (n \ge 1).$$

2.4 Sheffer sequences

Most properties of a basic sequence $(q_n)_{n \in \mathbb{N}}$ essentially depend only on the property $Q q_n = q_{n-1}$ (cf. Definition 2.2.13). Thus it seems plausible that the theory of basic sequences can be extended under weaker conditions. This is indeed the case, as the theory of Sheffer sequences shows (see [202, Chapter 2] or [210, Section 5]). In this section we will slightly generalize the Rota notion of Sheffer sequence.

Definition 2.4.1 Let Q be a delta operator. A sequence of polynomials $(s_n)_{n \in \mathbb{N}}$ is called a wide sense Sheffer sequence for Q if:

- 1. s_0 is constant
- 2. $Q s_n = s_{n-1}, n = 1, 2, \dots$

If moreover $s_0 \neq 0$, then $(s_n)_{n \in \mathbb{N}}$ is called a strict sense Sheffer sequence for Q.

The definition of Sheffer sequence in [210] is (apart from a factor n!) what we have called strict sense Sheffer sequence.

Note that if $(s_n)_{n \in \mathbb{N}}$ is a Sheffer sequence in the strict sense then, by Corollary 2.2.9b, deg $s_n = n$ for all $n \in \mathbb{N}$.

Theorem 2.4.2 Let Q be a delta operator. A sequence $(w_n)_{n \in \mathbb{N}}$, which is not identically zero, is a wide sense Sheffer sequence for Q if and only if there exist an $N \in \mathbb{N}$ and a strict sense Sheffer sequence $(s_n)_{n \in \mathbb{N}}$ for Q such that $w_n = 0$ for n < N and $w_n = s_{n-N}$ for $n \ge N$.

Proof: ' \Leftarrow ' Clearly w_0 is constant and $Q w_n = w_{n-1}$ for all $n \ge 1$. ' \Rightarrow ' If w_0 is a nonzero constant, then there is nothing to prove. Assume that $w_0 = 0$. Let $N := \min\{n : w_n \ne 0\}$. Then w_N is a (nonzero) constant by Corollary 2.2.9b since $Q w_N = w_{N-1} = 0$. Define $(s_n)_{n \in \mathbb{N}}$ by $s_n := w_{n+N}$. Then $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence for Q.

Theorem 2.4.2 seems to indicate that the notion of wide sense Sheffer sequence is not very useful. However, we will see nice applications of this notion in the proofs of Corollary 2.4.10 and Theorem 3.3.2.

It follows from Theorems 2.2.19 and 2.4.2 that a sequence of polynomials can be a Sheffer sequence of either type for at most one delta operator.

Examples 2.4.3 a) Strict sense Sheffer polynomials for the differentiation operator D are called Appell polynomials. They were studied by Appell in [10]. Examples of Appell polynomials include the Hermite polynomials H_n , defined by

$$\sum_{n=0}^{\infty} H_n(x) z^n = \exp \left(xz - \frac{1}{2}z^2\right),$$

and the Bernoulli polynomials B_n , defined by

$$\sum_{n=0}^{\infty} B_n(x) z^n = \frac{z}{e^z - 1} e^{x z}.$$

It follows directly from their generating functions or from Theorem 2.4.4d that these polynomials are Appell polynomials, i.e. $DH_n = H_{n-1}$ and $DB_n = B_{n-1}$ for $n \ge 1$. We will see at the end of this section that the Hermite and Bernoulli polynomials belong to the class of Wick polynomials, which is a subclass of the Appell polynomials.

b) The Laguerre polynomials of order α are strict sense Sheffer sequences for the Laguerre operator of Example 2.2.3g. The Laguerre polynomials of Example 2.3.8e are the Laguerre polynomials of order $\alpha = -1$ (cf. [202, p. 108]).

Both types of Sheffer sequences satisfy a convolution-like equation (see Theorem 2.4.4b below).

Theorem 2.4.4 Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. Then the following are equivalent:

- a) $(w_n)_{n \in \mathbb{N}}$ is a wide sense Sheffer sequence for Q.
- b) $w_n(x+y) = \sum_{k=0}^n w_k(x) q_{n-k}(y)$ for all $n \in \mathbb{N}$ and all x, y. c) $w_n = \sum_{k=0}^n w_k(0) q_{n-k}$ for all $n \in \mathbb{N}$.
- d) there exists a sequence $(a_n)_{n \in \mathbb{N}}$ such that $w_n = \sum_{k=0}^n a_k q_{n-k}$ for all $n \in \mathbb{N}$.

Proof: a \Rightarrow b' Fix an arbitrary x. Applying the Polynomial Expansion Theorem 2.2.21 to $E^x w_n$ we obtain

$$E^{x} w_{n} = \sum_{i=0}^{\infty} \left(Q^{i} E^{x} w_{n} \right) (0) q_{i} = \sum_{i=0}^{n} w_{n-i}(x) q_{i},$$

since deg $w_n \leq n$. Hence, it follows that

$$w_n(x+y) = \sum_{i=0}^n w_{n-i}(x) q_i(y) = \sum_{k=0}^n w_k(x) q_{n-k}(y)$$

for all x and y, since x is arbitrary.

'b \Rightarrow c' This follows by setting x = 0.

'c \Rightarrow d' Take $a_k := w_k(0)$.

'd \leftarrow a' Note that w_0 is constant because $w_0 = a_0 q_0 = a_0$. If $n \ge 1$, then $Q w_n = Q (\sum_{k=0}^n a_k q_{n-k}) = \sum_{k=0}^{n-1} a_k q_{n-1-k} = w_{n-1}$. Hence, $(w_n)_{n \in \mathbb{N}}$ is a wide sense Sheffer sequence for Q.

Remarks 2.4.5 a) Theorem 2.4.4 also holds for strict sense Sheffer sequences if we add the condition $w_0 \neq 0$ to b) and c) and if we add the condition $a_0 \neq 0$ to d).

b) If $(q_n)_{n \in \mathbb{N}}$ is merely a sequence of polynomials of convolution type instead of a basic sequence (cf. Remark 2.2.20a), then b), c) and d) of Theorem 2.4.4 are still equivalent and Corollary 2.4.6 below also holds.

Corollary 2.4.6 Let $(w_n)_{n \in \mathbb{N}}$ be a wide sense Sheffer sequence for the delta operator Q with basic sequence $(q_n)_{n \in \mathbb{N}}$. Let $(g_n)_{n \in \mathbb{N}}$ be the coefficient sequence of $(q_n)_{n \in \mathbb{N}}$. Then the following formal generating function identity holds:

$$\sum_{n=0}^{\infty} w_n(x) t^n = \left(\sum_{n=0}^{\infty} w_n(0) t^n\right) exp\left(x \sum_{n=0}^{\infty} g_n t^n\right).$$

Proof: This follows directly from Theorems 2.1.13d and 2.4.4c.

The following theorem describes the difference between wide sense and strict sense Sheffer sequences (of a delta operator Q with basic sequence $(q_n)_{n \in \mathbb{N}}$) in terms of the linear operator A on \mathcal{P} , defined by $Aq_n := s_n$. It follows directly from Theorem 2.2.22a that $A = \sum_{k=0}^{\infty} s_k(0) Q^k$ (cf. the proof of [210, Corollary 1]). We also give a description of strict sense Sheffer sequences in terms of delta operators and functionals in the style of [202, 207]. We first need a lemma.

Lemma 2.4.7 Let Λ be a linear functional such that $\Lambda 1 \neq 0$ and let Q be a delta operator on \mathcal{P} . There exists a unique sequence of polynomials $(p_n)_{n \in \mathbb{N}}$ with deg $p_n = n$ for all $n \in \mathbb{N}$ such that $\Lambda Q^k p_n = \delta_{nk}$ for all $k, n \in \mathbb{N}$, where δ_{nk} denotes the Kronecker delta.

Proof: Existence follows in the same way as in the proof of Theorem 2.2.15. In order to prove uniqueness, consider another sequence $(\tilde{p}_n)_{n\in\mathbb{N}}$ such that $\Lambda Q^k p_n = \Lambda Q^k \tilde{p}_n$ for all $k, n \in \mathbb{N}$. Suppose there is an $n \in \mathbb{N}$ such that $p_n \neq \tilde{p}_n$. Let ℓ be the degree of $p_n - \tilde{p}_n$. Then $Q^\ell(p_n - \tilde{p}_n)$ is a non-zero constant, which contradicts $\Lambda Q^k(p_n - \tilde{p}_n) = 0$.

Theorem 2.4.8 Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. Let $(s_n)_{n \in \mathbb{N}}$ be a sequence of polynomials and define the linear operator A on \mathcal{P} by $Aq_n := s_n$ for all $n \in \mathbb{N}$. Then:

- a) $(s_n)_{n \in \mathbb{N}}$ is a wide sense Sheffer sequence for Q if and only if A is shiftinvariant.
- b) ([210, Proposition 1]) $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence for Q if and only if A is shift-invariant and invertible.
- c) $(s_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence for Q if and only if there exists a linear functional Λ on \mathcal{P} such that $\Lambda 1 \neq 0$ and $\Lambda Q^k s_n = \delta_{nk}$ for all $k, n \in \mathbb{N}$, where δ_{nk} denotes the Kronecker delta.
- d) If $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence, then $\Lambda p = A^{-1}p(0)$ for all $p \in \mathcal{P}$, where A is as in a).

Proof: a) ' \Rightarrow ' Since $(q_n)_{n \in \mathbb{N}}$ is of convolution type, we have for all y

$$A E^{y} q_{n} = A \left(\sum_{k=0}^{n} q_{k}(y) q_{n-k} \right) = \sum_{k=0}^{n} q_{k}(y) s_{n-k} = E^{y} s_{n} = E^{y} A q_{n}.$$

Hence, by linearity, $A E^y = E^y A$ for all y.

' \Leftarrow ' Corollary 2.2.9a and $s_0 = A q_0 = A 1$ together imply that s_0 is constant. Using Corollary 2.2.12 we see that $Q s_n = Q A q_n = A Q q_n = A q_{n-1} = s_{n-1}$ for $n \ge 1$.

b) ' \Rightarrow ' Shift-invariance follows from a). By Corollary 2.2.9b, deg $s_n = n$ for all $n \in \mathbb{N}$. Hence, A is invertible by Corollary 2.2.11.

' \Leftarrow ' We need only prove that $s_0 \neq 0$ because of a). This follows from Corollary 2.2.11 and $s_0 = A q_0$, since A is invertible.

c) ' \Rightarrow ' Define the linear functional Λ by $\Lambda s_n = \delta_{0n}$. Because s_0 is a nonzero constant, we have $\Lambda 1 \neq 0$. Moreover, since shift-invariant operators commute by Corollary 2.2.12, it follows that $\Lambda Q^k s_n = \Lambda Q^k A q_n = \Lambda A Q^k q_n = \delta_{0,n-k} = \delta_{nk}$.

' \Leftarrow ' Define the polynomials r_n by $r_n := Q s_{n+1}$ $(n \in \mathbb{N})$. Then $\Lambda Q^k (Q s_{n+1}) = \delta_{k+1,n+1} = \delta_{k,n}$. By the uniqueness part of Lemma 2.4.7, we have $Q s_{n+1} = s_n$ for all $n \in \mathbb{N}$. Thus $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence.

It follows from $\Lambda Q^k s_n = \Lambda s_{n-k} = \delta_{nk}$ with k = 0 that $\Lambda s_n = \delta_{0n}$. Since $A^{-1} s_n(0) = q_n(0) = \delta_{0n}$ by Definition 2.2.13 and deg $s_n = n$ for all $n \in \mathbb{N}$, the results follows.

The operator A of the above theorem is called **invertible operator**.

Corollary 2.4.9 Let $(s_n)_{n\in\mathbb{N}}$ be a strict sense Sheffer sequence for the delta operator Q with basic sequence $(q_n)_{n\in\mathbb{N}}$ and invertible operator A. Let $(g_n)_{n\in\mathbb{N}}$ be the coefficient sequence of $(q_n)_{n\in\mathbb{N}}$ and let g be the formal power series defined by $g(t) := \sum_{n=0}^{\infty} g_n t^n$. Then the following formal generating function identity holds:

$$\sum_{n=0}^{\infty} s_n(x) t^n = f(g(t)) e^{x g(t)},$$

where A = f(D).

Proof: Define $s(t) := \sum_{n=0}^{\infty} s_n(x) t^n$. It follows from Theorem 2.2.22a that $A = \sum_{k=0}^{\infty} s_k(0) Q^k$. Hence, by the Isomorphism Theorem 2.3.1, we have A = s(Q). Since $g_0 = 0$, the formal power series is invertible (w.r.t. to composition, cf. [172]). Hence, there exists a formal power series f such that $s = f \circ g$. By Theorem 2.2.22b, we have A = f(g(Q)) = f(D). The result now follows from Corollary 2.4.6.

Corollary 2.4.10 Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$.

- a) The sequence $(s_n)_{n \in \mathbb{N}}$ defined by $s_n(x) := (n+1) \frac{q_{n+1}(x)}{x}$ $(x \neq 0)$ and $s_n(0) := (n+1) (q_{n+1})'(0)$ is a strict sense Sheffer sequence.
- b) The sequence $(w_n)_{n \in \mathbb{N}}$ defined by $w_n(x) := n \frac{q_n(x)}{x}$ $(x \neq 0)$ and $w_n(0) := n q'_n(0)$, is a wide sense Sheffer sequence.
- c) The sequence $(s_n)_{n \in \mathbb{N}}$ defined by $s_n := (q_{n+1})'$ is a strict sense Sheffer sequence.
- d) The sequence $(w_n)_{n\in\mathbb{N}}$ defined by $w_n(x) := q'_n$ is a wide sense Sheffer sequence.
- e) (Niederhausen) The sequence $(s_n)_{n \in \mathbb{N}}$ defined by

$$s_n(x) := \frac{x - an - b}{x - b} q_n(x - b)$$

is a strict sense Sheffer sequence.

Proof: a) Recall that $q_n(0) = 0$ for $n \ge 1$ by Theorem 2.1.12e). Then $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence by Theorem 2.4.8b, since $s_n = (Q')^{-1} q_n$ by Theorem 2.3.6d.

b) This follows from a) and Theorem 2.4.2.

d) It follows from Theorem 2.4.8a that $(w_n)_{n\in\mathbb{N}}$ is a wide sense Sheffer sequence. c) By Theorem 2.2.22b, $D q_{n+1} = \sum_{k=0}^{n+1} g_k q_{n-k}$. Thus $s_0 = g_1 \neq 0$ and $(s_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence by Theorem 2.4.4 or Remark 2.4.5. e) First note that $(E^{-b} q_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence by Theorem 2.4.8b. By b) and Theorem 2.4.8a, $\left(n\frac{q_n(x-b)}{x-b}\right)_{n\in\mathbb{N}}$ is a wide sense Sheffer sequence. Since linear combinations of wide sense Sheffer sequences are wide sense Sheffer, the decomposition

$$\frac{x-an-b}{x-b}q_n(x-b) = q_n(x-b) - \frac{an}{x-b}q_n(x-b)$$

shows that $(s_n)_{n\in\mathbb{N}}$ is a wide sense Sheffer sequence. A closer look at the decomposition reveals that deg $s_n = n$, thus $(s_n)_{n\in\mathbb{N}}$ is even a strict sense Sheffer sequence.

We now extend the Expansion Theorems 2.2.21 and 2.2.22 to strict sense Sheffer sequences.

Theorem 2.4.11 Let $(s_n)_{n \in \mathbb{N}}$ be a strict sense Sheffer sequence with delta operator Q and let A be the linear operator on \mathcal{P} defined by $Aq_n := s_n$.

a) For all $p \in \mathcal{P}$, we have

$$p = \sum_{k=0}^{\infty} \left(A^{-1} \, Q^k p \right)(0) \, s_k.$$

b) If T is a linear shift-invariant operator, then

$$T = \sum_{k=0}^{\infty} \left(Ts_k(0) \right) A^{-1} Q^k$$

Proof: a) Apply Theorem 2.2.21 to $p = A(A^{-1}p)$ and use shift-invariance. b) Apply Theorem 2.2.22 to $T = A^{-1}(AT)$ and use shift-invariance.

Theorem 2.4.8 enables us to generalize Theorem 2.3.10 to strict sense Sheffer sequences. A generalization to wide sense Sheffer sequences is not possible (see Remark 2.4.13).

Theorem 2.4.12 Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. Let $(s_n)_{n \in \mathbb{N}}$ be a strict sense Sheffer sequence for Q and let A be the linear operator on \mathcal{P} defined by $Aq_n := s_n$. The following are equivalent for a sequence $(r_n)_{n \in \mathbb{N}}$ of polynomials:

- a) $(r_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence and there exists a basic sequence $(p_n)_{n\in\mathbb{N}}$ such that $r_n = A p_n$ for all $n \in \mathbb{N}$.
- b) there exists a sequence $(\gamma_n)_{n \in \mathbb{N}}$ with $\gamma_0 = 0$ and $\gamma_1 \neq 0$ such that $r_n = \sum_{k=0}^n \gamma_n^{k*} s_k$ for all $n \in \mathbb{N}$.

Proof: 'a \Rightarrow b' Since $(p_n)_{n \in \mathbb{N}}$ is a basic sequence, Theorem 2.3.10 yields the existence of a sequence $(\gamma_n)_{n \in \mathbb{N}}$ with $\gamma_0 = 0$ such that $p_n = \sum_{k=0}^n \gamma_n^{k*} q_k$ for all $n \in \mathbb{N}$. Since deg $p_1 = 1$ (Remark 2.2.14), we have $\gamma_1 \neq 0$. Since $Aq_n = s_n$ for all $n \in \mathbb{N}$, it follows that $r_n = A p_n = \sum_{k=0}^n \gamma_n^{k*} s_k$ for all $n \in \mathbb{N}$.

'b \Rightarrow a' Define polynomials $p_n \ (n \in \mathbb{N})$ by $p_n := \sum_{k=0}^n \gamma_n^{k*} q_k$. Since $\gamma_0 = 0$ and $\gamma_1 \neq 0$, it follows from Theorem 2.3.10 and Theorem 2.2.19 that $(p_n)_{n \in \mathbb{N}}$ is a basic sequence. Moreover, it is obvious that $A p_n = r_n$ for all $n \in \mathbb{N}$ since $A q_n = s_n$. It follows from Theorem 2.4.8b that $(r_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence.

Remark 2.4.13 There exists no analogue of Theorem 2.4.12 for wide sense Sheffer sequences $(r_n)_{n\in\mathbb{N}}$. First of all, it is necessary that $(s_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence: if $(q_n)_{n\in\mathbb{N}}$ is not a basic sequence, then the operator Aneed not exist (cf. Remark 2.1.13b). Suppose $(s_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence and $r_n = \sum_{k=0}^n \gamma_n^{k*} s_k$ for all $n \in \mathbb{N}$. If $\gamma_1 \neq 0$, then $(r_n)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence by Theorem 2.4.12. If $\gamma_1 = 0$, then the proof of Theorem 2.1.12c yields that deg $r_n \leq [n/2]$ for all $n \in \mathbb{N}$. It follows from Theorem 2.4.2 that $(r_n)_{n\in\mathbb{N}}$ cannot be a wide sense Sheffer sequence.

As a corollary to Theorem 2.4.12, we now derive a Rodrigues Formula for strict sense Sheffer sequences (cf. Theorem 2.3.6d). This form of the Rodrigues Formula is due to Avramjonok (see [13]).

Theorem 2.4.14 (Avramjonok) Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. Let $(s_n)_{n \in \mathbb{N}}$ be a strict sense Sheffer sequence for Q and let A be the linear operator on \mathcal{P} defined by $Aq_n := s_n$ for all $n \in \mathbb{N}$. Then we have

$$n s_n(x) = \left(x \left(Q' \right)^{-1} + \left(Q' \right)^{-1} A' A^{-1} \right) s_{n-1}(x).$$
(2.5)

Proof: By Theorem 2.3.6d, we have $n q_n(x) = x (Q')^{-1} q_{n-1}(x)$ for all $n \ge 1$ and all x. Writing $q_k = A^{-1} A q_k$ (k = n - 1, n), we obtain $n s_n(x) = A x (Q')^{-1} A^{-1} s_{n-1}(x)$. By the definition of Pincherle derivative, we may write A x = x A + A'. Substituting this into the expression for $n s_n(x)$, we obtain the result.

We conclude this section with a probabilistic subclass of Appell polynomials.

Definition 2.4.15 Let X be a random variable with finite moments of all orders. The Wick polynomial sequence associated to X is the unique sequence $(p_n)_{n \in \mathbb{N}}$ of polynomials satisfying:

- 1. $D p_n = p_{n-1} \text{ for } n \ge 1$
- 2. $E p_n(X) = \delta_{0n}$,

where D denotes the differentiation operator and δ_{0n} the Kronecker delta.

Since Λ , defined by $\Lambda(p) := E p(X)$ is a linear functional on \mathcal{P} such that $\Lambda 1 = 1$, it follows from Lemma 2.4.7 with Q = D and Theorem 2.4.8 that $(p_n)_{n \in \mathbb{N}}$ is a (well-defined) strict sense Sheffer sequence.

Wick polynomials occur in quantum mechanics and in probability theory. In the latter case, they are used for noncentral limit theorems (see e.g. [12, 106]).

Theorem 2.4.16 Let X be a random variable with distribution function F such that its moment generating function $\int_{-\infty}^{\infty} e^{zt} dF(t)$ is analytic on some disc in the complex plane. Then the Wick polynomials $(p_n)_{n \in \mathbb{N}}$ associated to X possess the following generating function:

$$\sum_{n=0}^{\infty} p_n(X) \, z^n = \frac{e^{zx}}{\int_{-\infty}^{\infty} e^{zt} \, dF(t)}.$$

Proof: let A be the linear operator on \mathcal{P} defined by $Axfacn := p_n$ for all $n \in \mathbb{N}$. First note that by Theorem 2.2.22a, we have $A = \sum_{k=0}^{\infty} p_k(0) Q^k$. and By Theorems 2.2.7 and 2.4.10d, we have

$$A^{-1} = \sum_{n=0}^{\infty} \left(A^{-1} \, \frac{x^n}{n!} \right) (0) \, D^n = \sum_{n=0}^{\infty} \Lambda \left(\frac{x^n}{n!} \right) D^n = \sum_{n=0}^{\infty} E\left(\frac{x^n}{n!} \right) D^n.$$

The result now follows from Theorem 2.3.1 and Corollary 2.4.6, since under the conditions of the theorem the moment generating function equals $\sum_{n=0}^{\infty} E(X^n) \frac{z^n}{n!}.$

- **Examples 2.4.17** 1. If X is distributed according to the standard normal distribution, then its moment generating function equals $e^{\frac{1}{2}z^2}$. Thus the Wick polynomials for the standard normal distribution are the Hermite polynomials (cf. Example 2.4.3a).
 - 2. If X is distributed according to the uniform distribution on [0, 1], then its moment generating function equals $(e^{z} 1)/z$. Thus the Wick polynomials for the uniform distribution on [0, 1] are the Bernoulli polynomials (cf. Example 2.4.3a).

The standard theory of Wick polynomials can be derived easily from the theory of this section (cf [12, 106]).

We conclude this section by remarking that Al-Salam and Verma have generalized Sheffer sequences by considering sequences of polynomials satisfying $Qs_n = s_{n-r} \ (r \in \mathbb{N})$ for a delta operator Q (see [7]).

2.5 Cross sequences and Steffensen sequences

In the previous section we extended the notion of basic sequence by relaxing one of the defining properties. In this section we extend the notion of basic sequence by adding an extra parameter. This extra parameter comes in naturally for basic sequences connected to probability distributions. E.g., for the Poisson-Charlier polynomials this extra parameter is the parameter of the underlying Poisson distribution and for the Hermite polynomials it is the variance of the underlying zero-mean normal distribution. This section unites and extends the results of [183], [202, Section 5.3], [210, Section 8] and [38]. New is the introduction of semigroups of shift-invariant operators.

In this section we assume that the polynomials are defined on \mathbb{R} and have real coefficients. We also assume that each sequence of polynomials $(p_n)_{n \in \mathbb{N}}$ satisfies deg $p_n = n$. Unlike the previous section, there is no use in considering *weak* and *strong* versions of cross or Steffensen sequences.

Definition 2.5.1 A sequence of polynomials $(q_n^{[\lambda]})_{n \in \mathbb{N}}$ is said to be a cross sequence if

a)
$$\left(q_n^{[\lambda]}\right)_{n\in\mathbb{N}}$$
 is a sequence of polynomials for fixed λ .
b)

$$q_n^{[\lambda+\mu]}(x+y) = \sum_{k=0}^n q_k^{[\lambda]}(x) q_{n-k}^{[\mu]}(y)$$
(2.6)

for all $n \in \mathbb{N}$ and all $x, y, \lambda, \mu \in \mathbb{R}$.

It is obvious that any sequence of polynomials with (formal) generating function of the form

$$\sum_{n=0}^{\infty} q_n^{[\lambda]}(x) t^n = e^{\lambda h(t)} A(t) e^{x g(t)}$$

is a cross sequence. Also note that $(q_n^{[\lambda]})_{n\in\mathbb{N}}$ is Sheffer for fixed λ and that $(q_n^{[\lambda]}(x))_{n\in\mathbb{N}}$ is a cross sequence in the variable λ with parameter x.

We now wish to give a characterization of cross sequences in terms of shiftinvariant operators. Since this involves (semi-)groups of shift-invariant operators, we digress a little bit by studying these semigroups.

Definition 2.5.2 A family $(T_t)_{t>0}$ of linear shift-invariant operators on \mathcal{P} is a semigroup if $T_{s+t} = T_s T_t$ for all s, t > 0.

Theorem 2.5.3 If $(T_t)_{t>0}$ is a semigroup of linear shift-invariant operators on \mathcal{P} , then T_t is invertible for all t > 0 and hence, $(T_t)_{t>0}$ can be extended to a group $(T_t)_{t\in\mathbb{R}}$.

Proof: It follows from Corollary 2.2.9a that there exist non-negative integers n(t) such that $\deg(T_tp) = \max\{-1, \deg(p) - n(t)\}$ for all t > 0 and all $p \in \mathcal{P}$. The semigroup property implies that n(s+t) = n(t) + n(s) for all s, t > 0. Hence, n(t) = 0 for all t > 0, since n(t) is integer-valued for all t > 0. Thus T_t is invertible for all t > 0 by Corollary 2.2.11. Define $T_0 := I$ and $T_t := (T_{-t})^{-1}$ for all t < 0. Then $(T_t)_{t \in \mathbb{R}}$ is a group, since obviously $T_{s+t} = T_s T_t$ for all $s, t \in \mathbb{R}$.

We now expand the operators T_t into powers of D as in Theorem 2.2.7 and study the coefficients of these expansions.

Theorem 2.5.4 Let $(T_t)_{t>0}$ be a semigroup of linear shift-invariant operators on \mathcal{P} and let the functions $a_n (n \in \mathbb{N})$ be defined by $T_t = \sum_{n=0}^{\infty} a_n(t) D^n$ for all t > 0 and all $n \in \mathbb{N}$. Then:

- a) the sequence $(a_n)_{n \in \mathbb{N}}$ is a sequence of functions of convolution type.
- b) if $(a_n)_{n \in \mathbb{N}}$ is a sequence of measurable functions, then there exists a linear shift-invariant operator T on \mathcal{P} such that $T_t = e^{tT}$ for all t > 0 and $(T_t)_{t>0}$ can be extended to a group $(T_t)_{t \in \mathbb{R}}$.

Proof: a) This follows from

$$\sum_{n=0}^{\infty} a_n(s+t) D^n = T_{s+t} = T_s T_t =$$
$$\sum_{m=0}^{\infty} a_m(s) D^m \sum_{r=0}^{\infty} a_r(t) D^r = \sum_{n=0}^{\infty} \sum_{k=0}^n a_k(s) a_{n-k}(t) D^n$$

for all s, t > 0.

b) By Theorem 2.1.8, the measurability of a_n implies that there exist an $a \in \mathbb{R}$ and a sequence of real numbers $(g_n)_{n \in \mathbb{N}}$ such that $a_n(t) = e^{at} \sum_{k=0}^n g_n^{k*} \frac{t^k}{k!}$. Define the linear shift-invariant operator T on \mathcal{P} by $T := aI + \sum_{k=1}^{\infty} g_k D^k$. Then

$$e^{tT} = \exp\left(taI + t\sum_{r=0}^{\infty} g_r D^r\right) = \left(\sum_{k=0}^{\infty} \frac{t^k a^k I^k}{k!}\right) \left(\sum_{m=0}^{\infty} \frac{t^m}{m!} \sum_{r=0}^{\infty} g_r^{m*} D^r\right) = e^{at} \left(\sum_{r=0}^{\infty} \sum_{m=0}^{\infty} \frac{g_r^{m*} t^m}{m!} D^r\right) = e^{at} \sum_{r=0}^{\infty} a_r(t) D^r = T_t.$$

Moreover, we may extend $(T_t)_{t>0}$ to a group $(T_t)_{t\in\mathbb{C}}$ by setting $T_t := e^{tT}$ for all $t \in \mathbb{C}$.

The operator T of Theorem 2.5.4 is called **infinitesimal generator** of the semigroup $(T_t)_{t>0}$ in standard semigroup theory.

The following theorem describes when the functions a_n of the above theorem are measurable.

Theorem 2.5.5 Let $(T_t)_{t>0}$ be a semigroup of linear shift-invariant operators on \mathcal{P} and let the functions $a_n(n \in \mathbb{N})$ be defined by $T_t = \sum_{n=0}^{\infty} a_n(t) D^n$ for all t > 0 and all $n \in \mathbb{N}$. Then the following are equivalent:

- a) the functions a_n are continuous.
- b) the functions a_n are measurable.
- c) $\lim_{t\downarrow 0} \left(\frac{T_t-I}{t}p\right)(x) = Tp(x)$ for all $p \in \mathcal{P}$ and all $x \in \mathbb{R}$, where T is the infinitesimal generator of $(T_t)_{t>0}$.
- d) $\lim_{t \downarrow 0} (T_t p)(x) = p(x)$ for all $p \in \mathcal{P}$ and all $x \in \mathbb{R}$.

Proof: ' $a \Rightarrow b$ ' This is trivial, since continuous functions are measurable. ' $b \Rightarrow c$ ' By Theorem 2.5.4, we have

$$\lim_{t \downarrow 0} \frac{T_t - I}{t} p(x) = \lim_{t \downarrow 0} \frac{e^{tT} - I}{t} p(x) = \lim_{t \downarrow 0} \frac{1}{t} \sum_{n=1}^{\infty} (tT)^n p(x) = T p(x),$$

since there are only finitely many nonzero terms in the summation. $c \Rightarrow d$ This is trivial.

 ${}^{'}d \Rightarrow a'$ Note that $(T_t x^n)(0) = \sum_{k=0}^n a_k(t) (D^k x^n)(0) = a_n(t)$ for all t > 0and all $n \in \mathbb{N}$. Hence, $\lim_{t \downarrow 0} a_n(t)$ exists for all $n \in \mathbb{N}$. Now Theorem 2.5.4a and Remark 2.1.10g yield that a_n is continuous for all $n \in \mathbb{N}$.

We now return to cross sequences. The following theorem characterizes cross sequences as the orbit of a basic sequence under a group of shift-invariant operators.

Theorem 2.5.6 ([210]) A sequence $(q_n^{[\lambda]})_{n\in\mathbb{N}}$ is a cross sequence if and only if there exists a delta operator Q with basic sequence $(q_n)_{n\in\mathbb{N}}$ and a group of shift-invariant operators $(T_t)_{t\in\mathbb{R}}$ such that $q_n^{[\lambda]} = T_{\lambda} q_n$.

Proof: ' \Rightarrow ' If $\left(q_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a cross sequence, then it follows from (2.6) and Theorem 2.2.19 that $\left(q_n^{[0]}\right)_{n\in\mathbb{N}}$ is of convolution type with delta operator Q, say. Moreover, by setting $\mu = 0$ in (2.6) we see that for fixed λ , $\left(q_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a Sheffer sequence with delta operator Q and invertible operator T_{λ} , say. In order to show that $(T_t)_{t\in\mathbb{R}}$ is a group of linear operators, it suffices to show that $T_{\lambda+\mu}q_n = T_{\lambda}T_{\mu}q_n$. By Theorem 2.4.4 and Remark 2.4.5a, we have $q_n^{[\lambda]}(x) =$ $\sum_{k=0}^n q_{n-k}^{[\lambda]}(0) q_k(x)$. Since $T_{\lambda}q_n = q_n^{[\lambda]}$, it follows from Theorem 2.2.22a that $T_{\lambda} = \sum_{k=0}^{\infty} q_k^{[\lambda]}(0) Q^k$. Now $T_{\lambda+\mu}q_n = T_{\lambda}T_{\mu}q_n$ follows from (2.6) with x = 0. ' \Leftarrow ' First note that since T_{λ} is an invertible shift-invariant operator, $\left(q_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a strict sense Sheffer sequence for fixed λ . Hence,

$$E^{y} q_{n}^{[\lambda]}(x) = \sum_{k=0}^{n} q_{k}^{[\lambda]}(y) q_{n-k}(x).$$

Applying the shift-invariant operator T_{μ} to both sides of the last equation, we obtain that $\left(q_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a cross sequence.

Theorem 2.5.7 Let $(q_n^{[\lambda]})_{n\in\mathbb{N}}$ be a cross sequence. Then each polynomial p can be expanded as

$$p = \sum_{n=0}^{\infty} \left(T_{\lambda} Q \right) (0) q_n^{[\lambda]}$$
(2.7)

where Q and $(q_n)_{n \in \mathbb{N}}$ are as in Theorem 2.5.6.

Proof: This follows directly from Theorem 2.4.11, since $\left(q_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a Sheffer sequence for fixed λ .

Examples 2.5.8 Examples of cross sequences include:

a) (Hermite polynomials): Q = D, $T_{\lambda} = e^{-\frac{1}{2}\lambda D^2}$ (see [202, pp.87-97] for more details). The more general cross sequence with $T_{\lambda} = e^{-\lambda D^m}$ is studied in [183]. Their formulas follow directly from the results of Section 2.4 and 2.5. E.g., the Rodrigues Formula for Sheffer sequences (Theorem 2.4.14) yields

$$s_n(x) = x s_{n-1}(x) + m \lambda D^{m-1} e^{\lambda D^m} x^{n-1}$$

= $x s_{n-1}(x) + m \lambda D^{m-1} s_{n-1}(x)$
= $x s_{n-1}(x) + m \lambda s_{n-m}(x)$,

which is [183, Formula (5.2)].

- b) (Bernoulli polynomials): $Q = D, T_{\lambda} = (e^D 1)/D)^{-\lambda} = (D/(e^D 1))^{\lambda}$ (see [202, pp. 93-100] for more details).
- c) (Euler polynomials): Q = D, $T_{\lambda} = ((e^D + 1)/2)^{-\lambda}$ (see [202, pp. 101-106] for more details.
- d) (**Poisson-Charlier polynomials**): $Q = e^D 1$, $T_{\lambda} = e^{-\lambda} (e^D 1)$. This differs a factor λ^n from the ordinary definition of Poisson-Charlier polynomial (see [202, pp. 119-122] for more details).
- e) (actuarial polynomials): $Q = \log(1-D), T_{\lambda} = (1-D)^{\lambda}$ (see [202, pp. 123-125] for more details).

Definition 2.5.9 A sequence of polynomials $(s_n^{[\lambda]})_{n\in\mathbb{N}}$ is said to be a Steffensen sequence if

a) $\left(s_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a sequence of polynomials for fixed λ .

b) there exists a basic sequence $(q_n)_{n\in\mathbb{N}}$ such that

$$s_{n}^{[\lambda+\mu]}(x+y) = \sum_{k=0}^{n} s_{k}^{[\lambda]}(x) q_{n-k}^{[\mu]}(y)$$
(2.8)

for all $n \in \mathbb{N}$ and all $x, y, \lambda, \mu \in \mathbb{R}$.

Theorem 2.5.10 Let $(s_n^{[\lambda]})_{n \in \mathbb{N}}$ be a sequence of polynomials for fixed λ . Then the following are equivalent:

- a) $\left(s_n^{[\lambda]}\right)_{n\in\mathbb{N}}$ is a Steffensen sequence
- b) there exists a cross sequence $(q_n^{[\lambda]})_{n \in \mathbb{N}}$ and an invertible shift-invariant operator A such that $s_n^{[\lambda]} = Aq_n^{[\lambda]}$ for all $n \in \mathbb{N}$
- c) there exists a group $(T_t)_{t\in\mathbb{R}}$ of linear shift-invariant operators and a Sheffer sequence $(s_n)_{n\in\mathbb{N}}$ such that $s_n^{[\lambda]} = T_{\lambda} s_n$.

Proof: ' $a \Rightarrow b$ ' Setting $\lambda = x = 0$ in (2.8) we see that $s_n^{[\mu]} = \sum_{k=0}^n s_k^{[0]}(0) q_{n-k}^{[\mu]}$. Note that $s_0^{[0]} \neq 0$, since $\left(s_n^{[0]}\right)_{n \in \mathbb{N}}$ is Sheffer. Hence, $s_n^{[\mu]} = Aq_n^{[\lambda]}$, where A is the invertible linear shift-invariant operator defined by $A := \sum_{k=0}^{\infty} s_k^{[0]}(0) Q^k$. ' $b \Rightarrow c$ ' This follows directly from Theorem 2.5.6. ' $c \Rightarrow a$ ' By Theorem 2.5.6, we have

$$s_n^{[\lambda+\mu]}(x+y) = E^y A q_n^{[\lambda+\mu]}(x) = E^y A \left(\sum_{k=0}^n q_k^{[\lambda]}(x) q_{n-k}^{[\mu]}\right)(0) = \sum_{k=0}^n q_{n-k}^{[\mu]} s_k^{[\lambda]}(x).$$

This concludes the proof.

Example 2.5.11 As an example of a Steffensen sequence we mention the Laguerre polynomials Q = D/(D-I), A = I - D, $T_{\lambda} = (I - D)^{\lambda}$ (see [202, pp. 108-113] for more details).

Theorem 2.5.12 Let $(s_n^{[\lambda]})_{n\in\mathbb{N}}$ be a Steffensen sequence. Then each polynomial p can be expanded as

$$p = \sum_{n=0}^{\infty} \left(A T_{\lambda} Q \right) (0) s_n^{[\lambda]}$$
(2.9)

where Q and $(q_n)_{n \in \mathbb{N}}$ are as in Theorem 2.5.10.

Proof: This follows directly form Theorem 2.4.11, since $(s_n^{[\lambda]})_{n \in \mathbb{N}}$ is a Sheffer sequence for fixed λ .

Theorem 2.5.13 ([38]) If $(s_n^{[\lambda]})_{n \in \mathbb{N}}$ is a Steffensen sequence and σ_n is a sequence of real numbers, then $(s_n^{[\sigma_n]})_{n \in \mathbb{N}}$ is a Sheffer sequence if and only if there exists real numbers α and β such that $\sigma_n = \alpha + \beta n$.

Proof: By Theorem 2.5.10, there exists a Sheffer sequence $(s_n)_{n\in\mathbb{N}}$ for some delta operator Q and a group of linear shift-invariant operators $(T_t)_{t\in\mathbb{R}}$ such that $s_n^{[\lambda]} = T_{\lambda} s_n$. Hence, $Qs_n^{[\alpha+\beta n]} = QT_{\alpha+\beta n} = T_{\alpha+\beta n} s_{n-1} = T_{\beta} s_{n-1}^{[\alpha+\beta(n-1)]}$. In other words, $(s_n^{[\alpha+\beta n]})_{n\in\mathbb{N}}$ is Sheffer for the delta operator $T_{-\beta}Q$. For the converse, see [38].

Example 2.5.14 For the Laguerre polynomials $(L_n^{[\lambda]})_{n\in\mathbb{N}}$, it is easy to compute that $L_n^{[x-n]}(\lambda)$ is the n^{th} Poisson-Charlier polynomial of Example 2.5.8d.

Chapter 3

Applications of the Umbral Calculus

In this chapter we present a miscellany of new applications and new results concerning Umbral Calculus and polynomials of convolution type.

In Section 3.1 all sequences of polynomials of convolution type are determined such that $q_n(1) = c$ for all $n \ge 1$. Section 3.2 shows how the theory of Chapter 2 yields identities with binomial coefficients. Moreover, a new proof of a result by G. Labelle on polynomials of convolution type is given. In Section 3.3 probability distributions arising from polynomials of convolution type are studied. The calculation of moments of these distributions (which is of importance for approximation theory) will be calculated using the operator methods of Chapter 2. A simplified proof of the classification of orthogonal Sheffer polynomials is presented in Section 3.4. In Section 3.5 polynomials of convolution type are related to semigroups of probability measures. It transpires that in this context umbral composition can be interpreted as subordination. It is shown in Section 3.6 that each shift-invariant operator can be written as an integral operator. As a corollary, a characterization of Sheffer sequences due to Sheffer is obtained. This representation is shown to be connected with moment problems. Finally, in Section 3.7 we study natural exponential families from an umbral point of view. It is shown that the variance function of a natural exponential family is intimately related to the delta operator of its associated Sheffer sequence. In fact, we will see that the classification of natural exponential families with quadratic variance function coincides with the classification of orthogonal Sheffer polynomials of Section 3.4. We will also see how natural exponential families are related to exponential operators appearing in approximation theory.

Contents of Chapter 3

3.1 Polynomials with $q_n(1) = c$ for $n \ge 1$.

- 3.2 Applications to combinatorial identities.
- 3.3 Discrete probability distributions.
- 3.4 Orthogonal Sheffer polynomials
- **3.5** Moment sequences
- 3.6 Shift-invariant operators and integral operators
- 3.7 Natural exponential families

3.1 Polynomials with $q_n(1) = c$ for $n \ge 1$.

In this section we determine all sequences $(q_n)_{n \in \mathbb{N}}$ of polynomials of convolution type such that $q_n(1) = c$ for $n \geq 1$. Recall from Theorem 2.1.14 that polynomials of convolution type are determined by the numbers $q_n(1)$.

Theorem 3.1.1 If $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type such that $q_n(1) = c$ for $n \ge 1$, then

$$q_n(x) = \sum_{k=0}^n \, \binom{x}{k} \, (c-1)^k \, \binom{x+n-k}{n-k} \, .$$

Proof: First note that there exists a unique sequence of polynomials of convolution type such that $q_n(1) = c$ for $n \ge 1$ by Theorem 2.1.14 with $x_n = 1$ for all $n \in \mathbb{N}$. Both $\binom{x}{n}(c-1)^n_{n\in\mathbb{N}}$ and $\binom{x+n}{n}_{n\in\mathbb{N}}$ are of convolution type (see Remark 2.2.18), so their convolution is also of convolution type by Remark 2.1.3e. Since $\sum_{k=0}^{\infty} (c-1)^k \binom{1+n-k}{n-k} = c$ for all $n \ge 1$, the theorem follows.

Remarks 3.1.2 : a) Another way of proving Theorem 3.1.1 is to use generating functions:

$$\sum_{n=0}^{\infty} q_n(1) z^n = 1 + c \sum_{n=0}^{\infty} z^n = 1 + \frac{c z}{1-z} = (1 + (c-1) z) \frac{1}{1-z}.$$

The theorem now follows by observing that

$$\sum_{n=0}^{\infty} {\binom{x+n-1}{n}} (c-1)^n z^n = (1+(c-1)z)^x$$

and

$$\sum_{n=0}^{\infty} \binom{x+n-1}{n} z^n = \left(\frac{1}{1-z}\right)^x.$$

b) It follows from a) that

$$\sum_{n=0}^{\infty} q_n(x) z^n = \left(\frac{1+(c-1)z}{1-z}\right)^x = (1-f(z))^{-x}$$

with $f(z) = \frac{cz}{1 + (c-1)z}$. If 0 < c < 1, then f is a probability generating function and $(n!q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of binomial type with the renewal property in the terminology of [224].

Examples 3.1.3 : a) If c = 1, then $q_n(x) = \binom{x+n-1}{n}$ (see Example 2.2.16c). b) If c = 2, then $q_n(x) = \sum_{k=0}^n \binom{x}{k} \binom{x+n-k}{n-k}$, the Mittag-Leffler polynomials (see [202, p. 75-76] note that the calculation on p. 76 contains an error).

3.2 Applications to combinatorial identities

In this section an identity for convolutions of sequences of numbers (in some field of characteristic zero \mathcal{K}) will be derived. Moreover, the theory of polynomials of convolution type of Chapter 2 will be used to calculate convolutions of such sequences. This will yield combinatorial identities.

Theorem 3.2.1 Let $(\beta_n)_{n \in \mathbb{N}}$ be a sequence in some field of characteristic zero. The r-fold convolution of the sequence $\left(\frac{1}{n+1}\beta_n^{(n+1)*}\right)_{n \in \mathbb{N}}$ is the sequence $\left(\frac{r}{n+r}\beta_n^{(n+r)*}\right)_{n \in \mathbb{N}}$. In particular, the following holds for $1 \leq i, j \leq n$: $\frac{i+j}{n+i+j}\beta_n^{(n+i+j)*} = \sum_{m=0}^n \frac{i}{m+i}\beta_m^{(m+i)*}\frac{j}{n-m+j}\beta_{n-m}^{(n-m+j)*}$.

Proof: If $\beta_0 = 0$, then $\beta_n^{m*} = 0$ for all m > n by Lemma 2.1.5a. Therefore we may suppose that $\beta_0 \neq 0$. Consider the operator $T := \sum_{r=0}^{\infty} \beta_r D^r$ on \mathcal{P} . It follows from Corollary 2.2.11 that T is invertible. Define $U := T^{-1}$ and consider the delta operator Q := DU with basic sequence $(q_n)_{n \in \mathbb{N}}$. It follows from Theorem 2.1.8 that $q_m(x) = \sum_{k=0}^m g_m^{k*} \frac{x^k}{k!}$. Since $U^{-m} = \sum_{r=0}^{\infty} \beta_r^{m*} D^r$, Theorem 2.3.6c yields

$$q_m(x) = \frac{x}{m} U^{-m} \frac{x^{m-1}}{(m-1)!} = \frac{x}{m} \sum_{r=0}^{m-1} \beta_r^{m*} D^r \frac{x^{m-1}}{(m-1)!} =$$
$$\sum_{r=0}^{n-1} \beta_r^{m*} \frac{x^{m-1-r}}{(m-1-r)!} = \frac{x}{m} \sum_{k=1}^m \beta_{m-k}^{m*} \frac{x^{k-1}}{(k-1)!} = \sum_{k=1}^m \frac{k}{m} \beta_{m-k}^{m*} \frac{x^k}{k!}.$$

Comparing coefficients of q_m yields $g_m^{k*} = \beta_{m-k}^{m*}$ for $1 \le k \le m$. Setting k = r and m = n + r yields the first statement.

The second statement follows from the first statement and Remark 2.1.14c. \Box

Using Lagrange inversion, Steutel derived a similar convolution identity and some extensions (see [232]).

Theorem 3.2.1 in itself yields interesting identities. However, more insight can be obtained by relating Theorem 3.2.1 to a non-abelian group structure on the set of sequences of polynomials of convolution type introduced by G. Labelle (see [140, Proposition 1]). We first need a lemma. The proof of Lemma 2.2.2 was shown to the author by Piet Bruinsma.

Lemma 3.2.2 Let P be a polynomial in two variables with coefficients in some field \mathcal{K} of characteristic zero such that P(l,m) = 0 for all $l,m \in \mathbb{N}$. Then P = 0.

Proof: Denote the coefficients of P by a_{ij} , i.e. $P(x,y) = \sum_{i,j=0}^{n} a_{ij} x^i y^j$. Consider the matrix P, defined by $P(i,j) := a_{ij}$, acting on \mathcal{K}^{n+1} . For each $l, m \in \mathbb{N}$, we have $(1, l, \ldots, l^n) P((1, m, \ldots, m^n)^t) = 0$. It follows directly from Vandermonde's determinant that the vectors $(1, l, \ldots, l^n)^t$, $l = 1, \ldots, n$, are a basis for \mathcal{K}^{n+1} . Hence, $(1, m, \ldots, m^n)^t$ belongs to the kernel of P for each $m \in \mathbb{N}$. Therefore P is the zero matrix and P = 0.

The following theorem was proved for basic sequences in [210, Proposition 4]. An extension to sequences of polynomials of convolution type was given in [140, Proposition 2]. We present here a new proof based on Theorem 3.2.1.

Theorem 3.2.3 If $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type, then for all a both $\left(\frac{x}{x+na}q_n(x+na)\right)_{n \in \mathbb{N}}$ and $\left(\frac{x}{x+na}q_n(-x-na)\right)_{n \in \mathbb{N}}$ are sequences of polynomials of convolution type.

Proof: If a = 0, then there is nothing to prove. If $q_0 = 0$, then the result follows from Lemma 2.1.7. Suppose $q_0 \neq 0$ and $a \neq 0$. It follows from (2.1) that $(r_n)_{n \in \mathbb{N}}$, defined by $r_n(x) := q_n(ax)$ for all $n \in \mathbb{N}$, is a sequence of polynomials of convolution type. It follows from Theorem 2.1.8 that $p_n(x) := \frac{x}{x+n}r_n(x+n)$ is a polynomial in x for all $n \in \mathbb{N}$. Define $\beta_n := r_n(1)$ for all $n \in \mathbb{N}$. It follows from (2.1) that $\beta_n^{k*} = r_n(k)$ for all $k, n \in \mathbb{N}$. Fix an arbitrary $n \in \mathbb{N}$. Define the polynomial P in two variables by $P(x, y) := p_n(x+y) - \sum_{j=0}^n p_j(x) p_{n-j}(y)$. It follows from Theorem 3.2.1 that P(l,m) = 0 for all $l,m \in \mathbb{N}$. By Lemma 3.2.2, P = 0. Since n was arbitrary, it follows that $(p_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type. By Remark 2.1.10b we have for all u, v:

$$\frac{u+v}{u+v+n} q_n(a(u+v)+an) =$$

$$\sum_{k=0}^n \frac{u}{u+k} q_k(au+ak) \frac{v}{v+n-k} q_{n-k}(av+a(n-k)).$$

Multiplying the numerators and denominators of the above identity with a and taking x = ua and y = va, we obtain for all x, y:

$$\frac{x+y}{x+y+na}\,q_n(x+y+an) =$$

$$\sum_{k=0}^{n} \frac{x}{x+ka} q_k(x+ak) \frac{y}{y+(n-k)a} q_{n-k}(y+a(n-k)).$$

Thus, $\left(\frac{x}{x+na}q_n(x+na)\right)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type. For the second statement note that $(u_n)_{n\in\mathbb{N}}$, defined by $u_n(x) := q_n(-x)$ for all $n \in \mathbb{N}$, is a sequence of polynomials of convolution type by Theorem 2.1.8 and Remark 2.1.10b. Applying the first statement to $(u_n)_{n\in\mathbb{N}}$ instead of $(q_n)_{n\in\mathbb{N}}$ yields the second statement.

Examples 3.2.4 a) If $q_n(x) = \frac{x^n}{n!}$, then $\frac{x}{x+na}q_n(x+na) = x(x+na)^{n-1}/n!$, the n^{th} Abel polynomial.

If $q_n(x) = {x \choose n}$, then $\frac{x}{x+na} q_n(x+na)$ is called the n^{th} Gould polynomial.

Remark 3.2.5 It is possible to derive Theorem 3.2.3 from [210, Proposition 4] in case the coefficients are real or complex. This proof is due to Aart Stam (private communication). Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type. If $g_1 \neq 0$, then deg $q_n = n$ for all $n \in \mathbb{N}$ by Theorem 2.1.12a and the result follows from Theorem 2.2.19 and [210, Proposition 4]. If $g_1 = 0$, then we define a sequence of complex numbers $(h_n)_{n\in\mathbb{N}}$ by $h_n := g_n$ if $n \neq 1$ and $h_1 := \varepsilon(\varepsilon \neq 0)$. Then $(h_n)_{n\in\mathbb{N}}$ is the coefficient sequence of a sequence $(r_n)_{n\in\mathbb{N}}$ of polynomials of convolution type with deg $r_n = n$ for all $n \in \mathbb{N}$. Letting ε go to zero and applying Lemma 2.1.5c, we obtain the desired result.

For examples of identities arising from Theorem 3.2.3, see Examples 3.2.7.

We now use Theorem 2.3.10 to calculate convolutions of scalar sequences.

Theorem 3.2.6 Let $(\beta_n)_{n\in\mathbb{N}}$ be a sequence in some field of characteristic zero with $\beta_0 = 0$ such that the following holds: there exists sequences of polynomials of convolution type $(p_n)_{n\in\mathbb{N}}$ and $(q_n)_{n\in\mathbb{N}}$ such that $\beta_n = a_{n,1}$ for all $n \in \mathbb{N}$, where the numbers $a_{n,k}$ are defined by $p_n = \sum_{k=0}^n a_{n,k} q_k$. Then $\beta_n^{k*} = a_{n,k}$ for all $k, n \in \mathbb{N}$.

Proof: This follows directly from Theorem 2.3.10.

Examples 3.2.7 a) Consider the basic sequence $\binom{x}{n}_{n \in \mathbb{N}}$ of Example 2.2.16b. By definition ([199, p. 33]), $\binom{x}{n} = \sum_{k=0}^{n} \frac{k!}{n!} s(n,k) \frac{x^{k}}{k!}$, where the numbers s(n,k) are the Stirling numbers of the first kind. If $g_n := s(n,1)/n! = (-1)^{n-1}/n$, then $g_n^{k*} = s(n,k)$ by Theorem 3.2.6. Thus Remark 2.1.3c yields after some simplifications:

$$\binom{k}{i}s(n,k)=\sum_{m=0}^{n}\binom{n}{m}s(m,i)s(n-m,k-i).$$

Similar identities can be obtained for the signless Stirling numbers of the first kind and the Stirling numbers of the second kind.

Applying Theorem 3.2.3 we obtain the following identity for Gould polynomials (see e.g., [210, Section 12]):

$$\frac{x+y}{x+y-an} \begin{pmatrix} x+y-an\\ n \end{pmatrix} =$$
$$\sum_{k=0}^{n} \frac{x}{x-ak} \begin{pmatrix} x-ak\\ k \end{pmatrix} \frac{y}{y-a(n-k)} \begin{pmatrix} y-a(n-k)\\ n-k \end{pmatrix}.$$

A similar identity holds for the upper factorials of Example 2.2.16c. The sequence of polynomials $\left(\frac{x}{x-an} \begin{pmatrix} x-an\\ n \end{pmatrix}\right)_{n\in\mathbb{N}}$ is the basic sequence of the operator $E^a(E-I)$ (cf. [210, Section 12]). Since

$$\left(E^{a}\left(E-I\right)\right)^{k} \left(\binom{x-an}{n}\right)(0) = \binom{-a(n-k)}{n-k},$$

Lemma 2.2.21 yields

$$\sum_{k=0}^{n} \frac{x}{x-ak} \begin{pmatrix} x-ak \\ k \end{pmatrix} \begin{pmatrix} -a(n-k) \\ n-k \end{pmatrix} = \begin{pmatrix} x-an \\ n \end{pmatrix}.$$

The above identity can also be obtained by noting that $\binom{x-an}{n}_{n\in\mathbb{N}}$ is a Sheffer sequence for the delta operator $E^a(E-I)$ (Definition 2.4.1) and by applying Theorem 2.4.4. For a generalization of this identity, see [210, p. 736]. b) Consider the basic sequence $\binom{x^n}{n!}_{n\in\mathbb{N}}$. Then Theorem 3.2.3 yields another proof of the Abel generalization of the Binomial Formula (cf. Remark 2.2.18d). c) Consider $g(z) = \frac{1}{2} \left(1 - (1 - 4z)^{\frac{1}{2}} \right)$, the generating function of the Catalan numbers $C_n = \frac{k}{n} \binom{2n-2}{n-1}$ $(n \ge 1)$. It is not easy to calculate convolutions of the Catalan numbers directly. Consider the compositional inverse of g. This is $f(z) = z - z^2$. Let Q be the delta operator $D - D^2$. It follows from Theorem 2.2.22b that Q is the delta operator of the basic polynomials $(q_n)_{n\in\mathbb{N}}$ whose coefficient sequence is the sequence of Catalan numbers. Theorem 2.3.6 yields

$$q_n(x) = \frac{x}{n!} (I - D)^{-n} x^{n-1} = \frac{x}{n!} \sum_{k=0}^{n-1} \binom{-n}{k} (-1)^k D^k x^{n-1} = \frac{1}{n} \sum_{k=0}^{n-1} \binom{n+k-1}{k} \frac{1}{(n-1-k)!} x^{n-k} = \sum_{i=0}^n \binom{2n-i-1}{n-i}.$$

Thus, $C_n^{k*} = \frac{k}{n} \binom{2n-k-1}{n-k}$ for $1 \le k \le n$ (cf. the approach in [174]). Note that the Catalan numbers are related to the Bessel polynomials introduced by Krall and Frink (see [202, sect. 4.1.7, pp. 78-79]).

3.3 Discrete probability distributions

In this section we study discrete probability distributions that arise from a general construction with polynomials of convolution type. These distributions arise as conditional distributions and in approximation theory (references are given below). We will show how to use Umbral Calculus for computing moments of these distributions. The approaches of [74] and [241] are presented.

Let $(q_k)_{k \in \mathbb{N}}$ be a sequence of polynomials of convolution type with real coefficients. Fix n, α , and β such that

- 1. $n \ge 1$ 2. $q_n(\alpha + \beta) \ne 0$ 3. $(q_k(\alpha) q_{n-k}(\beta))/(q_n(\alpha + \beta) \ge 0$
- 4. $\alpha \neq 0$
- 5. $\beta \neq 0$

Denote by $P_n^{\alpha,\beta}$ the probability distribution on $\{0, 1, ..., n\}$ with

$$P_n^{\alpha,\beta}\{k\} = \frac{q_k(\alpha) q_{n-k}(\beta)}{q_n(\alpha+\beta)}$$
(3.1)

Examples 3.3.1 Examples of probability distributions of the form (3.1) include:

a) If $q_k(x) = \frac{x^k}{k!}$ and $\alpha, \beta \in \mathbb{N}$, then

$$P_n^{\alpha,\beta}\{k\} = \binom{n}{k} \left(\frac{\alpha}{\alpha+\beta}\right)^k \left(\frac{\beta}{\alpha+\beta}\right)^{n-k}$$

Hence, $P_n^{\alpha,\beta}$ is the binomial distribution with parameters n and $\alpha/(\alpha + \beta)$.

b) If $q_n(x) = \binom{x}{n}$ and $\alpha, \beta \in \mathbb{N}$, then

$$P_n^{\alpha,\beta}\{k\} = \frac{\binom{\alpha}{k}\binom{\beta}{n-k}}{\binom{\alpha+\beta}{n}}.$$

Hence, if α, β are positive integers, then $P_n^{\alpha,\beta}$ is the hypergeometric distribution with parameters n, α , and β .

c) If $q_n(x) = \binom{x+n-1}{n}$ and $\alpha, \beta \in \mathbb{N}$, then

$$P_n^{\alpha,\beta}\{k\} = \frac{\binom{\alpha+k-1}{k}\binom{\beta+n-k-1}{n-k}}{\binom{\alpha+\beta+n-1}{n}}.$$

Hence, $P_n^{\alpha,\beta}$ is the Pólya-Eggenberger distribution (see [128, Chapter 9, Section 4]).

d) If $q_n(x) = x (x - an)^{n-1}$ (a < 0) and $\alpha, \beta > 0$, then $P_n^{\alpha,\beta}$ is the quasibinomial distribution (see [65]).

We now calculate the first moment of the distribution $P_n^{\alpha,\beta}$ defined above. By definition, the first moment of $P_n^{\alpha,\beta}$ equals

$$\sum_{k=0}^{n} k P_n^{\alpha,\beta}\{k\} = \sum_{k=0}^{n} k \frac{q_k(\alpha) q_{n-k}(\beta)}{q_n(\alpha+\beta)}$$

Theorem 3.3.2 Let $P_n^{\alpha,\beta}$ be the probability distribution defined in (3.1). Then the first moment of $P_n^{\alpha,\beta}$ equals $n\alpha/(\alpha+\beta)$.

Proof: We give two proofs.

First proof: define the wide sense Sheffer sequence $(w_k)_{k\in\mathbb{N}}$ by $w_k(x) := k x^{-1} q_k(x)$ (see Example 2.4.10a). Applying Theorem 2.4.4b to $(w_k)_{k\in\mathbb{N}}$ yields $\sum_{k=0}^{n} k \alpha^{-1} q_k(\alpha) q_{n-k}(\beta) = (\alpha + \beta)^{-1} n q_n(\alpha + \beta)$. Hence, the first moment equals $n \alpha/(\alpha + \beta)$.

Second proof: define the linear operator T on \mathcal{P} by $Tq_m := \sum_{k=0}^m k q_k(\alpha) q_{m-k}$ for all $m \in \mathbb{N}$. Then the first moment of $P_n^{\alpha,\beta}$ equals $(Tq_n)(\beta)/q_n(\alpha+\beta)$. It follows from the Expansion theorem 2.2.22 that $T = \sum_{m=0}^{\infty} (Tq_m)(0) Q^m =$ $\sum_{m=0}^{\infty} m q_m(\alpha) Q^m$, which equals $\alpha E^{\alpha} g'(Q) Q$ by Corollary 2.3.2. We therefore have $(Tq_n)(\beta) = (\alpha E^{\alpha} \sum_{k=0}^n k g_k q_{n-k})(\beta)$. By Theorem 2.3.6e, the first moment of $P_n^{\alpha,\beta}$ equals $n \alpha/(\alpha+\beta)$.

Remark 3.3.3 The formula for the first moment of $P_n^{\alpha,\beta}$ is formula 17 of [140]. The proof in [140] uses formal generating functions. The proofs of Theorem 3.3.2 are therefore new proofs of this formula.

If $\sum_{n=0}^{\infty} g_n z^n$ has a positive radius of convergence, $g_n \ge 0$ for all $n \in \mathbb{N}$ and both α and β are non-negative real numbers, then a probabilistic proof of Theorem 3.3.2 is possible. Take $\theta > 0$ such that $\sum_{n=0}^{\infty} g_n \theta^n < \infty$. Let Xand Y be independent random variables with $P(X = k) = \theta^k q_k(\alpha) e^{-\alpha g(\theta)}$ and $P(Y = h) = \theta^h q_h(\beta) e^{\beta g(\theta)}$. It follows from Theorem 2.1.12d that $\sum_{k=0}^{\infty} P(X = k) = \sum_{k=0}^{\infty} P(Y = k) = 1$. Then

$$P(X = k \mid X + Y = n) = \frac{P(X = k) P(Y = n - k)}{P(X + Y = n)} = \frac{q_k(\alpha) q_{n-k}(\beta)}{q_n(\alpha + \beta)}.$$

Suppose α and β are rational. Then there exist $r, s \in \mathbb{N}$ and $M \in \mathbb{R}$ such that $\alpha = r/M$ and $\beta = s/M$. Define random variables $X_i(i = 1, \ldots, r + s)$ by $P(X_i = k) = \theta^k q_k(1/M) e^{-g(\theta)/M}$. The convolution property of the polynomials q_k implies that X has the same distribution as $X_1 + \ldots + X_r$ and that Y has the same distribution as $X_{r+1} + \ldots + X_{r+s}$. Since $E(X_1 + \ldots + X_{r+s} \mid X + Y = n) = E(X + Y \mid X + Y = n) = n$, we have $E(X \mid X + Y = n) = rn/(r+s) = \alpha n/(\alpha + \beta)$. The general case where α and β are real follows from a continuity argument.

A characterization of probability distributions of the type P(X = k | X + Y = n) can be found in [110]. There are examples of polynomials of convolution type such that $q_k(\alpha) q_{n-k}(\beta))/q_n(\alpha + \beta) = P(X = k | X + Y = n)$ yields known distributions. For the Abel polynomials $x (x - an)^{n-1}/n!$ see [65], for the Gould polynomials from Example 3.2.4b) (cf. [210, Section 12]), see [125]. For applications of these probability distributions, we refer to [64, 66, 67].

We can now calculate the second moment of $P_n^{\alpha,\beta}$. As is often the case with discrete distributions, it is easier to calculate descending factorial moments than moments (cf. [128, p. 19]). We use the idea of the second proof of Theorem 3.3.2.

Theorem 3.3.4 Let $P_n^{\alpha,\beta}$ be the probability distribution defined in 3.1. Then the second factorial moment of $P_n^{\alpha,\beta}$ equals

$$\frac{\alpha}{q_n(\alpha+\beta)} \left(E^{\alpha} Q^2 \left\{ g''(Q) + \alpha \left(g'(Q) \right)^2 \right\} q_n \right) (\beta).$$
(3.2)

Proof: Define the linear operator V on \mathcal{P} by $Vq_n := \sum_{k=0}^n k(k-1)q_k(\alpha)q_{n-k}$ for all $n \in \mathbb{N}$. Then the second descending factorial moment of $P_n^{\alpha,\beta}$ equals $((Vq_n)(\beta))/q_n(\alpha+\beta)$. It follows from the Expansion Theorem 2.2.22 that

$$V = \sum_{n=0}^{\infty} (V q_n)(0) Q^n = \sum_{n=0}^{\infty} n(n-1) q_n(\alpha) Q^n.$$

The formal generating formula 2.1.12d yields

$$\sum_{n=0}^{\infty} n(n-1) q_n(\alpha) z^n = z^2 \frac{d^2}{dz^2} e^{\alpha g(z)} = \alpha z^2 e^{\alpha g(z)} \left\{ g''(z) + \alpha \left(g'(z) \right)^2 \right\}.$$

It follows from the Isomorphism Theorem 2.3.1 that

$$V = \alpha Q^2 e^{\alpha g(Q)} \left\{ g''(Q) + \alpha \left(g'(Q) \right)^2 \right\}.$$

Since g(Q) = D by Theorem 2.2.22b, we have $e^{\alpha g(Q)} = E^{\alpha}$ by Example 2.2.8a. Putting everything together yields the result.

The following method of calculating moments is adapted from [241], where it is used in the context of approximation operators (cf. [182]). These operators are defined for continuous functions on [0, 1] by

$$(L_n f)(x) = \frac{1}{q_n(1)} \sum_{k=0}^n q_k(x) q_{n-k}(1-x) f(k/n),$$

where $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type with $q_n(x) \ge 0$ on [0,1]. For obvious reasons, it is important to calculate the action of these operators for $f(x) = x^m$ ($m \in \mathbb{N}$), which is nothing but computing the moments of the distribution defined by (3.1). More information on Umbral Calculus and approximation theory can be found in Subsection 3.7.2.

The Manole approach in [241] rests on the following lemmas. Note that Formula (3.3) only holds for $m \leq n$ (cf. [241]).

Lemma 3.3.5 ([241, Lemma 1]) Let Q be a delta operator with basic sequence $(q_n)_{n \in \mathbb{N}}$. Then we have for all $m \leq n$:

$$(PQ)^{m} = \sum_{k=0}^{n} S(m,k) P^{k} Q^{k}, \qquad (3.3)$$

where $P = x (Q')^{-1}$ and S(m,k) denotes the Stirling number of the second kind.

Proof: It suffices to check that both sides of (3.3) agree when applied to q_n for all $n \in \mathbb{N}$.

First note that by Theorem 2.3.6d, we have $Pq_n = (n+1)q_{n+1}$ for all $n \in \mathbb{N}$. Hence, $(PQ)^m q_n = n^m q_n$. By definition, the connection constants for expressing x^m in terms of the lower factorials $(x)_n = x(x-1)\dots(x-k+1)$ are the Stirling numbers of the second kind, i.e. $x^m = \sum_{k=0}^m S(m,k)(x)_k$. Since for $k \leq m \leq n$ we have $P^k Q^k q_n = P^k q_{n-k} = (n)_k q_n$, it follows that

$$(PQ)^m q_n = n^m q_n = \left(\sum_{k=0}^m S(m,k) (x)_k\right) q_n = \sum_{k=0}^m S(m,k) P^k Q^k q_n.$$

This concludes the proof, since n was arbitrary.

Lemma 3.3.6 ([241, Lemma 2]) Let $P_n^{\alpha,\beta}$ be the probability distribution defined in 3.1. Then the ℓ^{th} moments of $P_n^{\alpha,\beta}$ equals

$$\frac{1}{q_n(\alpha+\beta)} \sum_{k=0}^{\ell} S(\ell,k) P^k E^{\beta} q_n(\alpha+\beta).$$
(3.4)

Proof: Let T be the linear operator on \mathcal{P} defined by $T := x (Q')^{-1} Q$. It follows from Theorem 2.3.6d (Rodrigues Formula) that $T q_n = n q_n$. Hence,

$$\sum_{k=0}^{n} k^{\ell} q_{k}(\alpha) q_{n-k}(\beta) = \left(T^{\ell} \sum_{k=0}^{n} q_{n-k}(\beta) q_{k} \right) (\alpha) = \left(T^{\ell} E^{\beta} q_{n} \right) (\alpha).$$

Now note that T = PQ, where $P := x (Q')^{-1}$. Substituting Formula (3.3) into the last expression, we obtain the result.

Formula (3.4) is too general to be used for direct computations. The following theorem presents a simplification of the special m = 2 of Formula 3.4, which is suitable for computations. Note that P is not shift-invariant and does not commute with shift-invariant operators.

 $^{^{1}}$ There is a disagreement in notation for the lower factorials between combinatorics and special functions. We follow the convention of combinatorics.

Theorem 3.3.7 (Manole ([241])) Let $P_n^{\alpha,\beta}$ be the probability distribution defined in 3.1. Then the second moment of $P_n^{\alpha,\beta}$ equals

$$n^{2} \frac{\alpha}{\alpha+\beta} - \frac{\alpha\beta(Q')^{-2}q_{n+2}(\alpha+\beta)}{q_{n}(\alpha+\beta)}.$$
(3.5)

Proof: It follows from $x^2 = x + x(x - 1)$ that the Stirling numbers of the second kind satisfy S(2,0) = 0 and S(2,1) = S(2,2) = 1. In the rest of the proof we will repeatedly use the Rodrigues Formula (Theorem 2.3.6d). Since shift-invariant operators commute by Corollary 2.2.12, we may rewrite $P E^{\beta}$ as $x E^{\beta} (Q')^{-1}$ which yields

$$P E^{\beta} q_{n-j}(x) = (n-j+1) \frac{x}{x+\beta} q_{n-j+1}(x+\beta).$$
(3.6)

Writing $\frac{x}{x+\beta} = 1 - \frac{\beta}{x+\beta}$, we obtain the following convenient version of (3.6):

$$P E^{\beta} q_{n-j}(x) = (n-j+1) E^{\beta} q_{n-j+1}(x) - \beta E^{\beta} (Q')^{-1} q_{n-j}(x)$$
(3.7)

Using first (3.7) and then (3.7), we find that

$$P^{2} E^{\beta} q_{n-2}(x) = P\left((n-1) E^{\beta} q_{n-1}(x) - \beta E^{\beta} (Q')^{-1} q_{n-2}(x)\right)$$

= $n (n-1) \frac{x}{x+\beta} q_{n}(x+\beta) - x\beta E^{\beta} (Q')^{-2} q_{n-2}(x)$
= $n (n-1) \frac{x}{x+\beta} q_{n}(x+\beta) - x\beta (Q')^{-2} q_{n-2}(x+\beta).$

Substituting the above into (3.4), we obtain the desired result.

Examples 3.3.8 We now compute the second (factorial) moments of the probability distributions discussed in Examples 3.3.1.

a) For the binomial distribution with parameters n and $\alpha/(\alpha + \beta)$, we have $q_k(x) = \frac{x^k}{k!}$, Q = D and g(z) = z. Thus Formula (3.2) yields that the second descending factorial moment of $P_n^{\alpha,\beta}$ equals $n(n-1)(\alpha/(\alpha + \beta)))^2$.

b) For the hypergeometric distribution with parameters n, α and β , we have $q_k(x) = \binom{x}{k}$, $Q = E^1 - I$ and $g(z) = \log(1 + z)$. It follows that $g'(Q) = (I+Q)^{-1} = (E^1)^{-1} = E^{-1}$ and that $g''(Q) = -(I+Q)^{-2} = -(E^1)^{-2} = -E^{-2}$. Thus Formula (3.2) yields that the second descending factorial moment of $P_n^{\alpha,\beta}$ equals

$$\alpha \begin{pmatrix} \alpha + \beta \\ n \end{pmatrix}^{-1} \left(\left(-E^{\alpha - 2} + \alpha E^{\alpha - 2} \right) Q^2 \begin{pmatrix} x \\ n \end{pmatrix} \right) (\beta) = \alpha (\alpha - 1) \begin{pmatrix} \alpha + \beta \\ n \end{pmatrix}^{-1} \begin{pmatrix} \alpha + \beta - 2 \\ n - 2 \end{pmatrix} = n (n - 1) \frac{\alpha (\alpha - 1)}{(\alpha + \beta) (\alpha + \beta - 1)}.$$

c) For the Pólya-Eggenberger distribution , we have $q_k(x) = \binom{x+n-1}{k}, Q = I - E^{-1}$ and $g(z) = -\log(1-z)$. It follows that $g'(Q) = (I-Q)^{-1} = (E^{-1})^{-1} = E^1$

and that $g''(Q) = (I - Q)^{-2} = (E^1)^{-2} = E^{-2}$. Thus Formula (3.2) yields that the second descending factorial moment of $P_n^{\alpha,\beta}$ equals

$$n(n-1)\frac{\alpha(\alpha+1)}{(\alpha+\beta)(\alpha+\beta+1)}.$$

d) For the quasi-binomial distribution we have $q_k(x) = x (x - ak)^{k-1}/k!$ and $Q = D E^a$. Since there does not exist a closed formula for g(z), we cannot use Formula (3.2). By Lemma 2.3.4a, we have $Q' = e^{a D} (1 + a D) = E^a (1 + a D)$ and thus

$$(Q')^{-2} q_{n-2} = (Q')^{-1} (n-1) \frac{q_{n-1}}{x}$$

= $(Q')^{-1} \frac{(x-(n-1)a)^{n-2}}{(n-2)!}$
= $e^{-a D} \sum_{j=0}^{n-2} (-a)^j D^j \frac{(x-(n-1)a)^{n-2}}{(n-2)!}$
= $(-a)^{n-2} \sum_{i=0}^{n-2} \frac{(-a(x-na))^i}{i!}.$

Now Formula (3.5) yields that the second moment of $P_n^{\alpha,\beta}$ equals

$$n^2 \frac{\alpha}{\alpha+\beta} - n! \frac{\alpha \beta}{(\alpha+\beta) (\alpha+\beta-na)^{n-1}} (-a)^{n-2} \sum_{i=0}^{n-2} \frac{(-a (x-na))^i}{i!}.$$

The trivial relation $E(X(X-1)) = E(X^2) - E(X)$, the above formula immediately yields that the second factorial moment of $P_n^{\alpha,\beta}$ equals

$$n(n-1)\frac{\alpha}{\alpha+\beta} - n!\frac{\alpha\beta}{(\alpha+\beta)(\alpha+\beta-na)^{n-1}}(-a)^{n-2}\sum_{i=0}^{n-2}\frac{(-a(x-na))^i}{i!}$$

3.4 Orthogonal Sheffer polynomials

In this section we show how to use the Umbral Calculus for finding all orthogonal Sheffer polynomials². By Favard's Theorem (see e.g. [58, Theorem 4.4]), a polynomial sequence $(s_n)_{n \in \mathbb{N}}$ is orthogonal if and only if it satisfies the following three-term recurrence relation for $n \geq 0$:

$$s_{n+1}(x) = (a_n x - b_n) s_n(x) - c_n s_{n-1}(x), \qquad (3.8)$$

where $s_1 = 0$, s_0 is a non-zero constant, and $c_n a_n a_{n-1} > 0$ for $n \ge 1$. We must be careful with normalizations when dealing with Sheffer sequences, because if

 $^{^{2}}$ All Sheffer polynomials are orthogonal in the sense that they are orthogonal with respect to some functional (see [210, Section 9]).

 $(s_n)_{n\in\mathbb{N}}$ is Sheffer, then $(\lambda_n s_n)_{n\geq 0}$ need not be Sheffer. Hence, we must use the non-monic form (3.8).

The following theorem shows the relation between the three-term recurrence relation (3.8) and differential equations for the delta operator and shift-invariant operator of a Sheffer sequence. These differential equations are to be understood in the Pincherle sense (cf. Definition 2.3.3).

Theorem 3.4.1 Let $(s_n)_{n \in \mathbb{N}}$ be a Sheffer sequence with delta operator Q and invertible operator A. If $(s_n)_{n \in \mathbb{N}}$ satisfies the three-term recurrence relation (3.8), then

$$Q' = \frac{1}{a_0} I + \left(\frac{b_1}{a_1} - \frac{b_0}{a_0}\right) Q + \left(\frac{b_2}{a_2} - \frac{c_1}{a_1}\right) Q^2$$
(3.9)

$$A' = \frac{b_0}{a_0} A + \frac{c_1}{a_0} A Q.$$
 (3.10)

If conversely $Q' = d_1 + d_2Q + d_3Q^2$ and $A' = d_4A + d_5AQ$, then $(s_n)_{n \in \mathbb{N}}$ satisfies the following three-term recurrence relation:

$$(d_3 + d_5) s_{n-1} + d_2 s_n + d_1 s_{n+1}$$
$$s_{n+1}(x) = (a_n x - b_n) s_n(x) - c_n s_{n-1}(x)$$

Proof: We make extensive use of the Operator Expansion Theorem 2.4.11 and of the fact that $q_n(0) = 0$ for $n \ge 1$, where $(q_n)_{n \in \mathbb{N}}$ is the basic sequence of Q. In particular,

$$T' = \sum_{k=0}^{\infty} \left[T's_k \right]_{x=0} AQ^k$$

where T' is the Pincherle derivative of the operator T. Notice that

$$[T'p]_{x=0} = [Txp - xTp]_{x=0} = [Txp]_{x=0}.$$
(3.11)

It will be convenient to adopt the convention that $q_k = 0$ and $s_k = 0$ for k < 0. Assume that $(s_n)_{n \in \mathbb{N}}$ satisfies the three-term recurrence relation (3.8). We first apply (3.11) to T = A', which yields

$$\begin{aligned} A' &= \sum_{k=0}^{\infty} \left[Axs_k \right]_{x=0} A Q^k \\ &= \sum_{k=0}^{\infty} \frac{1}{a_k} \left[A \left(s_{k+1} + b_k s_k + c_k s_{k-1} \right) \right]_{x=0} A Q^k \\ &= \sum_{k=0}^{\infty} \frac{1}{a_k} \left(q_{k+1}(0) + b_k q_k(0) + c_k q_{k-1}(0) \right) A Q^k \\ &= \frac{b_0}{a_0} A + \frac{c_1}{a_0} A Q. \end{aligned}$$

We now apply (3.11) to T = (AQ)'.

$$\begin{aligned} AQ' + A'Q &= (AQ)' \\ &= \sum_{k=0}^{\infty} \left[(AQ)' s_k \right]_{x=0} A Q^k \\ &= \sum_{k=0}^{\infty} \left[AQ \, xs_k \right]_{x=0} A Q^k \\ &= \sum_{k=0}^{\infty} \frac{1}{a_k} \left[AQ \left(s_{k+1} + b_k s_k + c_k s_{k-1} \right) \right]_{x=0} A Q^k \\ &= \frac{1}{a_0} A + \sum_{k=1}^{\infty} \frac{1}{a_k} \left(q_k(0) + b_k q_{k-1}(0) + c_k q_{k-2}(0) \right) A Q^k \\ &= \frac{1}{a_0} A + \frac{b_1}{a_1} A Q + \frac{c_2}{a_2} A Q^2. \end{aligned}$$

Now we eliminate A' in the last equation by using the invertibility of A and Formula (3.10):

$$\begin{aligned} AQ' + A'Q &= \frac{1}{a_0} A + \frac{b_1}{a_1} AQ + \frac{c_2}{a_2} AQ^2 \\ AQ' &= \frac{1}{a_0} A + \left(\frac{b_1}{a_1} - \frac{b_0}{a_0}\right) AQ + \left(\frac{b_2}{a_2} - \frac{c_1}{a_1}\right) AQ^2 \\ Q' &= \frac{1}{a_0} I + \left(\frac{b_1}{a_1} - \frac{b_0}{a_0}\right) Q + \left(\frac{b_2}{a_2} - \frac{c_1}{a_1}\right) Q^2. \end{aligned}$$

Conversely, assume that $Q' = d_1 + d_2Q + d_3Q^2$ and $A' = d_4A + d_5AQ$. We apply the Polynomial Expansion Theorem 2.4.11a to xs_n , which yields

$$\begin{aligned} xs_n &= \sum_{k=0}^{n+1} \left[AQ^k xs_n \right]_{x=0} s_k \\ &= \sum_{k=0}^{n+1} \left[(AQ^k)'s_n \right]_{x=0} s_k \\ &= \sum_{k=0}^{n+1} \left\{ \left((A'Q^k) + k AQ^{k-1}Q' \right) s_n(0) \right\} s_k \\ &= \sum_{k=0}^{n+1} \left\{ A's_{n-k}(0) + AQ's_{n-k+1}(0) \right\} s_k \\ &= \sum_{k=0}^{n+1} \left\{ (d_4A + d_5AQ) s_{n-k}(0) + \left(d_1A + d_2AQ + d_3AQ^2 \right) s_{n-k+1}(0) \right\} s_k \end{aligned}$$

$$= \sum_{k=0}^{n+1} \{ d_4 q_{n-k}(0) + d_5 q_{n-k-1}(0) + d_1 q_{n-k+1}(0) + d_2 q_{n-k}(0) + d_3 q_{n-k-1}(0) \} s_k$$

= $\frac{d_4}{n!} s_n + \frac{d_5}{(n-1)!} s_{n-1} + \frac{d_1}{(n+1)!} s_{n+1} + \frac{d_2}{n!} s_n + \frac{d_3}{(n-1)!} s_{n-1}$
= $(d_3 + d_5) s_{n-1} + d_2 s_n + d_1 s_{n+1}$

Since $d_1 \neq 0$, this means that $(s_n)_{n \in \mathbb{N}}$ satisfies a three-term recurrence relation of the form (3.8).

Solving the differential equations for Q and A, we find all orthogonal Sheffer polynomials. Thus, we have a new proof of the Meixner classification of orthogonal Sheffer polynomials (see [157] for the original proof, cf. [5, 57, 100, 132, 141, 206, 202]). The advantage of our proof is that it is a constructive proof based on first principles of the Umbral Calculus. A generalization of the Meixner classification was obtained by Al-Salam in [6], where it is shown that the Meixner result remains true even if we consider the more general class of polynomials with generating function $e^{Q(x,t)}$, where Q(x,t) is a polynomial in x and a power series in t.

It is an open problem to determine which Sheffer sequences are orthogonal with respect to a Borel measure in the complex plane, cf. [210, p. 751]. Two such sequences are the polynomials $x^n/n!$, which are orthogonal with respect to arc length on the unit circle, and the lower factorial polynomials (see [153] and references therein). All Sheffer sequences orthogonal on the unit circle have been classified by Kholodov (see [132]).

3.5 Moment systems

In the work of Feinsilver (e.g. [89, 90, 91]; see also [117]) sequences $(p_n)_{n \in \mathbb{N}}$ of polynomials appear that satisfy

$$p_n(t) = \int_{-\infty}^{\infty} x^n \, d\mu_t(x), \qquad (3.12)$$

where $(\mu_t)_{t\geq 0}$ is a convolution semigroup of probability measures (usually induced by a stochastic process with stationary independent increments). It follows directly from the Binomial Formula that Formula (3.12) implies

$$p_n(x+y) = \sum_{k=0}^n \binom{n}{k} p_k(x) p_{n-k}(y).$$
(3.13)

In this section we will study when a sequence $(p_n)_{n \in \mathbb{N}}$ satisfying (3.13) admits a representation of the form (3.12). We will show that representations of the form (3.12) are related to umbral operators (cf. Section 2.3) and to moment sequences of infinitely divisible probability measures. Moreover, the relation between umbral composition and subordination of probability measures is pointed out. Representations of the form (3.12) by groups of complex Borel measures or by convolution semigroups of probability measures with support in $[0, \infty)$ are also studied. At the end of this section we present some explicit examples.

This section is based on [75].

Definition 3.5.1 Let $(\mu_t)_{t\geq 0}$ be a collection of complex Borel measures on the real line.

If $\mu_s * \mu_t = \mu_{s+t}$ for all $s, t \ge 0$ (where * denotes convolution), then $(\mu_t)_{t\ge 0}$ is a said to be a convolution semigroup.

If $\int_{-\infty}^{\infty} f(x) d\mu_t(x)$ is a measurable function of t for each bounded continuous function f on the real line, then $(\mu_t)_{t\geq 0}$ is said to be weakly measurable.

If $\int_{-\infty}^{\infty} f(x) d\mu_t(x)$ converges to $\int_{-\infty}^{\infty} f(x) d\mu_0(x)$ for each continuous function f on the real line with compact support and $\mu_t(\mathbb{R})$ converges to $\mu_0(\mathbb{R})$ as t goes to zero, then $(\mu_t)_{t>0}$ is said to be weakly continuous.

Lemma 3.5.2 Let μ and ν be probability measures on the real line. If $\mu * \nu$ has finite moments of all orders, then both μ and ν have finite moments of all orders.

Proof: The definition of Lebesgue integrals implies that $\mu * \nu$ has finite *absolute* moments of all orders. Let r be a positive integer. By the definition of convolution,

$$\int_{-\infty}^{\infty} |t|^r d(\mu * \nu)(t) = \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} |x+y|^r d\mu(x) d\nu(y) < \infty.$$

Hence, $\int_{-\infty}^{\infty} |x+y|^r d\mu(x) < \infty$, μ -a.e. in y. Since

$$|x|^r \le 2^r |x+y|^r + 2^r |y|^r$$

it follows that

$$\int_{-\infty}^{\infty} |x|^r \ d\mu(x) < \infty.$$

Likewise we see that ν has a finite absolute moment of order r. \Box

The following theorem shows that convolution semigroups of measures generate umbral operators (cf. Definition 2.3.9).

Theorem 3.5.3 Let $(\mu_t)_{t\geq 0}$ be a weakly measurable convolution semigroup of complex Borel measures on the real line having finite moments of all orders and let $(p_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type. If $\mu_t(\mathbb{R}) = 1$ for all nonnegative t, then $(q_n)_{n\in\mathbb{N}}$ is also a sequence of polynomials of convolution type, where

$$q_n(t) = \int_{-\infty}^{\infty} p_n(x) \, d\mu_t(x).$$

Proof: The sequence $(q_n)_{n \in \mathbb{N}}$ is well-defined since $(\mu_t)_{t \ge 0}$ has finite moments of all orders. We verify that $(q_n)_{n \in \mathbb{N}}$ is of convolution type:

$$q_n(s+t) = \int_{-\infty}^{\infty} p_n(x) \ d\mu_{s+t}(x) = \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} p_n(x+y) \ d\mu_s(x) \ d\mu_t(y) = \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \sum_{k=0}^{n} p_k(x) \ p_{n-k}(y) \ d\mu_s(x) \ d\mu_t(y) = \sum_{k=0}^{n} q_k(s) \ q_{n-k}(t).$$

If $p_0(0) = 0$, then $p_n = 0$ for all n and there is nothing to prove. If $p_0(0) \neq 0$, then $p_0 = 1$ and we have $q_0 = 1$ because $\mu_t(\mathbb{R}) = 1$. Moreover, since $(\mu_t)_{t\geq 0}$ is weakly measurable, the functions q_n are Borel measurable. It now follows from Theorem 2.1.8 that $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials. \Box

Remark 3.5.4 The linear operator U, defined by $Up(t) := \int_{-\infty}^{\infty} p(x) d\mu_t(x)$, is an umbral operator in the sense of Definition 2.3.9. We will see in Corollary 3.5.11 which umbral operators can be represented in this way.

The following two theorems describe necessary and sufficient conditions for the existence of convolution semigroups of probability measures.

Definition 3.5.5 A complex-valued function f on the real line is negative definite if

$$\sum_{k=1}^{n} c_k = 0 \text{ implies } \sum_{j,k=1}^{n} c_j \bar{c}_k f(s_j - s_k) \leq 0$$

for all nonnegative sequences s_1, \ldots, s_n and all complex sequences c_1, \ldots, c_n .

A real-valued C^{∞} -function f on $(0,\infty)$ is said to be a Bernstein function if

 $f \geq 0$ and $(-1)^k D^k f \geq 0$ for all integers $k \geq 1$

Theorem 3.5.6 If a continuous real-valued function f on the real line is negative definite and satisfies f(0) = 0, then there exists a weakly continuous semigroup $(\mu_t)_{t\geq 0}$ of probability measures on the real line such that for all real y

$$e^{-tf(y)} = \int_{-\infty}^{\infty} e^{-ixy} \, d\mu_t(x)$$

Proof: This follows from [24, Chapter 2, Theorem 8.3 and Corollary 8.6]. \Box .

The following theorem is the analogue of Theorem 3.5.6 for convolution semigroups of probability measures on $[0, \infty)$.

Theorem 3.5.7 If a real-valued function f on $(0, \infty)$ is a Bernstein function and satisfies f(0) = 0, then there exists a weakly continuous semigroup $(\mu_t)_{t\geq 0}$ of probability measures on $[0, \infty)$ such that for all positive y

$$e^{-tf(y)} = \int_0^\infty e^{-xy} d\mu_t(x)$$

Proof: See [24, Chapter 2, Theorem 9.18 and Remark 9.19].

We now investigate which sequences of polynomials of convolution type admit a representation of the form (3.12), i.e. can be represented as moment systems in the terminology of Feinsilver (see [89, 91, 117]). Our first theorem relates negative definiteness of -g to representability as moment system.

Theorem 3.5.8 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n \in \mathbb{N}}$. Suppose that

- 1. $\sum_{n=0}^{\infty} g_n z^n$ has a positive radius of convergence
- 2. the function $x \mapsto -g(-ix)$ has a continuous, negative definite extension to the real line.

Then there exists a weakly continuous convolution semigroup $(\mu_t)_{t\geq 0}$ of probability measures on the real line such that for $t\geq 0$

$$q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} \ d\mu_t(x)$$

Proof: It follows from $g_0 = 0$, condition 2) and Theorem 3.5.6 that there exists a weakly continuous semigroup $(\mu_t)_{t\geq 0}$ of probability measures such that

$$\int_{-\infty}^{\infty} e^{-ixy} d\mu_t(x) = e^{t g(-iy)}$$

for all real y. It follows from condition 1) that there exists r > 0 such that g is analytic for |z| < r. Thus by [151, Theorem 7.1.1]

$$\int_{-\infty}^{\infty} e^{zx} d\mu_t(x) = e^{t g(z)}$$

for |z| < r. Since g is analytic, it follows from [151, Corollary 1 to Theorem 2.3.1] that each μ_t has finite moments of all orders. Moreover, [151, Corollary 2 to Theorem 2.3.1]) yields

$$e^{t g(z)} = \int_{-\infty}^{\infty} e^{zx} d\mu_t(x) = \sum_{n=0}^{\infty} \left(\int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x) \right) z^n$$

for |z| < r. It now follows from Theorem 2.1.12d that $q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x)$.

We will see in Theorem 4.4.10 that condition 1) is equivalent to: there exists r > 0 such that for all t > 0

$$\sum_{n=0}^{\infty} |q_n(t)| r^n < \infty.$$

Of course, there is a corresponding result for probability measures on $[0,\infty)$.

Theorem 3.5.9 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(q_n)_{n \in \mathbb{N}}$. Suppose that

- 1. $\sum_{n=0}^{\infty} g_n z^n$ has a positive radius of convergence
- 2. the function $x \mapsto -g(-x)$ has an extension to a Bernstein function on $(0,\infty)$.

Then there exists a weakly continuous convolution semigroup $(\mu_t)_{t\geq 0}$ of probability measures on $[0,\infty)$ such that for $t\geq 0$

$$q_n(t) = \int_0^\infty \frac{x^n}{n!} \ d\mu_t(x)$$

Proof: The proof is analogous to the proof of Theorem 3.5.8 (use Theorem 3.5.7 instead of Theorem 3.5.6). \Box

A sequence $(q_n)_{n \in \mathbb{N}}$ of polynomials of convolution type is determined by the numbers $q_n(1)$ by Theorem 2.1.14. The following theorem gives a necessary and sufficient condition on the numbers $n! q_n(1)$ for the representation of Theorem 3.5.8 to hold.

Theorem 3.5.10 Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type. There exists a weakly continuous convolution semigroup $(\mu_t)_{t\geq 0}$ of probability measures on the real line such that $q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x)$ for $t \geq 0$ if and only if $(n!q_n(1))_{n\in\mathbb{N}}$ is the moment sequence of an infinitely divisible probability measure on the real line.

Proof: " \Rightarrow " This follows from $n! q_n(1) = \int_{-\infty}^{\infty} x^n d\mu_1(x)$, since μ_1 is clearly infinitely divisible.

" \Leftarrow " Let μ be an infinitely divisible probability measure with moment sequence $(n! q_n(1))_{n \in \mathbb{N}}$. By [93, Chapter 9, Section 5, Theorem 2], there exists a weakly continuous convolution semigroup of probability measures $(\mu_t)_{t\geq 0}$ on the real line such that $\mu_1 = \mu$. It follows from Lemma 3.5.2 that each probability measure μ_t has finite moments of all orders. Thus Theorem 3.5.3 implies that the sequence $(h_n)_{n\in\mathbb{N}}$, defined by $h_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x)$ is a sequence of polynomials of convolution type. This sequence is determined by the numbers $h_n(1)$ by Theorem 2.1.14. Hence, $h_n = q_n$ for all n, since $h_n(1) = q_n(1)$.

As a corollary we now describe when an umbral operator (cf. Definition 2.3.9) can be represented as an integral operator. This representation differs from the integral operator representation for shift-invariant operators in Section 3.6, since umbral operators are *never* shift-invariant (except for the identity operator) by Theorem 2.3.11b.

Corollary 3.5.11 Let U be an umbral operator. Then there exists a semigroup $(\mu_t)_{t>0}$ of probability measures on the real line such that

$$(Up)(t) = \int_{-\infty}^{\infty} p(x) \ d\mu_t(x)$$

if and only if $((Ux^n)(1))_{n \in \mathbb{N}}$ is the moment sequence of an infinitely divisible probability measure.

Proof: Define $q_n = U \frac{x^n}{n!}$. Since U is an umbral operator, it follows that $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type. The result now follows from Theorem 3.5.10.

The following theorem relates umbral operators to subordination of convolution semigroups of probability measures. For another relation between umbral operators and subordination, see [222].

Theorem 3.5.12 Let $(\mu_t)_{t\geq 0}$ and $(\nu_t)_{t\geq 0}$ be weakly measurable convolution semigroups of probability measures and let $(p_n)_{n\in\mathbb{N}}$ and $(q_n)_{n\in\mathbb{N}}$ be the associated sequences of polynomials of convolution type, i.e. $p_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x)$ and $q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\nu_t(x)$. Let U be the umbral operator that maps $\frac{x^n}{n!}$ to p_n and define polynomials r_n by $r_n = Uq_n$. Then

$$r_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} \, d\rho_t(x)$$

where $(\rho_t)_{t\geq 0}$ is the convolution semigroup subordinated to $(\nu_t)_{t\geq 0}$ by means of $(\mu_t)_{t\geq 0}$.

Proof: This follows from

$$r_n(t) = Uq_n(t)$$

= $\int_{-\infty}^{\infty} q_n(x) d\mu_t(x)$
= $\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \frac{y^n}{n!} d\nu_x(y) d\mu_t(x)$

where $\int_{-\infty}^{\infty} d\mu_t(x) \nu_x$ is the probability measure resulting from subordinating to $(\nu_t)_{t\geq 0}$ by means of $(\mu_t)_{t\geq 0}$ (cf. [24, Section 9.20]).

The following theorem states when a sequence of polynomials of convolution type is generated by a *group* of complex Borel measures on the real line.

Definition 3.5.13 A group $(\mu_t)_{t \in \mathbb{R}}$ of probability measures on the real line is said to be strongly continuous if the operators $f \mapsto f * \mu_t$ form a strongly continuous group on $L^1(-\infty, \infty)$.

A complex Borel measure μ on the real line is said to be **invertible**, if there exists a complex Borel measure ν on the real line such that $\mu * \nu = \delta_0$, where δ_0 is the point mass at 0.

Theorem 3.5.14 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type. There exists a strongly continuous group of complex Borel measures $(\mu_t)_{t \in \mathbb{R}}$ such that

$$q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} \ d\mu_t(x)$$

if and only if $(n!q_n(1))_{n\in\mathbb{N}}$ is the moment sequence of an invertible complex Borel measure on the real line with total mass one.

Proof: ' \Rightarrow ' This follows from $n! q_n(1) = \int_{-\infty}^{\infty} x^n d\mu_1(x)$. Note that μ_1 is invertible, since $(\mu_t)_{t \in \mathbb{R}}$ is a group.

'⇐' Let μ be an invertible complex Borel measure with moment sequence $(n! q_n(1))_{n \in \mathbb{N}}$. It follows from [95] there exists a strongly continuous convolution group of complex Borel measures $(\mu_t)_{t \in \mathbb{R}}$ on the real line such that $\mu_1 = \mu$ and $\mu_t(\mathbb{R}) = 1$. It follows from Theorem 3.5.3 that $(h_n)_{n \in \mathbb{N}}$, defined by $h_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x)$, is a sequence of polynomials of convolution type. This sequence is determined by the numbers $h_n(1)$ by Theorem 2.1.14. Hence, $h_n = q_n$ for all n, since $h_n(1) = q_n(1)$.

We conclude this section with examples of sequences of polynomials of convolution type that are moment systems.

Examples 3.5.15 Explicit examples of sequences of polynomials of convolution type that are moment systems include:

1. Take $\mu_t = \delta_t$. Then

$$q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\delta_t(x) = \frac{t^n}{n!}$$

2. Take $\mu_t = \sum_{k=0}^{\infty} e^{-t} \frac{t^k}{k!} \delta_k$ (Poisson semigroup). Then

$$q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x) = \frac{1}{n!} \sum_{k=0}^{\infty} e^{-t} \frac{t^k}{k!} k^n = \frac{e^{-t}}{n!} \sum_{k=0}^{\infty} \frac{t^k k^n}{k!}$$

The ratio test yields that this series converges absolutely for all real t. The series

$$e^{-t}\sum_{k=0}^{\infty}\frac{t^kk^n}{k!}$$

is known as the Dobinski Formula for the exponential polynomials (see [202, p. 66] and [152]).

3. Take $d\mu_t(x) = 1_{(0,\infty)}(x) \frac{1}{\Gamma(t)} x^{t-1} e^{-x} dx$ (Gamma-semigroup). Then

$$q_n(t) = \int_{-\infty}^{\infty} \frac{x^n}{n!} d\mu_t(x) = \int_0^{\infty} \frac{x^{n+t-1}}{n! \, \Gamma(t)} \, e^{-x} dx = \frac{\Gamma(n+t)}{\Gamma(t) \, n!} = \binom{t+n-1}{n}$$

4. Take $d\mu_t(x) = \frac{1}{\sqrt{2\pi t}} e^{-x^2/2t} dx$ (Brownian semigroup).

Then $q_n(t) = 0$ if n is odd and for n even we have

$$q_n(t) = \frac{1}{\sqrt{2\pi t}} \int_{-\infty}^{\infty} \frac{x^n}{n!} \ e^{-x^2/2t} \ dx = \frac{t^{n/2}}{2^{n/2} \ (n/2)!}$$

Note that the degree of q_n is less than n.

3.6 Shift-invariant operators and integral operators.

Theorems 2.2.7, 2.2.22 and 2.4.11b show that shift-invariant operators can be represented by power series. In this section we prove that each shift-invariant operator can be represented as a random shift, i.e. as an integral operator. This representation will be used to give a new proof of a characterization theorem for Sheffer polynomials (due to Sheffer) which characterizes Sheffer sequences as shifted moments of a complex Borel measure on the real line. This section is based on [75].

The following lemma is essential.

Lemma 3.6.1 (Boas) For each sequence $(a_n)_{n \in \mathbb{N}}$ of complex numbers, there exist infinitely many complex Borel measures μ on the real line such that

$$a_n = \int_{-\infty}^{\infty} y^n \, d\mu(y).$$

(Pólya) Among these measures are discrete measures and absolutely continuous measures.

Proof: Boas and Pólya stated their results for real sequences (see [27] and [181]). Our lemma follows immediately from their results by considering real and imaginary parts. $\hfill \Box$

If μ is a complex Borel measure on the real line having finite moments of all orders, then

$$(Tp)(x) = \int_{-\infty}^{\infty} p(x+y) \, d\mu(y)$$

defines a linear shift-invariant operator on the space of polynomials. The converse is also true as the following theorem shows (cf. [223, formula 13].

Theorem 3.6.2 If T is a linear shift-invariant operator, then there exists a complex Borel measure μ on the real line such that for all polynomials p

$$(Tp)(x) = \int_{-\infty}^{\infty} p(x+y) \, d\mu(y).$$
Proof: By the Expansion Theorem 2.2.7, we have

$$T = \sum_{k=0}^{\infty} \left(T \ \frac{x^k}{k!} \right) (0) D^k.$$

By Lemma 3.6.1, there exists a complex Borel measure μ on the real line such that $(T x^k) (0) = \int_{-\infty}^{\infty} y^k d\mu(y)$. Define the linear shift-invariant operator V by $(Vp)(x) = \int_{-\infty}^{\infty} p(x+y) d\mu(y)$ for all polynomials p. Then

$$\left(V\frac{x^k}{k!}\right)(0) = \int_{-\infty}^{\infty} \frac{y^k}{k!} d\mu(y) = \left(T\frac{x^k}{k!}\right)(0.)$$

It follows from the Expansion Theorem 2.2.7 that T = V.

Note that if $T \neq 0$ is a non-invertible shift-invariant operator, then there does not exist a non-negative Borel measure such that $(Tp)(x) = \int_{-\infty}^{\infty} p(x+y) d\mu(y)$ for all polynomials p. Indeed, this would imply T1 = 0, since $T1 \neq 0$ is equivalent to invertibility. Hence, $\mu(\mathbb{R}) = 0$. Since $\mu \neq 0$ it follows that μ cannot be a non-negative Borel measure.

Examples 3.6.3 We present some explicit examples of representations of linear shift-invariant operators. Let δ_0 be the point mass at 0. Recall that the measure μ of Theorem 3.6.2 is not unique.

Linear shift-invariant operator	Measure μ of Theorem 3.6.2
Identity operator	δ_0
Laguerre operator	$\delta_0 - e^{-t} dt$
Weierstrass operator	$\frac{1}{\sqrt{2\pi}} e^{-t^2/2} dt$

The Weierstrass operator is the invertible operator of the Hermite polynomials (cf. Example 2.4.3a.

Remark 3.6.4 The proof of Theorem 3.6.2 show that the existence of absolutely continuous measures for the identity operator, the differentiation operator, and the shift operators E^a , are equivalent to the following moment problems:

1. identity operator:
$$\int_{-\infty}^{\infty} d\mu(y) = 1, \int_{-\infty}^{\infty} y^n d\mu(y) = 0 \text{ for } n > 0.$$

2. differentiation operator:
$$\int_{-\infty}^{\infty} y d\mu(y) = 1, \int_{-\infty}^{\infty} y^n d\mu(y) = 0 \text{ for } n \neq 1.$$

3. shift-operator E^a :
$$\int_{-\infty}^{\infty} y^n d\mu(y) = a^n.$$

As Erik Thomas showed to me (private communication), it is possible to solve these moment problems explicitly using the theory of tempered distributions.

We now use Theorem 3.6.2 to prove a characterization theorem for Sheffer polynomials due to Sheffer (see [218, Theorem 2]).

Theorem 3.6.5 (Sheffer) A sequence $(s_n)_{n \in \mathbb{N}}$ of polynomials is a strict sense Sheffer sequence with basic sequence $(q_n)_{n \in \mathbb{N}}$ if and only if there exists a complex Borel measure μ on the real line such that $\mu(\mathbb{R}) \neq 0$ and

$$s_n(x) = \int_{-\infty}^{\infty} q_n(x+y) \, d\mu(y)$$

for all $n \in \mathbb{N}$.

Proof: ' \Leftarrow ' The operator A, defined by $(Ap)(x) = \int_{-\infty}^{\infty} p(x+y) d\mu(y)$ is an invertible linear shift-invariant operator on \mathcal{P} . Clearly $s_n = Aq_n$ and hence, deg $s_n = n$. Moreover, $s_0 \neq 0$ since $q_0 = 1$ and $\mu(\mathbb{R}) \neq 0$. Thus $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence by Theorem 2.4.8b.

'⇒' If $(s_n)_{n \in \mathbb{N}}$ is a strict sense Sheffer sequence, then the linear operator A, defined by $Aq_n = s_n$, is shift-invariant by Theorem 2.4.8b. Thus Theorem 3.6.2 yields a complex Borel measure μ such that $(Ap)(x) = \int_{-\infty}^{\infty} p(x+y) d\mu(y)$ for all polynomials p. In particular, $s_n(x) = Aq_n(x) = \int_{-\infty}^{\infty} q_n(x+y) d\mu(y)$ for all n. Moreover, $\mu(\mathbb{R}) = s_0(0) \neq 0$. □

The formal moment generating function of the measure μ in Theorem 3.6.5 is equal to the formal power series $\sum_{n=0}^{\infty} s_n(0) t^n$ (see [218, Corollary on p. 742]). The following integral representation for Hermite polynomials

$$H_n(x) = \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\infty} \frac{(x+y)^n}{n!} e^{-y^2/2} \, dy$$

is an illustration of Theorem 3.6.5.

We now show that linear functionals can be represented by integrals (cf. Theorem 2.4.8).

Theorem 3.6.6 Let Λ be a linear functional on \mathcal{P} . Then there exists a complex Borel measure μ on the real line such that

$$\Lambda p = \int_{-\infty}^{\infty} p(x) \, d\mu(x)$$

for all polynomials p.

Proof: Define $(a_n)_{n\in\mathbb{N}}$ by $\Lambda \frac{x^n}{n!} := a_n$. Define the linear operator T on \mathcal{P} by $T := \sum_{n=0}^{\infty} a_n D^n$. Since $(T \frac{x^n}{n!})(0) = a_n$ for all $n \in \mathbb{N}$, we conclude that $\Lambda p = (Tp)(0)$ for all $p \in \mathcal{P}$. The theorem now follows from Theorem 3.6.2

The above theorem can be used to prove another characterization theorem for Sheffer polynomials. This theorem was proved by Thorne for Appell polynomials (see [236]) and extended by Sheffer to Sheffer polynomials (see [218, p. 744]).

Theorem 3.6.7 (Thorne-Sheffer) A sequence $(s_n)_{n \in \mathbb{N}}$ of polynomials such that deg $s_n = n$ is a Sheffer sequence if and only if there exist a delta operator Q and a complex Borel measure μ on the real line such that $\mu(\mathbb{R}) \neq 0$, μ has finite moments of all orders, and

$$\int_{-\infty}^{\infty} \left(Q^k \, s_n \right)(x) \, d\mu(x) = \delta_{kn}$$

Proof: ' \Rightarrow ' By Theorem 2.4.8c, there exist an invertible linear functional Λ (i.e., $\Lambda 1 \neq 0$) and a delta operator Q such that $\Lambda Q^k s_n = \delta_{nk}$. By Theorem 3.6.6, there exists a complex Borel measure μ on the real line such that $\Lambda p = \int_{-\infty}^{\infty} p(y) d\mu(y)$. Hence,

$$\int_{-\infty}^{\infty} \left(Q^k s_n \right) (x) \, d\mu(x) = \delta_{kn}.$$

'⇐' This follows directly from Theorem 2.4.8c, since $p \mapsto \int_{-\infty}^{\infty} p(x) d\mu(x)$ is an invertible linear functional on \mathcal{P} .

3.7 Exponential families

Exponential families of probability measures play a traditional role in statistics (dating back to the thirties) because of their nice estimation properties (see e.g. [221]). However, recently exponential families appear as the cornerstone of the important class of generalized linear models (see [78] for an excellent introduction). In [160], Morris studied natural exponential families on the real line. He showed that there are six classes of natural exponential families with quadratic variance function (i.e. where the variance is a polynomial of degree at most two). In this section we study natural exponential families in light of expansions of their density function in terms of Sheffer polynomials. The delta operator of the associated Sheffer sequence will be shown to relate directly to the variance function of natural exponential family. Using slightly different terminology, Feinsilver proved [89, Chapter 4] that a natural exponential family has a quadratic variance function if and only if the corresponding Sheffer polynomials are orthogonal. This result immediately follows from our approach to natural exponential families (cf. Section 3.4). It is interesting to note that the Morris classification was discovered a few years earlier in approximation theory by May (see [156] and for generalizations [124]). We discuss the relation between exponential families and exponential approximation operators. We also indicate how our approach differs from the approach in [124, 156].

This section is based on [77].

3.7.1 Natural exponential families

We begin by recalling the definition of a natural exponential family. Our notation closely follows [144].

Let ν be a measure on the real line. We assume that ν is not concentrated in one point.

Let the Laplace transform of ν be given by

$$L(\theta) = \int_{-\infty}^{\infty} e^{x\theta} d\nu(x).$$
 (3.14)

We define Θ to be the interior of the set $\{\theta \in \mathbb{R} \mid L(\theta) < \infty\}$. If Θ is non-empty, then the **natural exponential family** generated by ν is the set of probability distributions of the form

$$P_{\theta}(A) = \int_{A} e^{x\theta - k(\theta)} d\nu(x), \qquad (3.15)$$

where k is the **cumulant** of ν , i.e. $k(\theta) := \log L(\theta)$, and $\theta \in \Theta$. We will see later that different ν may generate the same natural exponential family. Since $k(\theta) = \log L(\theta)$, we have

$$e^{k(\theta)} = \int_{-\infty}^{\infty} e^{x\theta} \, d\nu(x). \tag{3.16}$$

It follows by differentiating (3.16) with respect to θ that

$$k'(\theta) = \int_{-\infty}^{\infty} x \, dP_{\theta}(x). \tag{3.17}$$

Differentiating (3.16) twice with respect to θ and using (3.17), we obtain

$$k''(\theta) = \int_{-\infty}^{\infty} (x - k'(\theta))^2 dP_{\theta}(x).$$
(3.18)

Let M_{ν} be the range of k', i.e. $M_{\nu} = k'(\Theta)$. Since k is strictly convex on Θ by the Hölder inequality³, $k' : \Theta \longrightarrow M_{\nu}$ is a bijection. Its inverse will be denoted by

$$\psi: M_{\nu} \to \Theta$$

This means that we may reparametrize the densities with respect to ν in (3.15) as

$$\varphi(m,x) = e^{x\psi(m)} - k(\psi(m)). \tag{3.19}$$

Using the reparametrization of (3.19), we now come to the following important definition.

³If
$$0 < \lambda^{-} < 1$$
, then $k(\lambda\theta + (1 - \lambda)\xi) = \log\left(\int_{-\infty}^{\infty} e^{\lambda\theta x} e^{(1 - \lambda)\xi x} d\nu(x)\right) < \log\left(\left(\int_{-\infty}^{\infty} e^{\theta x} d\nu(x)\right)^{\lambda} \left(\int_{-\infty}^{\infty} e^{\xi x} d\nu(x)\right)^{1-\lambda}\right) = \lambda k(\theta) + (1 - \lambda)k(\xi)$. Note that the inequality is strict, since ν is not concentrated in one point.

Definition 3.7.1 Let $\{P_{\theta} | \theta \in \Theta\}$ be the natural exponential family generated by a measure ν . The function $V_{\nu} : M_{\nu} \to \mathbb{R}$ defined by $V_{\nu}(m) = \int_{-\infty}^{\infty} (x - m)^2 \varphi(m, x) d\nu(x)$ is called the variance function of $\{P_{\theta} | \theta \in \Theta\}$.

A natural exponential family is uniquely determined by its variance function together with the domain of the variance function ([160]). In the theory of generalized linear models, the variance function is called the **link function** ([78]). The link function is essential for estimating purposes.

Before we continue, we give an example in order to illustrate the notions introduced above.

Example (Poisson family) Consider a Poisson distribution with parameter θ , i.e. $\Pr(k) = \frac{e^{-\theta} \theta^n}{n!}$ for $n = 0, 1, 2, \ldots$ Writing $\frac{e^{-\theta} \theta^n}{n!} = \frac{e^{n \log \theta}}{n! e^{e^{\log \theta}}}$, we see that $\{P_{\theta} \mid \theta \in (0, \infty)\}$ is a natural exponential family generated by the discrete measure $\nu\{n\} = 1/n!, n = 0, 1, 2, \ldots$ where P_{θ} is Poisson(log θ) distributed. An easy calculation shows that $k(\theta) = e^{\theta}, \Theta = \mathbb{R}, \psi(m) = \log m, M_{\nu} = (0, \infty)$, and $V_{\nu}(m) = m$. We see that the standard change from θ to the so-called natural parameter log θ (cf. [78] is nothing but our reparametrization (3.19). The following lemma is crucial to our approach.

Lemma 3.7.2 If $(P_{\theta} \mid \theta \in \Theta)$ is a natural exponential family, then there exist a real number t and a natural exponential family $\{\widetilde{P}_{\theta} \mid \theta \in \widetilde{\Theta}\}$ generated by a measure μ such that

- 1. $P_{\theta}(A) = \widetilde{P}_{\theta}(A+t)$ 2. $\int_{-\infty}^{\infty} x \, d\mu(x) = 0$ 3. $0 \in \widetilde{\Theta}$ 4. $\widetilde{k}'(0) = 0$
- 5. $V_{\mu}(m) = V_{\nu}(m+t)$.

Proof: First note that $\{P_{\theta} | \theta \in \Theta\}$ is also generated by the measure $e^{\theta_0 x} d\nu(x)$ for any $\theta_0 \in \Theta$ and that the corresponding parameter set $\widetilde{\Theta}$ equals $\Theta - \theta_0$. In particular, $0 \in \widetilde{\Theta}$. Now define the measure μ by $d\mu(x) = d\nu(x + k'(0))$. It follows from (3.17) that $\tilde{k}'(0) = 0$. Moreover, easy calculations shows that (1) and (5) hold with t = k'(0).

Example (Poisson family continued) Let μ be the measure obtained by shifting the generating measure ν one unit (=k'(0)) to the left, i.e. $\mu\{n\} = 1/(n+1)!$, n = -1, 0, 1, 2, ... An easy calculation yields that μ is of mean zero, $\tilde{\Theta} = \mathbb{R}$, $\tilde{k}'(\theta) = e^{\theta} - 1$, $M_{\mu} = (-1, \infty)$, $\tilde{k}'(0) = 0$, and $V_{\mu}(m) = m + 1$.

Thus we may and will assume without loss of generality that $(P_{\theta} \mid \theta \in \Theta)$ satisfies the extra conditions of the above Lemma. By well-known properties of Laplace transforms, k and ψ are analytic functions in a neighbourhood of zero. Hence, we may expand (3.19) into a power series in m for $m \in M_{\nu}$. It follows from (3.15) and (3.17) that k'(0) = 0. Moreover, since ν is not concentrated in one point, it follows from (3.18) that $k''(0) \neq 0$. Thus, $\psi(0) = 0$ and $\psi'(0) \neq 0$, which implies that s_n is a polynomial of degree exactly n. The **associated Sheffer polynomials** of a natural exponential family are the polynomials $(s_n)_{n \in \mathbb{N}}$ defined by

$$\varphi(m,x) = \sum_{n=0}^{\infty} s_n(x) m^n, \qquad (3.20)$$

where φ is defined by (3.19).

The following theorem relates the variance function of a natural exponential family to the delta operator of its associated Sheffer sequence.

Theorem 3.7.3 Let $\{P_{\theta} | \theta \in \Theta\}$ be a natural exponential family generated by a measure ν with associated Sheffer sequence $(s_n)_{n \in \mathbb{N}}$ (thus we assume without loss of generality that the extra conditions of Lemma 3.7.2 hold). Let Q = q(D)be the delta operator and A = f(D) be the invertible operator of $(s_n)_{n \in \mathbb{N}}$. Then $q'(D) = V_{\nu}(q(D))$ and f'(D) = q(D) f(D). Moreover, f is the Laplace transform of ν .

Proof: It follows from equations (3.19) and (3.20) and Corollary 2.4.9 that $q(D) = \psi^{-1}(D) = k'(D)$. Thus, by equation (3.18) and the definition of variance function, we arrive at $q'(D) = k''(D) = V_{\nu}(k'(D))$. For the second statement, note that by Corollary 2.4.9 we have $f(D) = e^{k(D)}$. Hence, $f'(D) = k'(D) e^{k(D)} = q(D) f(D)$. The last statement follows from $k(\theta) = \log L(\theta)$ and Equation (3.14).

We now are ready to prove the classification result mentioned in the introduction. The original proof is in [89]), another proof of this result can be found in [143, Theorem 4.1]. The merit of our proof is that it explains why the result is true.

Theorem 3.7.4 (Feinsilver) The variance function of a natural exponential family is quadratic if and only if the associated Sheffer polynomials are orthogonal.

Proof: Combine Theorems 3.7.3 and 3.4.1.

3.7.2 Natural exponential families and approximation theory

In this subsection we show that exponential families appear in disguise in approximation theory⁴. An important consequence of this is that the results of [156, 123, 124] are of importance for the statistics literature (in particular, it turns out that many of the results in [160] were predated by the abovementioned papers). We will also see how our approach differs from the approach in [123, 124, 156] (apart from different terminology).

We begin with recalling the basics of exponential-type approximation operators, following the exposition in [156] (see also [123, 124]). We slightly change the notation in order to be able to compare directly.

Let $W(\lambda, m, x)$ be the kernel of an exponential-type operator, i.e. $W(\lambda, m, x)$ is a generalized function⁵ such that

$$W(\lambda, m, x) \geq 0$$
 (3.21)

$$\int_{-\infty}^{\infty} W(\lambda, m, x) \, dx = 1 \tag{3.22}$$

$$\frac{\partial}{\partial m} W(\lambda, m, x) = \frac{\lambda}{p(m)} W(\lambda, m, x) (x - m), \qquad (3.23)$$

where p is analytic and positive on an interval on the real line. The corresponding positive approximation operator is defined by

$$(S_{\lambda} f)(t) = \int_{-\infty}^{\infty} W(\lambda, t, x) f(x) \, dx.$$
(3.24)

It is shown in [124, Corollary 3.2] that any solution of the partial differential equation (3.23) (together with the normalization condition (3.22)) is of the form

$$W(\lambda, m, x) = \exp \left(\lambda \int_{c}^{m} \frac{x - y}{p(y)} \, dy\right) C(\lambda, x).$$
(3.25)

The normalization condition (3.22) yields that exp $\left(\lambda \int_{c}^{g(m)} y/p(y) dy\right)$ is the Laplace transform of $C(\lambda, x)$, where $g(m) = \int_{c}^{m} 1/p(y) dy$. In other words, for fixed λ , the $W(\lambda, m, x)$ form a natural exponential family generated by $d\nu(x) = C(\lambda, x)$ (cf. formulas (3.15) and (3.19)) such that $\psi(m) = \int_{c}^{m} \lambda/p(y) dy$.

⁴On April 29, 1992, Gérard Letac delivered a beautiful lecture on natural exponential families at a one-day conference in Leuven, Belgium. This subsection arose out of remarks made on that occasion by Mourad Ismail.

⁵In fact, one would like to say that $W(\lambda, m, x)$ is the density function of a random variable. However, since we don't want to exclude random variables with discrete parts, we have to resort to generalized functions.

Conversely, let $\{P_{\theta} | \theta \in \Theta\}$ be the natural exponential family generated by a measure ν . Consider the reparametrization (3.19). Define the functions $W(\lambda, m, x)$ by

$$W(\lambda, m, x) := e^{x\psi(\lambda m) - k(\psi(\lambda m))}.$$
(3.26)

Since $\psi = k^{-1}$, it follows that

$$\psi'(m) = 1/(k''(\psi(m)))$$

and

$$(k(\psi(m))' = m/(k''(\psi(m))).$$

Hence, the functions $W(\lambda, m, x)$ defined by (3.26) satisfy

$$rac{\partial}{\partial m} W(\lambda,m,x) = rac{\lambda}{k''(\psi(m))} (x-m).$$

Note that $k''(\psi)$ is the variance function of $\{P_{\theta} | \theta \in \Theta\}$ by (3.19) (i.e. it is the p appearing in (3.23)).

We have thus obtained a complete correspondence between kernels of exponential-type approximation operators and natural exponential families. Hence, we have shown that the classification problems for exponential-type approximation operators and natural exponential families are equivalent. However, we now want to point some differences between the approach in [123, 124, 156] and our approach. Cast in our terminology, Ismail and May expand the moment generating function of ν and invert Laplace transforms, while we expand the densities with respect to ν in (3.15) and solve differential equations. As a consequence, the polynomial sequences that correspond to approximation operators as in [123] differ from our polynomial sequences. Thus although the classifications yield the same probability distributions, they yield different associated polynomial sequences.

3.7.3 Quadratic variance functions

In this section we use the results of the previous sections to present the classification for natural exponential families with quadratic variance function in full detail.

Theorem 3.7.3 tells us that we must solve the differential equations (3.9) and (3.10) in order to obtain all exponential families with quadratic variance function. Note that since Q = q(D) is a delta operator, we must have q(0) = 0 and $q'(0) \neq 0$. Since A' = AQ and A is invertible, it follows that $\log(A^{-1}A') = Q$. Hence,

$$a(D) = \exp\left(\int q(D) \, dD\right). \tag{3.27}$$

Note that the integration constant must be equal to zero, since a(t) is the Laplace transform of a mean zero distribution.

Under these conditions, we find the following natural exponential families:

Normal distribution

If the variance function is constant, then $Q' = \alpha$ with $\alpha > 0$. Hence, $Q = \alpha D$. Now (3.27) yields that $A = e^{\alpha D^2/2}$. Thus, the corresponding natural exponential family is generated by a normal distribution with mean zero and variance α for $\alpha > 0$. The associated Sheffer polynomials are the Hermite polynomials of variance α ([202, p. 87 ff.]).

Poisson distribution

If the variance function is a polynomial of degree one, then $Q' = \alpha + \beta Q$. Thus, $Q = \alpha \left(\frac{e^{\beta D} - 1}{\beta}\right)$ and $A = \exp\left(\frac{\alpha}{\beta}\left(\frac{1}{\beta}e^{\beta D} - D\right)\right)$. Thus, the corresponding family is the Poisson family. The associated Sheffer polynomials are the Poisson-Charlier polynomials (see Section 2.5 or cite[p. 119 ff.]Rom9).

Gamma distribution

If the variance function is a polynomial of degree two with two identical roots, then $Q' = \alpha (Q-\beta)^2$. Hence, $Q = \beta \frac{D}{D+1/(\alpha\beta)}$ and $A = e^{\beta D} (1+\alpha\beta D)^{1/\alpha}$. Thus, the corresponding natural exponential family is the gamma distribution family. The associated Sheffer polynomials are the Laguerre polynomials of variance α ([202, p. 108 ff.]).

Binomial distribution

If the variance function has two different positive roots, then the corresponding natural exponential family is the binomial distribution family. The associated Sheffer polynomials are the Krawtchouk polynomials ([202, p. 125-126]).

Negative binomial distribution

If the variance function has two different negative roots, then the corresponding natural exponential family is the negative binomial distribution family. The associated Sheffer polynomials are a subclass of the Meixner polynomials of the first kind ([202, p. 125-126]).

Hyperbolic distribution

If the variance function has two complex conjugate roots, then the associated Sheffer polynomials are a subclass of the Meixner polynomials of the second kind ([202, p. 126]). The corresponding natural exponential family is generated by the hyperbolic distribution (see [144]).

Conclusion

A few final remarks on generalizations are in order. The approach of this section is not restricted to natural exponential families with quadratic variance function. For example, it could be used to obtain a classification of natural exponential families with cubic variance function as in [144] (in [124] no attempt is being made to obtain a complete classification). Letac and Mora state that it seems hard to obtain classifications of natural exponential families with higher order polynomial variance functions. In light of our approach, this is probably related to the fact that the differential equations Q' = V(Q) are hard (resp. impossible) to solve explicitly when V is a polynomial of degree more than three (resp. four).

A more interesting direction is to generalize our approach to natural exponential families generated by multivariate distributions ([59, 129, 143, 14]).

Chapter 4

Banach algebras

Existence of logarithms of functions is needed in several parts of mathematics. E.g., in the theory of entire functions of a complex variable one needs that if f is a non-vanishing entire function, then there exists an entire function g such that $f(z) = e^{g(z)}$ for all $z \in \mathbb{C}$. In probability theory the following analogous result is essential for the theory of infinitely divisible probability measures: if f is a non-vanishing complex-valued continuous function on \mathbb{R} , then there exists a continuous function g such that $f(x) = e^{g(x)}$ for all $x \in \mathbb{R}$ (see e.g. [62, Chapter 7]).

These results are well-known, but their proofs use ad-hoc methods. The following theorem is not well-known and no elementary proof is known: if $f(z) = \sum_{n=0}^{\infty} a_n z^n$ is such that $\sum_{n=0}^{\infty} |a_n| < \infty$ and $f(z) \neq 0$ for $|z| \leq 1$, then there exists a function $g(z) = \sum_{n=0}^{\infty} b_n z^n$ such that $\sum_{n=0}^{\infty} |b_n| < \infty$ and $f(z) = e^{g(z)}$ for all $|z| \leq 1$. We will apply this theorem in Section 4.4, where convergence problems concerning polynomials of convolution type will be studied. The theorem is also useful in prediction theory, see [21, Theorem 4.1] or [237, Theorem 6].

In this chapter a unified approach is presented to these and related results. The approach, which seems to be new, uses only elementary Banach algebra techniques and is presented in Section 4.1. Sections 4.2 and 4.3 contain applications of the results of Section 4.1. E.g., the three results mentioned above are Theorems 4.2.11, 4.2.7 and 4.2.2. We apply the results of Sections 4.2 and 4.3 to obtain new analytical results on polynomials of convolution type in Section 4.4 and central limit theorems in Chapter 5. The reader should consult [111] or the survey [112] for other applications of Banach algebra theory to polynomials of convolution type. Finally, in Section 4.5 a two-sided analogue of functions of convolution type is introduced and studied.

This chapter is an extended version of [73].

Contents of Chapter 4

4.1 General Banach algebra techniques.

- 4.2 Algebras with contractible maximal ideal space.
- 4.3 Algebras on the unit circle.
- 4.4 Applications to polynomials of convolution type.
- 4.5 Applications to two-sided sequences of functions of convolution type.

4.1 General Banach algebra techniques

The purpose of this section is to set up the Banach algebra machinery for the approach mentioned in the introduction of this chapter. At the end of this section, a new discussion of the Arens-Royden theorem can be found.

We begin with a review of the basics of Banach algebra theory. Chapter 18 of [213] is recommended as a quick introduction to Banach algebra theory.

A Banach algebra \mathcal{B} is a complex Banach space that also possesses a multiplication (it is important that \mathcal{B} is a vector space over \mathbb{C} , see [212, Remarks 10.4]). This multiplication must obey the distributive and associative law and must satisfy the inequality $||xy|| \leq ||x|| ||y||$ for all $x, y \in \mathcal{B}$. A Banach algebra \mathcal{B} such that xy = yx for all $x, y \in \mathcal{B}$ is said to be commutative. An important example of a commutative Banach algebra is the Banach algebra $\mathcal{C}(\mathcal{K})$ of continuous complex-valued functions on a compact Hausdorff space \mathcal{K} ; addition and multiplication are defined pointwise. The norm of $\mathcal{C}(\mathcal{K})$ is the supremum norm.

An element $u \in \mathcal{B}$ is called **unit element** or **identity** of \mathcal{B} if xu = ux = x for all $x \in \mathcal{B}$ and ||u|| = 1 (the last requirement can be weakened, see e.g. [212, Theorem 10.2]). It follows from elementary algebra that at most one unit element exists. If a Banach algebra has no unit element, then a unit element can be adjoined (see [142] for a detailed account of the relations between a Banach algebra without unit element and the Banach algebra obtained by adjoining a unit element).

Let \mathcal{B} be a Banach algebra with unit element u. An element $x \in \mathcal{B}$ is **invertible** if there exists a $y \in \mathcal{B}$ such that xy = yx = u. The set of all invertible elements of \mathcal{B} will be denoted by inv \mathcal{B} . We equip inv \mathcal{B} with the norm topology inherited from \mathcal{B} . Equipped with this topology inv \mathcal{B} is a topological group.

Continuous linear functionals are important in the theory of Banach spaces . Their role is taken over by complex homomorphisms in the Banach algebra case. A **complex homomorphism** of a Banach algebra is a not identically zero continuous linear multiplicative map from the Banach algebra into \mathbb{C} . The set of complex homomorphisms of a Banach algebra \mathcal{B} is called the **maximal**

ideal space of \mathcal{B} and will be denoted by \mathcal{M} . The name maximal ideal space is explained by the fact that there is a one-to-one correspondence between maximal ideals of \mathcal{B} and null spaces of complex homomorphisms of \mathcal{B} (see [212, Theorem 11.15]). If \mathcal{B} is a commutative Banach algebra with unit element, then $\mathcal{M} \neq \emptyset$ (see [142, Theorem 3.3.2]). We will consider elements of the maximal ideal space as complex homomorphisms. We equip the maximal ideal space \mathcal{M} with the Gelfand topology, i.e. the topology of pointwise convergence. This makes \mathcal{M} into a compact Hausdorff space ([212, Theorem 11.9a]). Complex homomorphisms are useful in deciding the invertibility of an element as can be seen from Theorem 4.1.2.

The following theorem gives an explicit example of a maximal ideal space.

Theorem 4.1.1 Let \mathcal{K} be a compact Hausdorff space. The complex homomorphisms of $\mathcal{C}(\mathcal{K})$ are the point evaluations on \mathcal{K} . Moreover, the maximal ideal space of $\mathcal{C}(\mathcal{K})$ with its Gelfand topology is homeomorphic to \mathcal{K} .

Proof: See [212, Example 11.13a] or [83, Proposition 2.3]. \Box

Theorem 4.1.2 Let \mathcal{B} be a Banach algebra with unit element. Then $x \in inv \mathcal{B}$ if and only if $\Lambda(x) \neq 0$ for all $\Lambda \in \mathcal{M}$.

Proof: See [212, Theorem 11.5c] or [213, Theorem 18.17c]. □

Let \mathcal{B} be a Banach algebra with unit element u. Define $\exp \mathcal{B}$ to be the subset of \mathcal{B} consisting of those $x \in \mathcal{B}$ such that $x = e^y$ for some $y \in \mathcal{B}$. Here e^y is defined by $e^y := \sum_{n=0}^{\infty} \frac{y^n}{n!}$ for all $y \in \mathcal{B}$, where $e^0 := u$. It is easy to see that this series converges in the norm topology for all $y \in \mathcal{B}$.

Remarks 4.1.3 Let \mathcal{B} be a Banach algebra with unit element u.

a) It follows from $e^y e^{-y} = u$ for all $y \in \mathcal{B}$ (see [35, Lemma 1.4.1]), that $\exp \mathcal{B} \subset \operatorname{inv} \mathcal{B}$.

b) If $x \in \mathcal{B}$ and ||x - u|| < 1, then $x \in \exp \mathcal{B}$ (see [35, Lemma 1.4.2]). In particular, $x \in \operatorname{inv} \mathcal{B}$.

c) It holds true that inv \mathcal{B} is open in \mathcal{B} (see [212, Theorem 10.12]).

d) The following inequality holds true for all $x \in \mathcal{B}$ and all $t \in \mathbb{C}$:

$$\|e^{tx}\| = \|\sum_{k=0}^{\infty} \frac{t^k x^k}{k!}\| \le \sum_{k=0}^{\infty} \frac{t^k}{k!} \|x\|^k = e^{|t|} \|x\|$$

Let T be a topological space. A subset U of T is **connected** if $U = O_1 \cup O_2$ where O_1 and O_2 are disjoint open subsets of U, then $O_1 = \emptyset$ or $O_2 = \emptyset$ (see e.g. [84, Chapter 5]. A **component** of U is a connected subset of U which is not contained in a larger connected subset of U. Note that components are relatively closed ([84, Chapter 5, Theorem 3.2]), that components of open sets in a locally connected space ([84, Chapter 5, Definition 4.1]) are open ([84, Chapter 5, Theorem 4.2]) and that continuous images of connected sets are connected ([84, Chapter 5, Theorem 1.4]). Note that a union of non-disjoint connected sets is again connected ([84, Chapter 5, Theorem 1.5]).

We saw above that if \mathcal{B} is a Banach algebra with unit element, then inv \mathcal{B} is a topological group with the relative norm topology. Let \mathcal{G}_1 be the component of inv \mathcal{B} that contains the unit element of \mathcal{B} .

Theorem 4.1.4 Let \mathcal{B} be a commutative Banach algebra with unit element u. Then $\exp \mathcal{B} = \mathcal{G}_1$. In particular, $\exp \mathcal{B}$ is closed in inv \mathcal{B} .

Proof: An elementary proof of the first statement can be found in [35, Theorem 1.4.3]. The second statement follows from the first statement, since components are relatively closed. $\hfill\square$

Theorem 4.1.4 is difficult to use, since there is no general way to calculate \mathcal{G}_1 . For the algebras that will be discussed in Section 4.2, this problem will be solved by using the following theorem.

Definition 4.1.5 Let T be a topological space and $a, b \in T$. A path from a to b in T is a continuous function $f : [0,1] \to T$ such that f(0) = a and f(1) = b.

Theorem 4.1.6 Let \mathcal{B} be a commutative Banach algebra with unit element u. Then $x \in exp \ \mathcal{B}$ if and only if $x \in inv \ \mathcal{B}$ and there is path f in $inv \ \mathcal{B}$ from αu to x for some $\alpha \in \mathbb{C} \setminus \{0\}$.

Proof: ' \Rightarrow ' By definition there exists $y \in \mathcal{B}$ such that $x = e^y$. Then F, defined by $F(t) := e^{ty}$, is a path in inv \mathcal{B} from u to $e^y = x$.

'⇐' Let g be an arbitrary path in $\mathbb{C} \setminus \{0\}$ from 1 to α. Then h, defined by h(t) := g(t)u, is a path in inv \mathcal{B} from u to αu . Therefore F, defined by F(t) := h(2t) for $0 \le t \le \frac{1}{2}$ and F(t) := f(2t-1) for $\frac{1}{2} \le t \le 1$, is a path in inv \mathcal{B} from u to x. It follows from the continuity of F that F([0,1]) is connected. Hence $F([0,1]) \cup \mathcal{G}_1$ is connected, since $u \in F([0,1]) \cap \mathcal{G}_1$. Because \mathcal{G}_1 is the largest connected subset of inv \mathcal{B} that contains u, we have $F([0,1]) \subset \mathcal{G}_1$. It follows in particular that $x = F(1) \in \mathcal{G}_1$, hence $x \in \exp \mathcal{B}$ by Theorem 4.1.4. \Box

Remarks 4.1.7 a) Theorem 4.1.6 implies that \mathcal{G}_1 is a path-component of inv \mathcal{B} (see [84, Section 5.5]). This also follows from Theorem 4.1.4, since inv \mathcal{B} is locally path-connected (because it is an open subset of a normed linear space) and in a locally path-connected space components and path-components coincide ([84, Chapter 5, Theorem 5.5]). Note that in general path-connectedness is stronger than connectedness ([84, Chapter 5, Theorem 5.3]).

b) Theorems 4.1.4 and 4.1.6 are false when \mathcal{B} is not commutative. E.g., let $\mathcal{B}(\mathcal{H})$ denote the Banach algebra of all bounded linear operators on a Hilbert space \mathcal{H} . Then inv $\mathcal{B}(\mathcal{H})$ is connected ([212, Theorem 12.37]), but need not be equal to exp $\mathcal{B}(\mathcal{H})$ ([212, Theorem 12.38]).

The following theorem describes the form of the components of inv \mathcal{B} . Recall that $x \exp \mathcal{B} := \{xy : y \in \exp \mathcal{B}\}.$

Theorem 4.1.8 Let \mathcal{B} be a commutative Banach algebra with unit element u. Then the components of inv \mathcal{B} are of the form $x \exp \mathcal{B}$ with $x \in inv \mathcal{B}$.

Proof: Let C be a component of inv \mathcal{B} . Take an arbitrary element $x \in C$. The continuous image $x^{-1}C$ of C is connected and contains u. Hence, $x^{-1}C \subset \exp \mathcal{B}$ which implies that $C \subset x \exp \mathcal{B}$. In order to prove the other inclusion, note that $x \exp \mathcal{B}$ is a connected subset of inv \mathcal{B} (since it is the continuous image of a connected set by Theorem 4.1.4) that contains x. Hence, $x \exp \mathcal{B} \subset C$. \Box

In order to obtain further results on exp \mathcal{B} and inv \mathcal{B} , we need to discuss the Gelfand transform. This is necessary in those cases where Theorem 4.1.6 does not help us. The idea behind the Gelfand transform is to transfer problems in a Banach algebra (e.g., the calculation of \mathcal{G}_1) to a canonically associated Banach algebra of the form $\mathcal{C}(\mathcal{K})$, i.e. a Banach algebra of continuous functions. The advantage of this procedure is that Banach algebras of continuous functions are simpler to work with.

If $x \in \mathcal{B}$, then we define a continuous function \hat{x} on \mathcal{M} (the maximal ideal space of \mathcal{B} by $\hat{x}(\Lambda) := \Lambda x$ for all $\Lambda \in \mathcal{M}$. The function \hat{x} is called the **Gelfand transform** of x. Note that the Gelfand topology is the weakest topology that makes all functions \hat{x} continuous (see [142, Corollary 3.3.1]). The Gelfand transform maps \mathcal{B} onto a subalgebra \mathcal{B} of $\mathcal{C}(\mathcal{M})$. The image of the algebra \mathcal{B} under the Gelfand transform, equipped with the supremum norm, need not be a closed subalgebra of $\mathcal{C}(\mathcal{M})$.

When $\mathcal{B} = L^1(\mathbb{R})$, then the Gelfand transform is nothing but the Fourier transform (see e.g. [213, Chapter 18]). However, this is not a typical example for the sequel since $L^1(\mathbb{R})$ is a Banach algebra without unit.

The following two theorems show that the Gelfand transform is useful for our purposes.

Theorem 4.1.9 Let \mathcal{B} be a commutative Banach algebra with unit element u and let x be an arbitrary element of \mathcal{B} . Then $x \in inv \mathcal{B}$ if and only if $\hat{x} \in inv \mathcal{C}(\mathcal{M})$.

Proof: Note that \mathcal{M} with its Gelfand topology is a compact Hausdorff space ([212, Theorem 11.9a]). It follows from Theorems 4.1.1 and 4.1.2 that $\hat{x} \in$ inv $\mathcal{C}(\mathcal{M})$ if and only if $\hat{x}(\Lambda) \neq 0$ for all $\Lambda \in \mathcal{M}$. By Theorem 4.1.2, $x \in$ inv \mathcal{B} if and only if $\Lambda(x) \neq 0$ for all $\Lambda \in \mathcal{M}$. The theorem now follows from the definition of \hat{x} .

The usual proof of Theorem 4.1.9 (see e.g. [83, Proposition 2.34]) uses the correspondence between maximal ideals and complex homomorphisms.

The following theorem is an analogue of Theorem 4.1.9 for $\exp \mathcal{B}$.

Theorem 4.1.10 Let \mathcal{B} be a commutative Banach algebra with unit element u and let x be an arbitrary element of \mathcal{B} . Then $x \in exp \ \mathcal{B}$ if and only if $\hat{x} \in exp \ \mathcal{C}(\mathcal{M})$.

Proof: ' \Rightarrow ' Let $y \in \mathcal{B}$ be such that $x = e^y$. Since the Gelfand transform is continuous, it follows that $\hat{x} = e^{\hat{y}}$. Hence, $\hat{x} \in \exp \mathcal{C}(\mathcal{M})$. ' \Leftarrow ' See [101, Chapter 3, Corollary 6.2].

The proof of Theorem 4.1.10 in [101] uses holomorphic functions. Since the statement of Theorem 4.1.10 is a topological statement (cf. Theorem 4.1.4), it seems appropriate to prove Theorem 4.1.10 in a purely topological way. Unfortunately, I have not been able to find a topological proof of Theorem 4.1.10.

We are now able to say something more about the Gelfand transform. The following corollary shows that the image of a Banach algebra under the Gelfand transform is a special kind of subalgebra of $\mathcal{C}(\mathcal{M})$ (cf. [212, Theorem 10.18]).

Corollary 4.1.11 Let \mathcal{B} be a commutative Banach algebra with unit element u. The Gelfand transform maps distinct components of inv \mathcal{B} into distinct components of inv $\mathcal{C}(\mathcal{M})$.

Proof: Let $y, z \in \text{inv } \mathcal{B}$ arbitrary. Suppose that \widehat{y} and \widehat{z} are in the same component of inv $\mathcal{C}(\mathcal{M})$. An application of Theorem 4.1.8 to the Banach algebra $\mathcal{C}(\mathcal{M})$ yields that $\widehat{y} \in \widehat{z} \exp \mathcal{C}(\mathcal{M})$, so $\widehat{(yz^{-1})} = \widehat{y} \ \widehat{z}^{-1} \in \exp \mathcal{C}(\mathcal{M})$. It follows from Theorem 4.1.10 that $yz^{-1} \in \exp \mathcal{B}$, so $y \in z \exp \mathcal{B}$. Hence, y and z are in the same component of inv \mathcal{B} by Theorem 4.1.8.

We conclude this section with a discussion of the Arens-Royden Theorem on the structure of inv $\mathcal{B}/\exp \mathcal{B}$. It follows directly from Theorem 4.1.8 that there exists a one-to-one correspondence between inv $\mathcal{B}/\exp \mathcal{B}$ and the components of inv \mathcal{B} . The Arens-Royden Theorem says that the algebraic quotient group inv $\mathcal{B}/\exp \mathcal{B}$ (with multiplication as binary operation) is isomorphic to $\mathcal{H}^1(\mathcal{M},\mathbb{Z})$, the first Čech cohomology group of \mathcal{M} . Thus the Arens-Royden Theorem expresses inv $\mathcal{B}/\exp \mathcal{B}$ in terms of the maximal ideal space \mathcal{M} of \mathcal{B} . The original proofs of Arens and Royden can be found in [11] and [211]; another (elegant) proof is in [101, Corollary 7.4, p. 91]. Following a suggestion of Douglas (see [83, Chapter 2]), we state the theorem in terms of $\pi^1(\mathcal{M})$, the first cohomotopy group of \mathcal{M} (definition below). It is proved in [120, Chapter 11, Theorem 7.1] that $\pi^1(\mathcal{M})$ and $\mathcal{H}^1(\mathcal{M},\mathbb{Z})$ are isomorphic.

Definition 4.1.12 Let \mathcal{K} be a topological space and let $f, g : \mathcal{K} \to V \subset \mathbb{C}$ be continuous functions. A homotopy in V of f with g is a continuous function $H : [0,1] \times \mathcal{K} \to V$ such that H(0,z) = f(z) and H(1,z) = g(z) for all $z \in \mathcal{K}$. If moreover \mathcal{K} is compact and Hausdorff, then the first cohomotopy group $\pi^1(\mathcal{K})$ is defined to be the group of homotopy equivalence classes of continuous maps from \mathcal{K} to $\{z : |z| = 1\}$. The group operation of $\pi^1(\mathcal{K})$ is pointwise multiplication.

We now prove the special case $\mathcal{B} = C(\mathcal{K})$ of the Arens-Royden Theorem. This special case is due to Bruschlinsky and Eilenberg (see [43] and [85]). Our proof is inspired by the proof of [83, Theorem 2.18].

Theorem 4.1.13 If \mathcal{K} is a compact Hausdorff space, then inv $\mathcal{C}(\mathcal{K})/\exp \mathcal{C}(\mathcal{K})$ and $\pi^1(\mathcal{K})$ are isomorphic groups.

Proof: We define a homomorphism H from inv $\mathcal{C}(\mathcal{K})/\exp \mathcal{C}(\mathcal{K})$ to $\pi^1(\mathcal{K})$ as follows. Let [f] be an element of inv $\mathcal{C}(\mathcal{K})/\exp \mathcal{C}(\mathcal{K})$. Then H([f]) is defined to be the homotopy equivalence class of f/|f|, where f is any representative of [f]. Note that H is a well-defined homomorphism, since it follows from Theorem 4.1.6 that the elements of inv $\mathcal{C}(\mathcal{K})/\exp \mathcal{C}(\mathcal{K})$ are homotopy equivalence classes of continuous maps from \mathcal{K} into $\mathbb{C} \setminus 0$. Since it is obvious that H is surjective, it only remains to prove that H is injective. Suppose that H([f]) = H([g]). Let f, g be representatives of [f], [g] respectively. Since f and f/|f| are homotopic, it follows that f and g are homotopic. Hence, [f] = [g].

We conclude this section with a few words on the general case of the Arens-Royden Theorem. In view of Theorem 4.1.13, it suffices to show that

inv $\mathcal{B}/\exp \mathcal{B} \cong$ inv $\mathcal{C}(\mathcal{M})/\exp \mathcal{C}(\mathcal{M})$,

where \mathcal{M} is the maximal ideal space of \mathcal{B} . It follows from Corollary 4.1.11 that the canonical map induced by the Gelfand transform is an injective homomorphism from inv $\mathcal{B}/\exp \mathcal{B}$ into inv $\mathcal{C}(\mathcal{M})/\exp \mathcal{C}(\mathcal{M})$. The difficult point is to prove surjectivity. For a proof of surjectivity using holomorphic calculus, see [101, Chapter 3, Theorem 7.2].

For more information on cohomology and Banach algebras, we refer to the survey articles by Johnson and Taylor in [251].

4.2 Algebras with contractible maximal ideal space

In this section Theorem 4.1.6 will be applied to some explicit Banach algebras. We thus obtain among other things a simple proof of a theorem due to Borsuk (see Theorem 4.2.8).

Notation $\mathcal{D} := \{z \in \mathbb{C} : |z| < 1\}; \overline{\mathcal{D}} := \{z \in \mathbb{C} : |z| \le 1\}.$

For later use we prove the following uniqueness lemma.

Lemma 4.2.1 a) Let \mathcal{K} be a connected topological space. Suppose that g and h are complex-valued continuous functions on \mathcal{K} such that g(a) = h(a) for some $a \in \mathcal{K}$ and that $e^{g(z)} = e^{h(z)}$ for all $z \in \mathcal{K}$. Then g(z) = h(z) for all $z \in \mathcal{K}$.

b) Let \mathcal{B} be a Banach algebra of absolutely summable sequences with component-wise addition and convolution as multiplication. If $x, y \in \mathcal{B}$ and $x = e^y$, then y_n is uniquely determined for $n \ge 1$ and y_0 is determined by $x_0 = e^{y_0}$. In particular, if $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$ and there exists $h \in \mathcal{B}$ with $h_0 = 0$ such that $(q_n(t_0))_{n\in\mathbb{N}} = e^{t_0h}$, then $h_n = t_0g_n$ for all $n \in \mathbb{N}$.

Proof: a) It follows from $e^{g(z)} = e^{h(z)}$ that $g(z) = h(z) + 2\pi i k(z)$ with $k(z) \in \mathbb{Z}$. Hence, k is a continuous integer-valued function on \mathcal{K} with k(a) = 0. Since the continuous image of \mathcal{K} is connected, we must have k = 0.

b) Define $\gamma_0 := 0$ and $\gamma_n := y_n$ for $n \ge 1$. Then $x = e^y$ is equivalent to $x_0 = e^{y_0}$ and $x_n = e^{y_0} \sum_{k=0}^n \frac{\gamma_n^{k*}}{k!}$ for $n \ge 1$. We will now show by induction on n that y_n is uniquely determined by x_0, \ldots, x_n . The case n = 1 is clear, since $x_1 = e^{y_0} \gamma_1 = x_0 y_1$ (note that $x_0 \ne 0$). Suppose by induction that the statement is true at n. It follows from Lemma 2.1.5b and Lemma 2.1.5c that γ_{n+1}^{k*} is a polynomial in $\gamma_1, \ldots, \gamma_n$ with coefficients not depending on $(\gamma_n)_{n\in\mathbb{N}}$ for $2 \le k \le n+1$. Since $\gamma_n = y_n$ for $n \ge 1$, the induction hypothesis implies that y_{n+1} is uniquely determined by x_0, \ldots, x_{n+1} . Because $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type, we have $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ for all $n \in \mathbb{N}$ and $g_0 = 0$ by Theorem 2.1.8. Since $h_0 = 0$ and $(2_n)_{n\in\mathbb{N}} qt_0 = e^{t_0 h}$, the argument used above yields $h_n = t_0 g_n$ for all $n \in \mathbb{N}$.

4.2.1 Algebras of summable sequences

Let $(\alpha_n)_{n\in\mathbb{N}}$ be a sequence of positive numbers satisfying $\alpha_0 = 1$ and $\alpha_{n+m} \leq \alpha_n \alpha_m$ for all $n, m \in \mathbb{N}$. Let $\ell_1(\alpha)$ be the Banach algebra of all complex sequences $x = (x_n)_{n\in\mathbb{N}}$ such that $||x||_{1,\alpha} := \sum_{n=0}^{\infty} \alpha_n |x_n| < \infty$. Addition in $\ell_1(\alpha)$ is defined component-wise, multiplication is defined to be convolution. The complex homomorphisms of $\ell_1(\alpha)$ are of the form $\Lambda_z(x) = \sum_{n=0}^{\infty} x_n z^n$ with $|z| \leq e^{\rho}$, where $\rho := \lim_{n \to \infty} n^{-1} \log \alpha_n$ (see [105, Section 19, pp. 116-120]). If $\rho = -\infty$, then the only complex homomorphism of $\ell_1(\alpha)$ is $\Lambda_0(x) = x_0$. The unit element of $\ell_{1,\alpha}$ is the sequence $(1, 0, 0, \ldots)$.

If $\alpha_n = 1$ for all $n \in \mathbb{N}$, then $\ell_1(\alpha)$ is the usual Banach algebra ℓ_1 of absolutely summable sequences.

Theorem 4.2.2 $\{x \in \ell_1(\alpha) : \sum_{n=0}^{\infty} x_n z^n \neq 0 \text{ for all } |z| \leq e^{\rho} \} = inv \ \ell_1(\alpha) = exp \ \ell_1(\alpha).$

Proof: The first equality follows from Theorem 4.1.2.

For the second equality we only need to prove inv $\ell_1(\alpha) \subset \exp \ell_1(\alpha)$ by Remark 4.1.3a. Let $x \in \text{inv } \ell_1(\alpha)$ be arbitrary. Then $\sum_{n=0}^{\infty} x_n z^n \neq 0$ for all $|z| \leq e^{\rho}$. Hence, in particular $x_0 \neq 0$. Define $f : [0,1] \longrightarrow \ell_1(\alpha)$ by $f(t) := (t^n x_n)_{n \in \mathbb{N}}$. It follows from dominated convergence that $\lim_{t \to s} ||f(t) - f(s)||_{1,\alpha} = 0$. Hence, f is a path in inv $\ell_1(\alpha)$ from $x_0 u$ to x. Theorem 4.1.6 now yields $x \in \exp \ell_1(\alpha)$.

For an extension of Theorem 4.2.2, see Theorem 4.3.9.

4.2.2 Algebras of continuous functions

Let \mathcal{K} be a compact Hausdorff space. Denote by $\mathcal{C}(\mathcal{K})$ the Banach algebra of continuous functions on \mathcal{K} with pointwise addition and multiplication (see [212, Example 11.13a]). The norm on $\mathcal{C}(\mathcal{K})$ is the supremum norm, denoted by $\|f\|_{\infty}$. The unit element of $\mathcal{C}(\mathcal{K})$ is the function that is identically one.

We will prove that if \mathcal{K} is contractible to a point (see Definition 4.2.3 below), then an analogue of Theorem 4.2.2 holds for $\mathcal{C}(\mathcal{K})$.

Definition 4.2.3 Let \mathcal{K} be a topological space. A contraction of \mathcal{K} to $z_0 \in \mathcal{K}$ is a continuous mapping $H : [0,1] \times \mathcal{K} \to \mathcal{K}$ such that $H(0,z) = z_0$ and H(1,z) = z for all $z \in \mathcal{K}$. If there exists a contraction of \mathcal{K} to some point of \mathcal{K} , then \mathcal{K} is said to be contractible.

Examples 4.2.4 a) Any disc $z \in \mathbb{C}$: $|z| \leq r$ is contractible: take H(t, z) = tz. b) If $a, b \in \mathbb{R}$, then [a, b] is contractible: take H(t, x) = a + t(x - a). c) The set $\mathbb{C} \setminus (-\infty, 0]$ is contractible: take $H(t, re^{i\varphi}) := (tr + 1 - t)e^{it\varphi}$.

Theorem 4.2.5 If \mathcal{K} is a contractible compact Hausdorff space, then we have inv $\mathcal{C}(\mathcal{K}) = \exp \mathcal{C}(\mathcal{K})$. In particular, if \mathcal{K} is a contractible compact subset of \mathbb{C} and f is a non-vanishing continuous function on \mathcal{K} , then there exists a continuous function g on \mathcal{K} such that $f(z) = e^{g(z)}$ for all $z \in \mathcal{K}$. Moreover, gis analytic in those points in which f is analytic. If f(a) = 1 for some point $a \in \mathcal{K}$, then there is unique continuous function g on \mathcal{K} such that $f(z) = e^{g(z)}$ for all $z \in \mathcal{K}$ and g(a) = 0.

Proof: It follows from Remark 4.1.3a that $\exp C(\mathcal{K}) \subset \operatorname{inv} C(\mathcal{K})$. Let $f \in \operatorname{inv} C(\mathcal{K})$ be arbitrary and let H be an arbitrary contraction of \mathcal{K} to $a \in \mathcal{K}$, say. By the uniform continuity of f and H, $\lim_{t\to s} \|f(H(t,.)) - f(H(s,.))\|_{\infty} = 0$. Hence F, defined by F(t) := f(H(t,.)), is a path in inv $C(\mathcal{K})$ from f(a)u to f. Now Theorem 4.1.6 yields $f \in \exp C(\mathcal{K})$.

If f is analytic at z_0 , then for all z sufficiently close to z_0 , we have $g(z) = \zeta + \log \left\{ 1 + \frac{f(z) - f(z_0)}{f(z_0)} \right\}$, where log denotes the principal branch of the logarithm and ζ denotes some number such that $e^{\zeta} = f(z_0)$. Thus g is analytic in z_0 . The last statement follows directly from Lemma 4.2.1a.

Remarks 4.2.6 a) Theorem 4.2.5 also holds if each component of \mathcal{K} is compact and contractible or if \mathcal{K} is the union of an increasing sequence of compact contractible Hausdorff spaces.

b) Theorem 4.2.5 also holds for compact subsets of \mathbb{C} if contractibility of \mathcal{K} is weakened to connectedness of $\mathbb{C} \setminus \mathcal{K}$ ([45, Corollary 4.33]). I have not been able to find a simple proof of this result with the methods of this chapter (cf. Remark4.2.9).

The so-called topologist's sine-curve (see e.g. [245, pp. 44-45]) is an example of a compact connected subset of \mathbb{C} with connected complement which is not contractible (this example was shown to me by Jan van Mill). It follows from

the Alexander Duality Theorem ([245, Chapter 11]) that if \mathcal{K} is a compact contractible subset of \mathbb{C} , then both \mathcal{K} and $\mathbb{C} \setminus \mathcal{K}$ are connected.

c) Jan van Mill also pointed out to me that if \mathcal{K} is a compact connected subset of \mathbb{C} with connected complement, then it follows from [31, Theorem 7.6, p. 322], that there exists a decreasing sequence $(\mathcal{K}_n)_{n\in\mathbb{N}}$ of compact contractible subsets of \mathbb{C} such that $\mathcal{K} = \bigcap_{n\in\mathbb{N}}\mathcal{K}_n$. Using this result, we can easily prove the extension of Theorem 4.2.5 mentioned in b) as follows: let f be an arbitrary non-vanishing continuous function on \mathcal{K} . By the Tietze Extension Theorem ([213, Theorem 20.4]), f has a continuous extension F on \mathbb{C} . Suppose there is a sequence $(z_n)_{n\in\mathbb{N}}$ such that $z_n \in \mathcal{K}_n$ and $F(z_n) = 0$ for all $n \in \mathbb{N}$. Since each \mathcal{K}_n is compact, there exists a convergent subsequence $(z_{n_k})_{k\in\mathbb{N}}$ whose limit z_0 belongs to $\bigcap_{n\in\mathbb{N}}\mathcal{K}_n = \mathcal{K}$. Hence, $f(z_0) = F(z_0) = \lim_{n\to\infty} F(z_n) = 0$, which contradicts that f is non-vanishing. Thus we have shown that there exists an $N \in \mathbb{N}$ such that $F(z) \neq 0$ for all $z \in \mathcal{K}_N$. Now the result follows by applying Theorem 4.2.5 to \mathcal{K}_N and F.

d) The following example shows that contractibility of \mathcal{K} is not a necessary condition in Theorem 4.2.5. Let \mathcal{K} be a finite set with at least two elements and equip \mathcal{K} with the discrete topology. Then \mathcal{K} is a compact topological space, which is not contractible. It is easy to see that inv $\mathcal{C}(\mathcal{K}) = \exp \mathcal{C}(\mathcal{K})$.

The Banach algebra ℓ_{∞} of bounded complex sequences is a more sophisticated example. The norm on ℓ_{∞} is the supremum norm. Addition and multiplication are defined pointwise. It follows from general properties of the Čech-Stone compactification (see e.g. [101, Theorem 8.3, p. 17] or [142, p. 90]) that the Banach algebras ℓ_{∞} and $\mathcal{C}(\beta\mathbb{N})$, where $\beta\mathbb{N}$ denotes the Čech-Stone compactification of \mathbb{N} , are isomorphic. Note that $\beta\mathbb{N}$ is not contractible, since each $n \in \mathbb{N}$ is an isolated point of $\beta\mathbb{N}$. We will now show that inv $\mathcal{C}(\beta\mathbb{N}) = \exp \mathcal{C}(\beta\mathbb{N})$ by showing that inv $\ell_{\infty} = \exp \ell_{\infty}$. It is clear that $(x_n)_{n\in\mathbb{N}} \in \operatorname{inv} \ell_{\infty}$ if and only if $\inf_{n\in\mathbb{N}} |x_n| > 0$. Let $(x_n)_{n\in\mathbb{N}} \in \operatorname{inv} \ell_{\infty}$ be arbitrary. Choose $y_n \in \mathbb{C}$ such that $e^{y_n} = x_n$ and $Imy_n \in [0, 2pi]$ for all $n \in \mathbb{N}$. Then $(\operatorname{Re} y_n)_{n\in\mathbb{N}} \in \ell_{\infty}$, since $0 < \inf_{n\in\mathbb{N}} |x_n| \leq \sup_{n\in\mathbb{N}} |x_n| < \infty$. Thus $(y_n)_{n\in\mathbb{N}} \in \ell_{\infty}$ and $(x_n)_{n\in\mathbb{N}} \in \exp \ell_{\infty}$.

The special case $\mathcal{K} = [a, b]$ $(a, b \in \mathbb{R})$ of Theorem 4.2.5 and the following theorem are important in probability theory, see e.g. [62, Chapter 7].

Theorem 4.2.7 Let f be a non-vanishing continuous function on \mathbb{R} such that f(0) = 1. Then there exists a unique continuous function g on \mathbb{R} such that $f(x) = e^{g(x)}$ for all $x \in \mathbb{R}$ and g(0) = 0.

Proof: It follows from Theorem 4.2.5 and Example 4.2.4b that there exists for each $n \in \mathbb{N}$ a unique continuous function g_n such that $e^{g_n} = f$ on [-n, n] and $g_n(0) = 0$. By Lemma 4.2.1a, $g_n = g_m$ on [-n, n] if m > n. Hence, the function g, defined by $g(x) := g_n(x)$ if $|x| \leq n$, is well-defined, continuous, satisfies $f(x) = e^{g(x)}$ for all $x \in \mathbb{R}$, and g(0) = 0. Uniqueness follows from Lemma 4.2.1a.

If \mathcal{K} is an arbitrary compact subset of \mathbb{C} , then the following theorem due to Borsuk (see e.g. [45, Theorem 4.24]) states which continuous functions on \mathcal{K}

have continuous logarithms. Our proof, which is new, follows from the simple observation that if $f, g \in C(\mathcal{K})$ and there exists a homotopy in $\mathbb{C} \setminus 0$ of f with g, then there is a path in inv $C(\mathcal{K})$ (with respect to the norm topology) from f to g.

For the definition of homotopy appearing in the following theorem, see Definition 4.1.12.

Theorem 4.2.8 (Borsuk) Let \mathcal{K} be a compact subset of \mathbb{C} and let $f : \mathcal{K} \to \mathbb{C} \setminus 0$ be continuous. Then the following statements are equivalent:

- 1. there exists a homotopy in $\mathbb{C} \setminus 0$ of f with a constant function.
- 2. there exists a continuous function $g : \mathcal{K} \to \mathbb{C}$ such that $f(z) = e^{g(z)}$ for all $z \in \mathcal{K}$.
- 3. f has an extension to a continuous function $F : \mathbb{C} \to \mathbb{C} \setminus 0$.

Proof: We will prove $1 \Leftrightarrow 2$ and $2 \Leftrightarrow 3$.

 $(1 \Rightarrow 2)$ Let H be a homotopy of f in $\mathbb{C} \setminus 0$ with a constant α . By the uniform continuity of H on \mathcal{K} , we have $\lim_{t\to s} ||H(t,.) - H(s,.)||_{\infty} = 0$. Hence, H is a path in inv $\mathcal{C}(\mathcal{K})$ from f to αu . Now 2) follows from Theorem 4.1.6.

'2 \Rightarrow 1' If g is any function as in 2), then H, defined by $H(t,z) := e^{(1-t)g(z)}$, is a homotopy in $\mathbb{C} \setminus 0$ of f with the constant function 1.

 $^{\prime}2 \Rightarrow 3^{\prime}$ Let g be any function as in 2). By the Tietze Extension Theorem ([213, Theorem 20.4]), g has a continuous extension G on C. Obviously, the function e^{G} is a non-vanishing continuous extension of f to C.

'3 \Rightarrow 2' Since \mathcal{K} is compact, there exists r > 0 such that $\mathcal{K} \subset \{z \in \mathbb{C} : |z| \leq r\}$. By Example 4.2.4a and Theorem 4.2.5, there exists a continuous function g such that $F(z) = e^{g(z)}$ for all $|z| \leq r$.

Remark 4.2.9 If \mathcal{K} is a compact connected subset of \mathbb{C} with connected complement, then every non-vanishing continuous function on \mathcal{K} satisfies 1) of Theorem 4.2.8 (Robbert Fokkink pointed out to me that this is a special case of the Alexander Duality Theorem ([245, Chapter 11]); there seems to be no direct simple proof of this special case). Hence, every non-vanishing continuous function on \mathcal{K} has a continuous logarithm (cf. Remark 4.2.6b).

For topological proofs of the theorems on continuous functions in this section, see [45, Chapter IV] (uses homotopy) or [98, Chapter 1] (uses covering spaces).

4.2.3 Algebras of holomorphic functions

Let \mathcal{A}_r (r > 0) be the Banach algebra of all continuous functions on $\{z \in \mathbb{C} : |z| \leq r\}$ that are holomorphic on $z \in \mathbb{C} : |z| < r$. The norm is the supremum norm. If r = 1, then we write \mathcal{A} for \mathcal{A}_1 . The algebra \mathcal{A} is known as the disc algebra.

Theorem 4.2.10 If $f \in A_r$ does not vanish on $\{z \in \mathbb{C} : |z| \leq r\}$, then there is a $g \in A_r$ such that $f(z) = e^{g(z)}$ for all $|z| \leq r$.

Proof: We can use Theorem 4.2.5 or proceed as follows: the complex homomorphisms of \mathcal{A}_r are point evaluations on $\{z \in \mathbb{C} : |z| \leq r\}$ ([213, proof of Theorem 18.18]). Hence $f \in \text{inv } \mathcal{A}_r$ and F, defined by F(t)(z) := f(tz), is a path in inv \mathcal{A}_r from f(0)u to f. We conclude from Theorem 4.1.6 that $f \in \exp \mathcal{A}_r$.

Using the same trick as in the proof of Theorem 4.2.7, we now extend Theorem 4.2.10 to entire functions, i.e. functions holomorphic on \mathbb{C} .

Theorem 4.2.11 Let f be a non-vanishing entire function such that f(0) = 1. Then there exists a unique entire function g such that $f(z) = e^{g(z)}$ for all $z \in \mathbb{C}$ and g(0) = 0.

Proof: Applying Theorem 4.2.10 to the Banach algebras \mathcal{A}_n $(n \in \mathbb{N})$ and the restrictions f_n of f to $z \in \mathbb{C}$: $|z| \leq n$, we obtain holomorphic functions g_n on $z \in \mathbb{C}$: |z| < n such that $g_n(0) = 0$ and $e^{g_n(z)} = f_n(z)$ for all |z| < n. It follows from Lemma 4.2.1a that the function g, defined by $g(z) := g_n(z)$ for $|z| \leq n$, is well-defined. Clearly g is entire, g(0) = 0, and $f(z) = e^{g(z)}$ for all $z \in \mathbb{C}$. Uniqueness follows from Lemma 4.2.1a.

4.3 Algebras on the unit circle

In this section we will derive analogues of Theorem 4.2.5 for $\mathcal{C}(\mathbb{T})$, where $\mathbb{T} := \{z \in \mathbb{C} : |z| = 1\}$, and the Wiener algebra. These results will be used to prove Theorem 4.3.9, which is essential for Section 4.4.

Let us have a closer look at inv $\mathcal{C}(\mathbb{T})$ before stating and proving the correct analogue of Theorem 4.2.5. Note that \mathbb{T} is not contractible (see [42]) and that Theorem 4.2.5 is not true for $\mathcal{K} = \mathbb{T}$. E.g, $e^{i\theta} \in \text{inv } \mathcal{C}(\mathbb{T})$, but $e^{i\theta} \notin \exp \mathcal{C}(\mathbb{T})$ (see [42]).

Let $f \in \text{inv } \mathcal{C}(\mathbb{T})$ be arbitrary. Then f can be identified with a non-vanishing continuous function on $[-\pi,\pi]$. Hence, by Theorem 4.2.5 there exists a $\varphi \in \mathcal{C}([-\pi,\pi])$ such that

$$f(e^{i\theta}) = e^{\varphi(\theta)}$$
 for all $\theta \in [-\pi, \pi]$.

Moreover, if $\varphi_1, \varphi_2 \in \mathcal{C}([-\pi, \pi])$ both satisfy the above equation, then an application of Lemma 4.2.1a to $\mathcal{K} = [-\pi, \pi]$, $a = -\pi$, $g(\theta) := \varphi_1(\theta) - \varphi_1(-\pi) + \varphi_2(-\pi)$ and $h(\theta) := \varphi_2(\theta)$ yields $\varphi_1(\pi) - \varphi_1(-\pi) = \varphi_2(\pi) - \varphi_2(-\pi)$. Thus the following notion is well-defined:

Definition 4.3.1 Let f be a non-vanishing complex-valued continuous function on $\{z : |z| = R\}$. Then **ind** (f), the **index of** f, is defined to be $(2\pi i)^{-1} \varphi(\pi) - \varphi(-\pi)$ where φ is any continuous function on $[-\pi, \pi]$ satisfying $f(Re^{i\theta}) = e^{\varphi(\theta)}$ for all $\theta \in [-\pi, \pi]$.

Lemma 4.3.2 If f and g are non-vanishing complex-valued continuous functions on $\{z : |z| = R\}$, then ind $(f) \in \mathbb{Z}$ and ind (fg) = ind (f) + ind (g).

Proof: Let φ and γ be such that $f(Re^{i\theta}) = e^{\varphi(\theta)}$ and $g(Re^{i\theta}) = e^{\gamma(\theta)}$ for all $\theta \in [-\pi, \pi]$. Since $e^{\varphi(-\pi)} = e^{\varphi(\pi)}$, it follows that $\varphi(\pi) - \varphi(-\pi)$ is a multiple of $2\pi i$. Hence, ind $f \in \mathbb{Z}$.

The second statement follows from ind $(fg) = (\varphi + \gamma)(\pi) - (\varphi + \gamma)(-\pi) = \varphi(\pi) - \varphi(-\pi) + (\gamma(\pi) - \gamma(-\pi)) = \text{ind } (f) + \text{ind } (g).$

We can now state the analogues of Theorem 4.2.5 alluded to in the introduction of this section.

Theorem 4.3.3 Let $f \in C(\mathbb{T})$ be arbitrary. Then the following are equivalent:

- a) $f \in exp \ C(\mathbb{T})$.
- b) $f \in inv \mathcal{C}(\mathbb{T})$ and for all $n \in \mathbb{N}$ there exists $g \in \mathcal{C}(\mathbb{T})$ such that $g^n = f$.
- c) $f \in inv C(\mathbb{T})$ and ind (f) = 0.

Proof: 'a \Rightarrow b' It follows from Remark 4.1.3a that $f \in \text{inv } \mathcal{C}(\mathbb{T})$. If $f = e^h$, then $g := e^{h/n}$ satisfies $g^n = f$.

'b \Rightarrow c' Suppose ind $(f) \neq 0$. Take n > |ind(f)| and let g be such that $g^n = f$. Since nind $(g) = ind(g^n) = ind(f)$, it follows that ind $(g) \notin \mathbb{Z}$, which is absurd.

'c \Rightarrow a' Let g be a continuous function such that $f(e^{i\theta}) = e^{g(\theta)}$ for all $\theta \in [-\pi,\pi]$. It follows from ind (f) = 0 that $g(\pi) = g(-\pi)$. Hence G, defined by $G(e^{i\theta}) := g(\theta)$, belongs to $\mathcal{C}(\mathbb{T})$ and $f = e^{G}$.

It was mentioned in the introduction of this section that $e^{i\theta} \notin \exp \mathcal{C}(\mathbb{T})$. This follows directly from Theorem 4.3.3, since ind $(e^{i\theta}) = 1$.

The following theorem describes the components of inv $\mathcal{C}(\mathbb{T})$:

Theorem 4.3.4 Define $C_k := \{f \in inv \ C(\mathbb{T}) | ind (f) = k\}$ for all $k \in \mathbb{Z}$. The components of inv $C(\mathbb{T})$ are precisely the sets C_k .

Proof: It is clear that the sets C_k form a partition of inv $\mathcal{C}(\mathbb{T})$. Note that exp $\mathcal{C}(\mathbb{T})$ is a component by Theorem 4.1.4 and that $C_0 = \exp \mathcal{C}(\mathbb{T})$ by Theorem 4.3.3. Thus the theorem holds for k = 0. Define for each $k \in \mathbb{Z}$ the map F_k by $(F_k f)(e^{it}) := e^{itk}f(e^{it})$. It follows from ind $(e^{itk}) = k$ and ind $(fg) = \operatorname{ind} (f) + \operatorname{ind} (g)$ that F_k maps C_0 onto C_k . Moreover, it is easy to see that F_k is an isometry. Hence, the sets C_k are homeomorphic to C_0 , and we are done.

Theorem 4.3.4 enables us to prove that $\pi_1(\mathbb{T})$, the fundamental group of \mathbb{T} , is isomorphic to \mathbb{Z} (a theorem appearing in every introductory textbook on algebraic topology, see e.g. [155, Chapter 12, Section 5]). The fundamental group of \mathbb{T} consists of all homotopy equivalence classes of continuous maps f: $[0,1] \to \mathbb{T}$ such that f(0) = f(1) = 1. The group operation is defined as follows: if [f] and [g] are homotopy equivalence classes of continuous maps with the above mentioned properties, then $[f \circ g]$ is the map defined by $(f \circ g)(t) := f(2t)$ for $0 \le t \le \frac{1}{2}$ and $(f \circ g)(t) := g(2t-1)$ for $\frac{1}{2} \le t \le 1$. This group operation is well-defined, since it does not depend on the choice of the representatives fand g of [f], [g] respectively (see e.g. [155, Chapter 5]). Define $H: \pi_1(\mathbb{T}) \to \mathbb{Z}$ as follows. If $[f] \in \pi_1(\mathbb{T})$ has representative f, define H([f]) := ind (F), where F is defined by $F(e^{i\theta}) := f(\frac{1}{2} + \theta/(2\pi))$. This definition does not depend on the choice of the representative f: if f and g are homotopic, then the corresponding functions F and G are also homotopic, hence have the same index by Theorems 4.1.6 and 4.3.4. It follows from Lemma 4.3.2 that H is a homomorphism and it follows from Theorems 4.1.6 and 4.3.4 that H is injective. Hence, H is an isomorphism, since H is clearly surjective. We conclude that the fundamental group of \mathbb{T} is isomorphic to \mathbb{Z} .

The Wiener algebra \mathcal{W} consists of all continuous functions on $\mathbb{T} := \{z \in \mathbb{C} : |z| = 1\}$ that can be expanded as absolutely convergent Fourier series (see [212, Example 11.13b]).

Addition and multiplication are defined pointwise; the norm is defined by $||\sum_{n=-\infty}^{\infty} a_n e^{in\theta}|| := \sum_{n=-\infty}^{\infty} |a_n|$. Note that the algebra $\ell_1(\mathbb{Z})$ of absolutely summable two-sided sequences is isometric to \mathcal{W} . The elements of \mathcal{W} are precisely the Gelfand transforms of elements of $\ell_1(\mathbb{Z})$.

The complex homomorphisms of \mathcal{W} are of the form $\Lambda_z(a) = \sum_{n=-\infty}^{\infty} a_n z^n$ for some $z \in \mathbb{T}$ (see e.g. [83, Theorem 2.57] or [142, Section 4.6]). Thus Theorem 4.1.2 yields that the invertible elements of \mathcal{W} are precisely those elements of \mathcal{W} that do not vanish on \mathbb{T} (see e.g. [213, Lemma 11.6]). This is a famous theorem due to Wiener; the Banach algebra proof indicated above (due to Gelfand) was one the first successes of Banach algebra theory.

Since the canonical bijection $z \to \Lambda_z$ is a continuous map from the compact set \mathbb{T} onto the compact Hausdorff set $\mathcal{M}(\mathcal{W})$ (in its Gelfand topology), it follows that \mathbb{T} and $\mathcal{M}(\mathcal{W})$ are homeomorphic (cf. [212, Section 3.8]). The analogue of Theorem 4.2.5 for the Wiener algebra \mathcal{W} can be found in [46]. We state this result as Theorem 4.3.5 and remark that the proof in [46] uses a special case of the deep Wiener-Lévy Theorem ([212, Theorem 10.27]).

Theorem 4.3.5 Let $f \in W$ be arbitrary. Then the following are equivalent:

- a) $f \in exp \mathcal{W}$.
- b) $f \in inv \mathcal{W}$ and for all $n \in \mathbb{N}$ there exists $g \in C(\mathbb{T})$ such that $g^n = f$.

c) $f \in inv \mathcal{W}$ and ind (f) = 0.

Proof: 'a \Rightarrow b' It follows from Remark4.1.3a that $f \in \text{inv } \mathcal{W}$. If $f = e^h$, then $q := e^{h/n}$ satisfies $q^n = f$.

'b \Rightarrow c' Suppose ind $(f) \neq 0$. Take n > |ind (f)| and let g be such that $g^n = f$. Since n ind $(g) = \text{ind } (g^n) = \text{ind } (f)$ by Lemma 4.3.2, it follows that ind $(g) \notin \mathbb{Z}$, which is absurd.

'c⇒a' First note that $f \in \text{inv } C(\mathbb{T})$. By Theorem 4.3.3, there exists $h \in C(\mathbb{T})$ such that $f(z) = e^{h(z)}$ for all $z \in \mathbb{T}$. Define $H : \mathcal{M}(\mathcal{W}) \to \mathbb{C}$ by $H(\Lambda_z) = h(z)$. Then $H \in \mathcal{C}(\mathcal{M}(\mathcal{W}))$, since $\mathcal{M}(\mathcal{W})$ and \mathbb{T} are homeomorphic. Moreover, $\widehat{f}(\Lambda_z) = \Lambda_z(f) = f(z) = e^{h(z)} = e^{H(z)} = \left(e^H\right)(\Lambda_z)$ for all $z \in \mathbb{T}$. Thus $f \in \exp \mathcal{M}(\mathcal{W})$ and Theorem 4.1.10 yields that $f \in \exp \mathcal{W}$. □

Remark 4.3.6 The proof of $c \Rightarrow a$ of Theorem 4.3.5 implicitly contains the trivial result that if X and Y are homeomorphic compact Hausdorff spaces, then $\mathcal{C}(X)$ and $\mathcal{C}(Y)$ are isometric Banach spaces. The converse is also true and known as the Banach-Stone Theorem (see e.g. [22, Theorem 3, p. 130]).

Theorem 4.3.7 Define $\mathcal{W}_k := \{f \in inv \ \mathcal{W} | ind (f) = k\}$ for all $k \in \mathbb{Z}$. The components of inv \mathcal{W} are precisely the sets C_k .

Proof: The proof is identical to the proof of Theorem 4.3.3.

We conclude this section with a result (Theorem 4.3.9) which will be essential for one of the main theorems of this chapter (Theorem 4.4.1). Theorem 4.3.9 is an extension of Theorem 4.2.2.

Lemma 4.3.8 Let $f, g \in inv C(\mathbb{T})$ be such that |f(z) - g(z)| < |f(z)| for all $z \in \mathbb{T}$. Then ind (f) = ind(g).

Proof: The assumptions imply that |1 - g(z)/f(z)| < 1 for all $z \in \mathbb{T}$. Hence, ||1 - g(z)/f(z)|| < 1 since \mathbb{T} is compact. By Remark 4.1.3b, $g/f \in \exp \mathcal{C}(\mathbb{T})$ and Theorem 4.3.3 yields ind (g/f) = 0. It follows from Lemma 4.3.2 that ind $(g) = \operatorname{ind} (f) + \operatorname{ind} (g/f) = \operatorname{ind} (f)$.

For notation of the following theorem, see Subsection 4.2.1.

Theorem 4.3.9 Let $x \in \ell_1(\alpha)$ be arbitrary. Define $\xi : \mathbb{T} \to \mathbb{C}$ by $\xi(z) := \sum_{n=0}^{\infty} x_n (ze^{\rho})^n$ for all $z \in \mathbb{T}$. Then the following are equivalent:

- a) $\xi(z) \neq 0$ for all $z \in \mathbb{T}$ and ind $\xi = 0$.
- b) $x \in inv \ell_1(\alpha)$.
- c) $x \in exp \ \ell_1(\alpha)$.

Proof: 'a \Rightarrow b' Define ξ_r (0 < r < 1) on \overline{D} by $\xi_r(z) := \xi(rz)$. Since ξ is a non-vanishing continuous function on \mathbb{T} , there exists $\delta > 0$ such that $\delta < |\xi(z)|$ for all $z \in \mathbb{T}$. If r < 1 is close enough to 1, then $|\xi(z) - \xi_r(z)| < \delta < |\xi(z)|$ for all $z \in \mathbb{T}$. By Lemma 4.3.8, ind $(\xi_r) = \text{ind } (\xi) = 0$ (restrict ξ_r to \mathbb{T}). Now the Argument Principle ([45, Corollary 5.86, p. 179]) yields that $\sum_{n=0}^{\infty} x_n z^n \neq 0$ for $|z| \leq re^{\rho}$. Since r can be arbitrarily close to 1, $\sum_{n=0}^{\infty} x_n z^n \neq 0$ for $|z| < e^{\rho}$. Hence, $x \in \text{inv } \ell_1(\alpha)$ by Theorem 4.2.2.

'b \Rightarrow c' This follows from Theorem 4.2.2.

'c \Rightarrow a' Since $x \in \exp \ell_1(\alpha)$, we have $\sum_{n=0}^{\infty} x_n z^n \neq 0$ for all $|z| \leq e^{\rho}$. A similar use of the Argument Principle as above yields that ind $\xi = 0$.

4.4 Applications to polynomials of convolution type

In this section we will study the analytical behaviour of the following generating function for polynomials of convolution type (Theorem 2.1.12d):

$$\sum_{n=0}^{\infty} q_n(t) \, z^n = e^{t \, g(z)},\tag{4.1}$$

where g(z) denotes the formal power series $\sum_{k=0}^{\infty} g_k z^k$. In particular, we will study absolute convergence and radius of convergence of the left-hand side of (4.1).

Notation If $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type, we will write $\psi(t, z) := \sum_{n=0}^{\infty} q_n(t) z^n$ whenever this series converges absolutely. We write g for the coefficient sequence $(g_n)_{n \in \mathbb{N}}$ of $(q_n)_{n \in \mathbb{N}}$ and q(t) for $(q_n)_{n \in \mathbb{N}} t$.

For the notation in the following theorem we refer to Subsection 4.2.1.

Theorem 4.4.1 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $g = (g_n)_{n \in \mathbb{N}}$. Let $(\alpha_n)_{n \in \mathbb{N}}$ be a sequence of positive numbers satisfying $\alpha_0 = 1$ and $\alpha_{n+m} \leq \alpha_n \alpha_m$ for all $n, m \in \mathbb{N}$. Then the following are equivalent:

- a) $g \in \ell_{1,\alpha}$.
- b) There exists M > 0 such that $||q(t)||_{1,\alpha} \leq e^{|t|M}$ for all $t \in \mathbb{C}$.
- c) $\lim_{t\downarrow 0} ||q(t)||_{1,\alpha} = 1.$
- d) $\limsup_{t \mid 0} \|q(t)\|_{1,\alpha} < 2.$
- e) There are $\delta > 0$ and $t_0 \in (0, \delta)$ such that $q(t) \in \ell_{1,\alpha}$ for all $t \in (0, \delta)$ and $\psi(t_0, z) \neq 0$ if $|z| = e^{\rho}$.

- f) There is $t_0 \in \mathbb{C} \setminus \{0\}$ such that $q(t_0) \in \ell_{1,\alpha}$ and $\psi(t_0, z) \neq 0$ for all $|z| \leq e^{\rho}$.
- g) There is $t_0 \in \mathbb{C} \setminus \{0\}$ such that $q(t_0) \in \ell_{1,\alpha}$ and $q(-t_0) \in \ell_{1,\alpha}$.

Moreover, if one of these conditions holds, then (4.1) holds and both series in (4.1) converge absolutely for all $t \in \mathbb{C}$ and all $|z| \leq e^{\rho}$.

Proof: ' $a \Rightarrow b$ ' We first show that $\left(e^{t\,g}\right)_{n} = q_{n}(t)$ for all $n \in \mathbb{N}$. Since the coordinate functionals of $\ell_{1,\alpha}$ are continuous, we have $\left(e^{t\,g}\right)_{n} = \left(\sum_{k=0}^{\infty} \frac{t^{k}\,g^{k}}{k!}\right)_{n} = \sum_{k=0}^{\infty} \left(\frac{t^{k}\,g^{k}}{k!}\right)_{n} = \sum_{k=0}^{\infty} g_{n}^{k*} \frac{t^{k}}{k!} = \sum_{k=0}^{n} g_{n}^{k*} \frac{t^{k}}{k!} = q_{n}(t)$ for all $n \in \mathbb{N}$. Now b) follows from Remark 4.1.3d.

'b \Rightarrow c' This follows from $q_0 = 1$ and $\alpha_0 = 1$.

' $c \Rightarrow d$ ' This is trivial.

 $d \Rightarrow e'$ It follows that there exists t_0 such that $||q(t)-u||_{1,\alpha} < 1$ for all $t \in [0, t_0]$. In particular, $q(t_0) \in \exp \ell_{1,\alpha}$ by Remark 4.1.3b. The statement now follows from Theorem 4.2.2.

'e $\Rightarrow f$ ' Since $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type, we have $\psi(u + v, z) = \psi(u, z) \psi(v, z)$ if $u, v, u + v \in [0, \delta)$ and $|z| = e^{\rho}$. Hence, $\psi(t, z) \neq 0$ for all $t \in (0, \delta)$ and all $|z| = e^{\rho}$ by [118, Theorem 4.17.1, p. 144]. Recall from Lemma 4.3.2 that the index of a non-vanishing continuous function on \mathbb{T} (Definition 4.3.1) is always an integer. If ind $\psi(t, .) \neq 0$ for some $t \in (0, \delta)$, then ind $\psi(t/n, .) \notin \mathbb{Z}$ for n large enough by Lemma 4.3.2, which is impossible. Hence Theorem 4.2.2 and Theorem 4.3.9 imply that for all $t \in (0, \delta), \psi(t, z) \neq 0$ if $|z| \leq e^{\rho}$.

 ${}^{'}f \Rightarrow g'$ Since $q_0(t_0) \neq 0$, there exists a unique sequence $(a_k)_{k\in\mathbb{N}}$ such that $\sum_{k=0}^{n} q_k(t_0) a_{n-k} = \delta_{0n}$ for all $n \in \mathbb{N}$. By Theorem 4.2.2, $(a_k)_{k\in\mathbb{N}} \in \ell_{1,\alpha}$. Using the defining property of convolution type we see that $a_k = q_k(-t_0)$ for all $k \in \mathbb{N}$. Hence, $q(-t_0) \in \ell_{1,\alpha}$.

 $g \Rightarrow a$ It follows from the defining property of polynomials of convolution type that $q(-t_0) = q(t_0)^{-1}$. Hence, $q(t_0) \in inv(\ell_{1,\alpha})$ and by Theorem 4.2.2, there exists a sequence $b = (b_n)_{n \in \mathbb{N}} \in \ell_{1,\alpha}$ with $b_0 = 0$ such that $q(t_0) = e^b$. It follows from Lemma 4.2.1b that $b_n = t_0 g_n$ for all $n \in \mathbb{N}$, which implies $g \in \ell_{1,\alpha}$.

The last statement follows from Theorem 2.1.8, since b) allows us to interchange summations. $\hfill \Box$

Remarks 4.4.2 a) It follows from the proof of ' $a \Rightarrow b$ ' that $M = ||g||_{1,\alpha}$ suffices in b).

b) If f) or g) hold, then (4.1) implies that they hold for all $t \in \mathbb{C}$.

c) If $g_n = 0$ for n even and $g \notin \ell_{1,\alpha}$, then $q(t) \notin \ell_{1,\alpha}$ for any $t \neq 0$ since in this case $q_n(-t) = (-1)^n q_n(t)$.

d) For an alternative proof of ' $e \Rightarrow f$ ' see Remark 4.5.4a.

e) We may weaken condition d) of Theorem 4.4.1 to d': 'there is $t \in \mathbb{C} \setminus \{0\}$ such that $||q(t)||_{1,\alpha} < 2$ ', since obviously $c \Rightarrow d' \Rightarrow g$.

Corollary 4.4.3 Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$. If there exists $M, \delta > 0$ such that $|q_n(t)| \leq M$ for all $t \in (0, \delta)$, then $\sum_{n=0}^{\infty} g_n z^n$ converges absolutely for |z| < 1.

Proof: Fix an arbitrary r with 0 < r < 1. Define $\alpha_n := r^n$. Since $\lim_{t \downarrow 0} q_n(t) = 0$ for $n \ge 1$ by Theorem 2.1.12e, it follows from dominated convergence that $\lim_{t \downarrow 0} ||q(t)||_{1,\alpha} = 1$. Thus Theorem 4.4.1 $c \Rightarrow a$ implies that $\sum_{n=0}^{\infty} g_n z^n$ converges absolutely for $|z| \le r$. Since r was arbitrary, it follows that $\sum_{n=0}^{\infty} g_n z^n$ converges absolutely for |z| < 1.

The converse of Corollary 4.4.3 is not true. E.g., take $g_n = n$ for all $n \in \mathbb{N}$. Then $|q_n(t)| \geq g_1 |t| = n |t|$. It is an open problem to find necessary and sufficient conditions on $(g_n)_{n \in \mathbb{N}}$ that insure that $\sum_{n=0}^{\infty} |q_n(t)| < \infty$ for all $t \in \mathbb{C}$ or for all $t \in (0, \infty)$.

We now prove an analogue of Theorem 4.4.1 for strict sense Sheffer sequences (see Section 2.4). We write s(t) for $(s_n(t))_{n \in \mathbb{N}}$.

The following lemma is needed for the proof of Theorem 4.4.5.

Lemma 4.4.4 Let $x(t) = (x_n(t))_{n \in \mathbb{N}}$ (t > 0) and $x = (x_n)_{n \in \mathbb{N}}$ be sequences in $\ell_{1,\alpha}$. If $\lim_{t \downarrow 0} ||x(t)||_{1,\alpha} = ||x||_{1,\alpha}$ and $\lim_{t \downarrow 0} x_n(t) = x_n$ for all $n \in \mathbb{N}$, then $\lim_{t \downarrow 0} ||x(t) - x||_{1,\alpha} = 0$.

 $\begin{array}{l} \text{Proof: Let } \epsilon > 0 \text{ be arbitrary. Choose } k \in \mathbb{N} \text{ such that } \|x\|_{1,\alpha} - \|P_k x\|_{1,\alpha} < \epsilon/5, \\ \text{where } P_k x := (x_0, x_1, \dots, x_k, 0, 0, \dots). \\ \text{Choose } s > 0 \text{ such that } \|x(t)\|_{1,\alpha} < \|x\|_{1,\alpha} + \epsilon/5 \text{ and } \|P_k x(t)\|_{1,\alpha} > \|P_k x\|_{1,\alpha} - \epsilon/5 \text{ for } 0 < t < s. \\ \text{If } 0 < t < s, \\ \text{then } \|x - x(t)\|_{1,\alpha} = \|P_k x - x(t)\|_{1,\alpha} + \|(I - P_k) x - x(t)\|_{1,\alpha} \le \epsilon/5 + \|(I - P_k) x\|_{1,\alpha} + \|(I - P_k) x(t)\|_{1,\alpha} \le \epsilon/5 + \epsilon/5 + \|x(t)\|_{1,\alpha} - \|P_k x(t)\|_{1,\alpha} \le 2\epsilon/5 + \|x\|_{1,\alpha} + \epsilon/5 - \|P_k x\|_{1,\alpha} + \epsilon/5 \le \epsilon. \\ \end{array}$

Theorem 4.4.5 Let $(s_n)_{n \in \mathbb{N}}$ be a strict sense Sheffer set for a delta operator Qwith basic set $(q_n)_{n \in \mathbb{N}}$. Let $g = (g_n)_{n \in \mathbb{N}}$ be the coefficient sequence of $(q_n)_{n \in \mathbb{N}}$. Let $(\alpha_n)_{n \in \mathbb{N}}$ be a sequence of positive numbers satisfying $\alpha_0 = 1$ and $\alpha_{n+m} \leq \alpha_n \alpha_m$ for all $n, m \in \mathbb{N}$. Then the following are equivalent:

- a) $g \in \ell_{1,\alpha}$ and $s(0) \in inv \ell_{1,\alpha}$.
- b) $s(0) \in inv \ell_{1,\alpha}$ and there exists M > 0 such that $e^{-tM} ||s(0)||_{1,\alpha} \leq ||s(t)||_{1,\alpha} \leq e^{tM} ||s(0)||_{1,\alpha}$ for t > 0.
- c) $s(0) \in inv \, \ell_{1,\alpha}$ and $\lim_{t \downarrow 0} \|s(t)\|_{1,\alpha} = \|s(0)\|_{1,\alpha}$.
- d) $g \in \ell_{1,\alpha}$ and $s(t) \in inv \, \ell_{1,\alpha}$ for some $t \in \mathbb{C} \setminus \{0\}$.

Moreover, if one of these conditions holds, then

$$\sum_{n=0}^{\infty} s_n(t) \, z^n = e^{t \, g(z)} \, \sum_{n=0}^{\infty} s_n(0) \, z^n$$

for all $t \in \mathbb{C}$ and all $|z| \leq e^{\rho}$.

Proof: We will use that by Theorem 2.4.4, s(t) = s(0) * q(t), where * denotes convolution.

 $a \Rightarrow b$ It follows from Theorem 4.4.1 that $||q(t)||_{1,\alpha} \leq e^{tM}$ for all t > 0. Hence, $||s(t)||_{1,\alpha} \leq e^{tM} ||s(0)||_{1,\alpha}$. For the other inequality, note that s(0) = s(0) * q(t) * q(-t) = s(t) * q(-t). Hence, for t > 0 we have $||s(0)||_{1,\alpha} \leq ||s(t)||_{1,\alpha} ||q(-t)||_{1,\alpha} \leq ||s(t)||_{1,\alpha} e^{tM}$.

 $b \Rightarrow c$ This is trivial.

 $c \Rightarrow d$ By Remark 4.1.3b, inv $\ell_{1,\alpha}$ is open. Since $\lim_{t\downarrow 0} s_n(t) = s_n(0)$ for all $n \in \mathbb{N}$, Lemma 4.4.4 yields $\lim_{t\downarrow 0} ||s(t) - s(0)||_{1,\alpha} = 0$. In particular, $s(t_0) \in \text{inv } \ell_{1,\alpha}$ for some $t_0 > 0$. Moreover, $q(t_0) = s(t_0) * s(0)^{-1} \in \text{inv } \ell_{1,\alpha}$. Now Theorem 4.4.1 yields $g \in \ell_{1,\alpha}$.

 $d \Rightarrow a'$ It follows from Theorem 4.4.1 that $q(-t) \in \text{inv } \ell_{1,\alpha}$. Hence, $s(0) \in \text{inv } \ell_{1,\alpha}$, since s(0) = s(t) * q(-t).

The last statement follows from s(t) = s(0) * q(t) (Theorem 2.4.4) and Theorem 4.4.1).

We now return to Theorem 4.4.1. We will try to obtain convergence results on (4.1) with weaker conditions on $(g_n)_{n \in \mathbb{N}}$.

The Banach algebra \mathcal{TA} consists of all one-sided sequences $(a_n)_{n\in\mathbb{N}}$ of complex numbers such that $f(z) := \sum_{n=0}^{\infty} a_n z^n$ is analytic on \mathcal{D} and can be extended to a continuous function on $\overline{\mathcal{D}}$. Addition is defined componentwise, multiplication is defined to be convolution. The norm on \mathcal{TA} is defined by $\|(a_n)_{n\in\mathbb{N}}\|_{\mathcal{TA}} := \sup_{|z|<1} |\sum_{n=0}^{\infty} a_n z^n|$. Note that if $(a_n)_{n\in\mathbb{N}} \in \mathcal{TA}$ and $f(z) := \sum_{n=0}^{\infty} a_n z^n$, then the Maximum Modulus Theorem yields $\|(a_n)_{n\in\mathbb{N}}\|_{\mathcal{TA}} = \sup_{|z|<1} |f(z)| = \sup_{|z|\leq 1} |f(z)| = \sup_{|z|=1} |f(z)|$ (here we denoted the extension of f to $\overline{\mathcal{D}}$ also by f). The space \mathcal{TA} is isometric to the disc algebra \mathcal{A} studied in Subsection 4.2.3).

Theorem 4.4.6 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $g = (g_n)_{n \in \mathbb{N}}$. Then the following are equivalent:

- a) $g \in TA$.
- b) There are $t \in \mathbb{C} \setminus \{0\}$ and $\delta > 0$ such that $q(t) \in \mathcal{T}A$ and $|\psi(t, z)| > \delta$ for all $z \in \mathcal{D}$.
- c) There exists $t \in \mathbb{C} \setminus \{0\}$ such that $q(t) \in \mathcal{T}\mathcal{A}$ and $q(-t) \in \mathcal{T}\mathcal{A}$.

If a), b) or c) holds, then :

- d) $q(t) \in \mathcal{T}\mathcal{A}$ for all $t \in \mathbb{C}$ and $|\psi(t, z)| > \delta(t) > 0$ for all $z \in \mathcal{D}$.
- e) (4.1) holds for all $t \in \mathbb{C}$ and all $z \in \mathcal{D}$.
- f) There exists M > 0 such $\|q(t)\|_{\mathcal{TA}} \leq e^{|t|M}$ for all $t \in \mathbb{C}$.

g) $\lim_{t\downarrow 0} ||q(t)||_{\mathcal{TA}} = 1.$

Proof ' $a \Rightarrow b$ ' Fix an arbitrary r with 0 < r < 1. Define α_n $(n \in \mathbb{N})$ by $\alpha_n := r^n$. Then Theorem 4.4.1 $a \Rightarrow e$ together with the last statement of Theorem 4.4.1 imply that $\psi(t, z) = e^{t g(z)}$ for all $t \in \mathbb{C}$ and all z with |z| < r. Since r was arbitrary, it follows that $\psi(t, z) = e^{t g(z)}$ for all $t \in \mathbb{C}$ and all $z \in \mathbb{C}$ and all $z \in \mathbb{C}$ and all $z \in \mathcal{D}$. Hence, $q(t) \in \mathcal{TA}$ for all $t \in \mathbb{C}$ since $e^{tg} \in \mathcal{TA}$. The second statement of b) also follows from $\psi(t, z) = e^{tg(z)}$.

 ${}^{b} \Rightarrow c'$ Since $q_{0}(t) \neq 0$, there exists a unique sequence $(a_{k})_{k\in\mathbb{N}}$ such that $\sum_{k=0}^{n} q_{k}(t) a_{n-k} = \delta_{0n}$ for all $n \in \mathbb{N}$. It follows Theorem 4.2.10 that $q(t) \in \exp \mathcal{TA}$. Hence, in particular $q(t) \in \operatorname{inv} \mathcal{TA}$ and $(a_{n})_{n\in\mathbb{N}} \in \mathcal{TA}$. Using the defining property of convolution type, we see that $a_{k} = q_{k}(-t)$ for all $k \in \mathbb{N}$. Hence, $q(-t) \in \mathcal{TA}$.

 ${}^{c} c \Rightarrow a'$ It follows from Theorem 4.2.10 that $q(t) \in \exp \mathcal{TA}$. Thus there exists $(b_n)_{n \in \mathbb{N}} \in \mathcal{TA}$ with $b_0 = 0$ such that $\psi(t, z) = \exp \left(\sum_{n=0}^{\infty} b_n z^n\right)$ for all $z \in \mathcal{D}$. It follows from Lemma 4.2.1b that $b_n = t g_n$ for all $n \in \mathbb{N}$. Hence, $g \in \mathcal{TA}$.

Statements d) and e) follow from the proof of $a \Rightarrow b$. In order to prove f), note that e) implies $\|q(t)\|_{\mathcal{TA}} \leq e^{|t|} \|g\|_{\mathcal{TA}}$. Finally, $\liminf_{t\downarrow 0} \|q(t)\|_{\mathcal{TA}} \geq \liminf_{t\downarrow 0} |\psi(t,0)| = q_0(t) = 1$ and f) implies that $\limsup_{t\downarrow 0} \|q(t)\|_{\mathcal{TA}} \leq 1$. Hence, $\lim_{t\downarrow 0} \|q(t)\|_{\mathcal{TA}} = 1$.

Remark 4.4.7 It is an open problem whether property f) of Theorem 4.4.6 implies any of the properties a), b) or c).

The next theorem gives a sufficient condition for boundedness of the coefficient sequence $(g_n)_{n \in \mathbb{N}}$ in terms of $(q_n)_{n \in \mathbb{N}}$.

Theorem 4.4.8 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n \in \mathbb{N}}$. If $\lim_{t \downarrow 0} q_n(t) = 0$ uniformly in n = 1, 2, ...and $\lim_{t \downarrow 0} \frac{q_n(t)}{t} = g_n$ uniformly in n = 1, 2, ..., then $(g_n)_{n \in \mathbb{N}}$ is a bounded sequence.

Proof: Suppose $(g_n)_{n\in\mathbb{N}}$ is unbounded. Choose δ , $0 < \delta < 1$ such that $|q_n(t)| \leq 1$ and $\left|\frac{q_n(t)}{t} - g_n\right| \leq 1$ for $0 < t < \delta$. Choose N such that $|g_N| > 4/\delta$. Then $|q_N(\frac{1}{2}\delta)| = \left|\frac{q_N(\frac{1}{2}\delta)}{\frac{1}{2}\delta} \frac{1}{2}\delta\right| = \frac{1}{2}\delta \left|g_N + \left(\frac{q_N(\frac{1}{2}\delta)}{\frac{1}{2}\delta} - g_N\right)\right| \geq \frac{1}{2}\delta |g_N| - \frac{1}{2}\delta \left|\frac{q_N(\frac{1}{2}\delta)}{\frac{1}{2}\delta} - g_N\right| \geq 2 - \frac{1}{2}\delta > 1\frac{1}{2}$. This contradicts the choice of δ .

Note that the converse of Theorem 4.4.8 does not hold. E.g. take $g_n = 1$ for all $n \in \mathbb{N}$. Then $g_n^{2*} = n - 1$ and $q_n(t) \ge \frac{1}{2}(n-1)t^2$.

We now study the relation between the radii of convergence of $\sum_{n=0}^{\infty} q_n(t) z^n$ and $\sum_{k=0}^{\infty} g_k z^k$, where $(g_n)_{n \in \mathbb{N}}$ is the coefficient sequence of $(q_n)_{n \in \mathbb{N}}$.

$$\begin{aligned} \mathcal{R}_g &:= \text{ radius of convergence of } \sum_{k=0}^{\infty} g_k \, z^k. \\ \rho_t &:= \text{ radius of convergence of } \sum_{n=0}^{\infty} q_n(t) \, z^n. \\ \mathcal{N}_t &:= \{z \, : \, |z| < \rho_t \text{ and } \psi(t, z) = 0\}. \\ \nu_t &:= \text{ inf}\{|z| \, : \, z \in \mathcal{N}_t\} \text{ if } \mathcal{N}_t \neq \emptyset, \nu_t := \rho_t \text{ if } \mathcal{N}_t = \emptyset. \end{aligned}$$

We start our discussion with some examples.

Examples 4.4.9 a)
$$q_n(t) = \frac{x^n}{n!}$$
: $\rho_t = \mathcal{R}_g = \infty$ for all $t \in \mathbb{C}$.
b) $q_n(t) = {t \choose n}$: $\mathcal{R}_g = 1$, $\rho_t = \infty$ for $t \in \mathbb{N}$, $\rho_t = 1$ for $t \notin \mathbb{N}$.
c) $q_n(t) = t(t-an)^{n-1}/n!$: $\rho_t = \mathcal{R}_g = (|a|e)^{-1}$ (use Stirling's Formula)

Note that $\rho_0 = \infty$ because $q_n(0) = 0$ for $n \ge 1$.

It follows from Theorem 4.4.1 $a \Rightarrow b$ that $\mathcal{R}_g \leq \rho_t$ for all $t \in \mathbb{C}$. The examples suggest that $\mathcal{R}_g = \rho_t$ for all except countably many t. Theorem 4.4.1 $g \Rightarrow a$ shows that it is not possible that both $\rho_t > \mathcal{R}_g$ and $\rho_{-t} > \mathcal{R}_g$. The following theorem shows that the zeros of the functions $\psi(t, .)$ determine \mathcal{R}_g . Moreover, it enables us to prove the important property stated as Theorem 4.4.10e. This property will play an important role in the rest of this section and in Section 4.5.

Theorem 4.4.10 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type. Then:

- a) If $\mathcal{R}_g = 0$, then $\rho_t = 0$ for all $t \in \mathbb{C} \setminus \{0\}$.
- b) If $\mathcal{R}_g = \infty$, then $\rho_t = \infty$ for all $t \in \mathbb{C}$.
- c) If $0 < \mathcal{R}_g < \infty$, then $\mathcal{R}_g = \nu_t$ for all $t \in \mathbb{C} \setminus \{0\}$. In particular, $\nu_t = \nu_s$ for all $t, s \in \mathbb{C} \setminus \{0\}$ and $\psi(t, z) \neq 0$ for all $t \in \mathbb{C}$ and all $|z| < \mathcal{R}_g$.
- d) There are at most countably many $t \in \mathbb{C}$ such that $\rho_t > \mathcal{R}_g$.
- e) If $|z| < \rho_t$ for uncountably many $t \in \mathbb{C}$, then $\psi(t, z) \neq 0$ for all $t \in \mathbb{C}$.

Proof: a) Suppose $\rho_t \neq 0$ for some $t \neq 0$. Since $\psi(t,0) = 1$ there is a δ , $0 < \delta < \rho_t$, such that $|\psi(t,z)| > 0$ for $|z| \leq \delta$. Now Theorem 4.4.1 $f \Rightarrow a$ with $\alpha_n = \delta^n$ implies that $(g_n)_{n \in \mathbb{N}} \in \ell_1(\alpha)$, hence $\mathcal{R}_g \geq \delta > 0$. b) This follows from Theorem 4.4.1 $a \Rightarrow b$.

c) Let $t \in \mathbb{C} \setminus \{0\}$ be arbitrary. We first prove $\mathcal{R}_g \leq \nu_t$. If $|z| < \mathcal{R}_g$, then $(g_n)_{n \in \mathbb{N}} \in \ell_1(\alpha)$ with $\alpha_n = |z|^n$. It follows from Theorem 4.4.1 that $\psi(t, z) = e^{t g(z)} \neq 0$ for $|z| < \mathcal{R}_g$. Hence, $\mathcal{R}_g \leq \nu_t$.

The reverse inequality $\mathcal{R}_g \geq \nu_t$ follows from Theorem 4.4.1 f \Rightarrow a.

d) First recall that an analytic function can have at most finitely many zeros on a compact set ([213, Corollary to Theorem 10.18, p. 226]). Suppose that $\rho_s > \mathcal{R}_g$ for some $s \in \mathbb{C} \setminus \{0\}$. We first prove that $\psi(s, .)$ has at least one zero on $|z| = \mathcal{R}_g$. If $\psi(s, z) \neq 0$ for all $|z| = \mathcal{R}_g$, then $\psi(s, z) \neq 0$ for all $|z| < \mathcal{R}_g + \eta$ for some $\eta > 0$ since $\rho_s > \mathcal{R}_g$. Thus $\nu_s > \mathcal{R}_g$, which is impossible by c. Therefore we may write $\psi(s, z) = f_s(z) \prod_{j=1}^k (1 - \alpha_j z)^{r_j}$ with $|1/\alpha_j| = \mathcal{R}_g$ and $r_j \in \mathbb{N}$, $j = 1, \ldots, k$. There exists $\delta > 0$ such that $\psi(s, .)$ has finitely many zeros on $\{z : \mathcal{R}_g \leq |z| \leq \mathcal{R}_g + \delta < \rho_s\}$. Hence f_s is a non-vanishing analytic function on $|z| < \mathcal{R}_g + \delta_1$ for some $\delta_1 > 0$. Since $f_s(0) = 1$, Theorem 4.2.10 yields a unique analytic function h such that h(0) = 0 and $f_s(z) = e^{h(z)}$ for $|z| < \mathcal{R}_g + \delta_1$. Hence, $\psi(s, z) = \exp\left\{h(z) + \sum_{j=1}^k r_j \log(1 - \alpha_j z)\right\}$ for $|z| < \mathcal{R}_g$, where log denotes the principal branch of the logarithm on $\mathbb{C} \setminus (-\infty, 0]$. By Lemma 4.2.1a and Theorem 4.4.1, $h(z) + \sum_{j=1}^k r_j \log(1 - \alpha_j z) = s g(z)$ for $|z| < \mathcal{R}_g$. It follows that

$$\begin{split} \psi(t,z) &= \exp \left\{ t/s \, h(z) \; + \; t/s \, \sum_{j=1}^{k} \; r_{j} \, \log \left(1 - \alpha_{j} \, z\right) \right\} \\ &= \exp \; \left(t/s \, h(z) \right) \, \prod_{j=1}^{k} \; (1 - \alpha_{j} \, z)_{r_{j} \, t/s} \end{split}$$

for all $t \in \mathbb{C}$ and all $|z| < \mathcal{R}_g$. We conclude from the analyticity of h on $|z| < \mathcal{R}_g + \delta_1$, that $\rho_t > \mathcal{R}_g$ if and only if $t/s \in \mathbb{N}$. e) It follows from d) that $|z| < \mathcal{R}_g$, hence $\psi(t, z) \neq 0$ by c).

4.5 Two-sided sequences of functions of convolution type

In this section we will study a two-sided analogue of sequences of polynomials of convolution type.

Definition 4.5.1 Let $(q_n)_{n \in \mathbb{Z}}$ be a two-sided sequence of Lebesgue measurable functions on $[0, \infty)$ such that not all functions q_n are identically zero. Then $(q_n)_{n \in \mathbb{Z}}$ is said to be a two-sided sequence of convolution type if

$$\sum_{k=-\infty}^{\infty} |q_k(t) q_{n-k}(s)| < \infty \text{ for all } s, t \ge 0$$

and

$$q_n(t+s) = \sum_{k=-\infty}^{\infty} q_k(t) q_{n-k}(s) \text{ for all } s, t \ge 0.$$

Notation We write $\varphi(t,z) := \sum_{n=-\infty}^{\infty} q_n(t) z^n$ whenever this series converges absolutely. Note that $\varphi(t+s,z) = \varphi(t,z) \varphi(s,z)$.

Contrary to the one-sided case, due to convergence problems there seems to be no algebraic theory for two-sided sequences of convolution type. Therefore our policy is to impose several analytical conditions on $\varphi(t, z)$ and study the consequences. Note that $(J_n)_{n\in\mathbb{Z}}$, where J_n is the Bessel function of the first kind of index n, is an example of a two-sided sequence of convolution type (cf. [184, Section 62, Theorem 39]).

For later use we state the following lemma.

Lemma 4.5.2 If $(c_n)_{n \in \mathbb{Z}}$ is a two-sided sequence of complex numbers and $(c_n \mathbb{R}^n)_{n \in \mathbb{Z}} \in \ell_1(\mathbb{Z})$ for some $\mathbb{R} > 0$, then $\sum_{n=-\infty}^{\infty} \sum_{k=0}^{\infty} \frac{t^k}{k!} c_n^{k*} \mathbb{R}^n$ converges absolutely for all $t \in \mathbb{C}$. In particular, $\sum_{k=0}^{\infty} \frac{t^k}{k!} c_n^{k*}$ converges absolutely for all $t \in \mathbb{C}$.

Proof: This follows from

$$\sum_{n=-\infty}^{\infty} \sum_{k=0}^{\infty} \left| \frac{t^k}{k!} c_n^{k*} R^n \right| \le \sum_{k=0}^{\infty} \left| \frac{t^k}{k!} \sum_{n=-\infty}^{\infty} |c_n^{k*}| R^n \right| \le \sum_{k=0}^{\infty} \left| \frac{t^k}{k!} \right| \left\{ \sum_{n=-\infty}^{\infty} |c_n| R^n \right\}^k < \infty.$$

We begin with demanding $\varphi(t,.)$ to be an invertible element of the Wiener algebra \mathcal{W} , i.e. $\varphi(t,z) \neq 0$ for all $z \in \mathbb{T}$ (see Section 4.3).

Theorem 4.5.3 Let $(q_n)_{n\in\mathbb{Z}}$ be a two-sided sequence of convolution type. If $\varphi(t,.) \in inv \mathcal{W}$ for all $t \geq 0$, then there exists an $h \in \mathcal{W}$ such that $\varphi(t,z) = e^{t h(z)}$ for all |z| = 1. In particular, there exists a two-sided sequence $(z_n)_{n\in\mathbb{N}} \in \ell_1(\mathbb{Z})$ such that $q_n(t) = \sum_{j=0}^{\infty} h_n^{j*} \frac{t^j}{j!}$.

Proof: Define φ_1 by $\varphi_1(t,\theta) := \varphi(t,e^{i\theta})$ $(t \ge 0, \theta \in \mathbb{R})$. The measurability of $\varphi(.,z)$ and [118, Corollary to Theorem 4.17.3, p. 145] or [3, Theorem 4, p. 56] yield the existence of complex numbers $\chi(\theta)$ $(\theta \in \mathbb{R})$ such that

$$\varphi_1(t,\theta)/\varphi_1(t,0) = \exp\left(t\,\chi(\theta)\right) \tag{4.2}$$

We now show that ind $\varphi(t,.) = 0$ for all $t \ge 0$. Recall from Lemma 4.3.2 that the index of a non-vanishing continuous function on \mathbb{T} is always an integer. If ind $\varphi(t_0,.) \ne 0$ for some $t_0 \ge 0$, then ind $\varphi(t_0/n,.) \not\in \mathbb{Z}$ for n large enough by Lemma 4.3.2, which is impossible. It follows from Theorem 4.3.5 that there exist functions $\gamma(t,.) \in \mathcal{W}$ with $\gamma(t,1) = 0$ such that

$$\varphi_1(t,\theta)/\varphi_1(t,0) = \exp\left(\gamma(t,e^{i\theta})\right).$$
 (4.3)

Define continuous functions $\tilde{\gamma}(t,.)$ on \mathbb{R} by $\tilde{\gamma}(t,\theta) := \gamma(t,e^{i\theta})$. We get from (4.2) and (4.3):

$$\tilde{\gamma}(t,\theta) = t\,\chi(\theta) + k(t,\theta)\,2\pi i \tag{4.4}$$

with $k(t, \theta) \in \mathbb{Z}$. From (4.4) with t = 1 we get

$$\tilde{\gamma}(t,\theta) = t\,\tilde{\gamma}(1,\theta) + k(t,\theta) - t\,k(1,\theta)\,2\pi i \tag{4.5}$$

From (4.5), the continuity of $\gamma(t, .)$ and $\gamma(t, 0) = 0$ we obtain $k(t, \theta) - t k(1, \theta) = 0$ 0. Hence,

$$\varphi_1(t,\theta)/\varphi_1(t,0) = e^{(t\,\tilde{\gamma}(1,\theta))}.\tag{4.6}$$

From (4.6) and the measurability of $\varphi_1(t,0)$:

$$\varphi_1(t,\theta) = \exp \{at + t\,\tilde{\gamma}(1,\theta)\}. \tag{4.7}$$

Setting $h_0 := a$ and letting h_n be the n^{th} Fourier coefficient of $\gamma(1, .)$, we arrive at $\varphi(t, e^{i\theta}) = \exp \left\{ t \sum_{n=-\infty}^{\infty} h_n e^{in\theta} \right\}$ with $\sum_{n=-\infty}^{\infty} |h_n| < \infty$. \Box

Remarks 4.5.4 a) Using Theorem 4.5.3 we can give an interesting proof of Theorem 4.4.1 $e \Rightarrow f$: extend the sequences $(q_n)_{n \in \mathbb{N}} t$ to elements of $\ell_1(\mathbb{Z})$ by setting $q_n(t) = 0$ for n < 0. By Theorem 4.5.3, there exists a sequence $(c_n)_{n\in\mathbb{Z}}\in\ell_1(\mathbb{Z})$ such that

$$\sum_{n=-\infty}^{\infty} q_n(t) e^{in\theta} = \exp \left\{ t \sum_{n=-\infty}^{\infty} c_n e^{in\theta} \right\} = \sum_{n=-\infty}^{\infty} e^{in\theta} \sum_{k=0}^{\infty} \frac{t^k}{k!} c_n^{k*}$$

(Lemma 4.5.2 allows us to change the order of summation). Unicity of Fourier coefficients yields:

$$\sum_{k=0}^{\infty} \frac{t^k}{k!} c_n^{k*} = 0 \text{ for } n < 0; \sum_{k=0}^{\infty} \frac{t^k}{k!} c_n^{k*} = q_n(t) \text{ for } n \ge 0$$
(4.8)

Because (4.8) holds for all $t \ge 0$, we must have $c_n = 0$ for n < 0. Hence, $\sum_{n=0}^{\infty} q_n(t) z^n = \exp \{t \sum_{n=0}^{\infty} c_n z^n\} \ne 0$ for all $|z| \le 1$. b) It follows from $\varphi(t, e^{i\theta}) = \exp \{t \sum_{n=-\infty}^{\infty} h_n e^{in\theta}\}$ that for all $s \ge 0$

$$\lim_{t \to s} \sum_{n = -\infty}^{\infty} |q_n(t) - q_n(s)| = 0$$
(4.9)

If we weaken the assumptions of Theorem 4.5.2 by allowing $\varphi(t, .)$ to vanish, then (4.9) still holds for all s > 0 by [118, Theorem 9.3.1, p. 280].

7

Notation If $0 < a < b < \infty$, then $\mathcal{A}(a,b) := \{z \in \mathbb{C} : a < |z| < b\}$ and $\overline{\mathcal{A}}(a,b) := \{z \in \mathbb{C} : a \leq |z| \leq b\}.$

We denote the Banach algebra of all Laurent series that converge absolutely on $\overline{\mathcal{A}}(a, b)$ by $\mathcal{W}_{a,b}$.— Addition and multiplication are the usual addition and multiplication of series. The norm on $\mathcal{W}_{a,b}$ is defined by $\|\sum_{n=-\infty}^{\infty} x_n z^n\| :=$ $\max \{\sum_{n=-\infty}^{\infty} |x_n| a^n, \sum_{n=-\infty}^{\infty} |x_n| b^n\}.$

It is easy to see that $\mathcal{W}_{a,b}$ is complete and that the polynomials in z and 1/z are dense in $\mathcal{W}_{a,b}$.

The unit element of $\mathcal{W}_{a,b}$ is the Laurent series with $x_0 = 1$ and $x_n = 0$ for $n \neq 0$.

Lemma 4.5.5 The complex homomorphisms of $\mathcal{W}_{a,b}$ are point evaluations on $\overline{\mathcal{A}}(a,b)$.

Proof: Let $\Lambda \in \mathcal{M}(\mathcal{W}_{a,b})$ be arbitrary. From $\|\Lambda\| = 1$ ([212, Proposition 10.6 and Theorem 10.7]) we infer for the polynomial z that $|\Lambda(z)| \leq b$ and $|\Lambda(1/z)| \leq \|\Lambda\| \|z^{-1}\| = \|z^{-1}\| = a^{-1}$. Since 1/z is inverse to z, $|\Lambda(z)| = 1/|\Lambda(1/z)| \geq a$. Thus $\Lambda(z) = z_0$ for some $z_0 \in \overline{\mathcal{A}}(a, b)$. Hence, if p is a polynomial in z and 1/z, then $\Lambda(p) = p(\Lambda(z)) = p(z_0)$. Since the polynomials in z and 1/z are dense in $\mathcal{W}_{a,b}$, we conclude that $\Lambda(f) = f(z_0)$ for every $f \in \mathcal{W}_{a,b}$.

Theorem 4.5.6 Let $a, b \in \mathbb{R}$ (0 < a < b) and let $(q_n)_{n \in \mathbb{Z}}$ be a two-sided sequence of convolution type. If $\varphi(t, .) \in inv \mathcal{W}_{a,b}$ for all $t \geq 0$, then there exists a Laurent series $\sum_{n=-\infty}^{\infty} g_n z^n \in \mathcal{W}_{a,b}$ such that

$$\varphi(t,z) = exp \left\{ t \sum_{n=-\infty}^{\infty} g_n z^n \right\}$$

for all $z \in \overline{\mathcal{A}}(a,b)$. In particular, $q_n(t) = \sum_{k=0}^{\infty} g_n^{k*} \frac{t^k}{k!}$.

Proof: Let $r \in [a, b]$ be arbitrary. Then $\varphi(t, r e^{i\theta}) \neq 0$ for $\theta \in [-\pi, \pi]$. By Theorem 4.5.3 there exists $(c_n(r))_{n \in \mathbb{Z}} \in \ell_1(\mathbb{Z})$ such that

$$\varphi(t, r e^{i\theta}) = \exp \left\{ t \sum_{n=-\infty}^{\infty} c_n(r) e^{in\theta} \right\}$$
(4.10)

Define $g_n(r) := c_n(r) r^{-n}$. Then $(g_n r^n)_{n \in \mathbb{Z}} \in \ell_1(\mathbb{Z})$. We will now prove that $g_n(r)$ does not depend on r. By Lemma 4.5.2, we may change the order of summation in (4.10) which yields $q_n(t) = \sum_{k=0}^{\infty} g_n(r)^{k*} \frac{t^k}{k!}$. Since r was arbitrary and the right-hand side series defines a holomorphic function of t, we conclude that $g_n(r)$ does not depend on r. Define $g_n := g_n(a)$. Hence q_n has the form indicated above. Moreover, $(g_n r^n)_{n \in \mathbb{Z}} \in \ell_1(\mathbb{Z})$ for all $r \in [a, b]$ and thus (4.10) yields $\varphi(t, z) = \exp \{t \sum_{n=-\infty}^{\infty} g_n z^n\}$ for all $z \in \overline{\mathcal{A}}(a, b)$.

We now set out to prove the analogue of Theorem 4.5.6 for the open annulus. It turns out that two-sided sequences of convolution type possess a property that is analogous to the property for polynomials of convolution type as expressed in Theorem 4.4.10e. This property is stated in Theorem 4.5.8; the above mentioned analogue of Theorem 4.5.6 is Theorem 4.5.9.

Lemma 4.5.7 Let $(g_n)_{n\in\mathbb{Z}}$ be an arbitrary double-sided sequence of complex numbers such that $\sum_{n=-\infty}^{\infty} g_n z^n$ converges absolutely on the open annulus $\mathcal{A}(a,b)$. Define $q_n(t) := \sum_{k=0}^{\infty} g_n^{k*} \frac{t^k}{k!}$ for $t \in [0,\infty)$. Then $(q_n)_{n\in\mathbb{Z}}$ is a two-sided sequence of convolution type. If $z \in \mathcal{A}(a,b)$, then $\sum_{n=-\infty}^{\infty} q_n(t) z^n$ converges absolutely and does not vanish.

Proof: It follows from Lemma 4.5.2 that q_n $(n \in \mathbb{Z})$ is well-defined and that $\sum_{n=-\infty}^{\infty} q_n(t) z^n$ converges absolutely on $\mathcal{A}(a, b)$. Hence,

$$\sum_{n=-\infty}^{\infty} q_n(t) z^n = \sum_{n=-\infty}^{\infty} \sum_{k=0}^{\infty} g_n^{k*} \frac{t^k}{k!} z^n =$$
$$\sum_{k=0}^{\infty} \left\{ \sum_{n=-\infty}^{\infty} g_n^{k*} z^n \right\} \frac{t^k}{k!} = \exp \left\{ t \sum_{n=-\infty}^{\infty} g_n z^n \right\} \neq 0.$$

Theorem 4.5.8 Let $a, b \in \mathbb{R}$ (0 < a < b) and let $(q_n)_{n \in \mathbb{Z}}$ be a two-sided sequence of convolution type. If $\varphi(t, z)$ converges for all $z \in \mathcal{A}(a, b)$ and all $t \geq 0$, then $\varphi(t, z) \neq 0$ for all $t \geq 0$ and all $z \in \mathcal{A}(a, b)$.

Proof: Suppose there are $t_0 > 0$ and $z_0 \in \mathcal{A}(a,b)$ such that $\varphi(t_0,z_0) = 0$. It follows from [118, Theorem 4.17.1, p. 144], that $\varphi(t,z_0) = 0$ for all t > 0. Choose c, d with $a \leq c < |z_0| < d \leq b$ such that $\varphi(t_0, z) \neq 0$ for all t > 0 and all $z \in \mathcal{A}(c, |z_0|) \bigcup \mathcal{A}(|z_0|, d)$. This is possible since the functions $\varphi(t, .)$ are analytic and not identically zero. Choose c_1, c_2, d_1 and d_2 with $c < c_1 < c_2 < |z_0|$ and $|z_0| < d_1 < d_2 < d$. An application of Theorem 4.5.6 to the functions q_n on $\overline{\mathcal{A}}(c_1, c_2)$ yields a sequence $(g_n)_{n \in \mathbb{N}}$ of complex numbers such that $q_n(t) = \sum_{k=0}^{\infty} g_n^{k*} \frac{t^k}{k!}$. An application of Theorem 4.5.6 to the functions q_n on $\overline{\mathcal{A}}(d_1, d_2)$ yields a sequence $(h_n)_{n \in \mathbb{N}}$ of complex numbers such that $q_n(t) = \sum_{k=0}^{\infty} h_n^{k*} \frac{t^k}{k!}$. Differentiating with respect to t and substituting t = 0, we obtain $g_n = h_n$ for all $n \in \mathbb{Z}$. This implies that $\sum_{n=-\infty}^{\infty} g_n z^n$ converges for all $z \in \overline{\mathcal{A}}(c_1, c_2) \bigcup \overline{\mathcal{A}}(d_1, d_2)$, hence for all $z \in \overline{\mathcal{A}}(c_1, d_2)$. It follows from Lemma 4.5.7 that $\varphi(t_0, z) \neq 0$ for all $z \in \overline{\mathcal{A}}(c_1, d_2)$, which contradicts $\varphi(t_0, z_0) = 0$.

Theorem 4.5.9 Let $a, b \in \mathbb{R}$ (0 < a < b) and let $(q_n)_{n \in \mathbb{Z}}$ be a two-sided sequence of convolution type. Suppose $\varphi(t, z)$ converges absolutely for all $t \ge 0$ and all $z \in \mathcal{A}(a, b)$. Then there exists a Laurent series $\sum_{n=-\infty}^{\infty} g_n z^n$ that absolutely converges on $\mathcal{A}(a, b)$ and satisfies $\varphi(t, z) = \exp \{t \sum_{n=-\infty}^{\infty} g_n z^n\}$ for all $t \ge 0$ and for all $z \in \mathcal{A}(a, b)$. In particular, $\varphi(t, z)$ does not vanish on $\mathcal{A}(a, b)$ and $q_n(t) = \sum_{k=0}^{\infty} g_n^{k*} \frac{t^k}{k!}$.
Proof: It follows from Theorem 4.5.8 that $\varphi(t,z) \neq 0$ for all $t \geq 0$ and all $z \in \mathcal{A}(a, b)$. Applying Theorem 4.5.6 to $\mathcal{W}_{a+1/n, b-1/n}$ for all $n \in \mathbb{N}$ such that a + 1/n < b - 1/n, we obtain Laurent series $h_n \in \mathcal{W}_{a+1/n,b-1/n}$ such that $\varphi(t,z) = \exp (t h_n(z))$. Since $\exp (t h_n(z)) = \exp (t h_m(z))$ for all $t \in [0,\infty)$ on a circular region, $h_n(z) = h_m(z)$ for all z in their common domain by Lemma 4.2.1a. Hence, all the Laurent series h_n are identical. If we set $g := h_1$, then g converges absolutely on $\mathcal{A}(a,b)$ and $\varphi(t,z) = e^{t g(z)}$ for all $t \in \mathbb{C}$ and all $z \in \mathcal{A}(a, b)$. Π

We denote the Banach algebra of all Laurent series that are absolutely convergent on $\mathcal{A}(a,b)$ and have a continuous extension to $\overline{\mathcal{A}}(a,b)$ by $\mathcal{L}_{a,b}$. Addition and multiplication are defined pointwise. The norm is the supremum norm of the function corresponding to the Laurent series. Since the limit of a uniformly convergent sequence of continuous (holomorphic) functions is again continuous (holomorphic), $\mathcal{L}_{a,b}$ is complete. The unit element of $\mathcal{L}_{a,b}$ is the Laurent series with $x_0 = 1$ and $x_n = 0$ for $n \neq 0$.

Lemma 4.5.10 The complex homomorphisms of $\mathcal{L}_{a,b}$ are point evaluations on $\mathcal{A}(a,b).$

Proof: It suffices to show that the polynomials in z and 1/z are dense in $\mathcal{L}_{a,b}$, since we can then copy the proof of Lemma 4.5.5. If $\sum_{n=-\infty}^{\infty} a_n z^n \in \mathcal{L}_{a,b}$, then $\sum_{n=0}^{\infty} a_n z^n$ and $\sum_{n=-\infty}^{-1} a_n z^n$ can be approximated uniformly on $\overline{\mathcal{A}}(a,b)$ by polynomials in z, polynomials in 1/z respectively. This shows that the polynomials in z and 1/z are dense in $\mathcal{L}_{a,b}$.

Theorem 4.5.11 Let $a, b \in \mathbb{R}$ (0 < a < b) and let $(q_n)_{n \in \mathbb{Z}}$ be a two-sided sequence of convolution type. If $\varphi(t, z) \in inv \mathcal{L}_{a,b}$ for all $t \geq 0$, then there exists an $h \in \mathcal{L}_{a,b}$ such that $\varphi(t,z) = e^{t h(z)}$ for all $t \ge 0$ and all $z \in \overline{\mathcal{A}}(a,b)$. In particular, $q_n(t) = \sum_{k=0}^{\infty} g_n^{k*} \frac{t^k}{k!}$.

Proof: First note that Theorem 4.5.9 implies the existence of a Laurent series h which absolutely converges on $\mathcal{A}(a,b)$ and satisfies $\varphi(t,z) = e^{t h(z)}$ for all $z \in \mathcal{A}(a,b).$

Choose c, d such that a < c < d < b. Consider all $0 < \lambda < 1$ such that $\lambda c > a$. Write $\varphi_{\lambda}(t,z) := \varphi(t,\lambda z)$ for these λ . It follows from Theorem 4.5.6 that $\varphi_{\lambda}(1,.) \in \exp \ \mathcal{W}_{c,b} \subset \exp \ \mathcal{L}_{c,b}.$ Since $\varphi(1,.) \in \operatorname{inv} \ \mathcal{L}_{c,b}$ and $\lim_{\lambda \uparrow 1} \varphi_{\lambda}(1,.) =$ $\varphi(1,.)in\mathcal{L}_{c,b}$, the second statement of Theorem 4.1.4 implies that $\varphi(1,.) \in$ exp $\mathcal{L}_{c,b}$. In a similar way we see that $\varphi(1,.) \in \exp \mathcal{L}_{a,d}$. It follows from Lemma 4.2.1a that $\varphi(1,.) \in \exp \mathcal{L}_{a,b}$, i.e. there exists an $H \in \mathcal{L}_{a,b}$ such that $\varphi(1,z) = e^{(z)}$ for $z \in \mathcal{A}(a,b)$. It follows from Lemma 4.2.1a that H and h differ by a constant. We conclude that $h \in \mathcal{L}_{a,b}$.

For the last statement, see the end of the proof of Theorem 4.5.6.

Banach algebras

Chapter 5

Central limit theorems and infinite divisibility

In this chapter we study random variables Y_n $(n \in \mathbb{N})$ with probability generating function $q_n(\lambda x)/q_n(\lambda)$, where $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$. For an interpretation of the random variables Y_n in terms of a compound Poisson process, see [222]. Canfield [47, 48] proved a central limit theorem for $(Y_n)_{n\in\mathbb{N}}$ in case g(z) :=

 $\sum_{n=0}^{\infty} g_n z^n$ belongs to a class of entire functions including polynomials (see also [55, 216]). A central limit theorem for $(Y_n)_{n\in\mathbb{N}}$ in case g has a dominant logarithmic singularity on its circle of convergence can be found in [96, 97]. Stam [225] used renewal theory to obtain a central limit theorem. Moreover, in [222] he obtained results on the asymptotic behaviour of $q_n(x)/q_n(1)$. The main purpose of this chapter is to extend the results of [224].

Applications of these central limit theorems to asymptotic enumeration can be found in [47, 48, 96, 97, 214].

This chapter is organized as follows. Section 5.1 gives some auxiliary results that will be needed for the proof of the central limit theorem in Section 5.4. In Section 5.2 we determine the asymptotics of the polynomials q_n when g converges absolutely on its circle of convergence. In Section 5.3 we introduce the renewal approach to central limit theorems of [224] and show that his central limit theorem is a special case of the results of Section 5.4. Section 5.4 contains a central limit theorem for the case that g has a dominant logarithmic singularity on its circle of convergence. Finally, Section 5.5 deals with infinitely divisible probability measures on N. Using Sections 4.1 and 5.2, we give a new proof for a result of Embrechts and Hawkes [87] on the asymptotic behaviour of an infinitely divisible probability measure on N and its Lévy-measure.

Contents of chapter 5

5.1 Preliminaries.

- **5.2** Asymptotics when g converges on its circle of convergence.
- 5.3 Renewal theory.
- 5.4 Logarithmic singularities.
- 5.5 Infinitely divisible measures on N.

5.1 Preliminaries

This section contains several results that will be used in the next sections. For sake of brevity, we do not state these results in their most general form.

Lemma 5.1.1 a) If x > 1, then $\left(\binom{x+n-1}{n}\right)_{n \in \mathbb{N}}$ is an increasing sequence.

b) If 0 < x < 1, then $\binom{\binom{x+n-1}{n}}{n}_{n \in \mathbb{N}}$ is a decreasing sequence.

Proof: This follows from $\binom{x+n-1}{n} / \binom{x+n-2}{n-1} = \frac{x+n-1}{n} = 1 + \frac{x-1}{n}$.

Lemma 5.1.2 If $a_n := (\lambda \log n)^{1/2}$, then for all $\lambda, z > 0$ and all $k \in \mathbb{N}$:

$$\lim_{n \to \infty} e^{-a_n z} \left(\frac{\lambda e^{z/a_n} + n - k - 1}{n - k} \right) \left(\frac{\lambda + n - k - 1}{n - k} \right)^{-1} = e^{\frac{1}{2}z^2}.$$

For fixed z > 0, $e^{-a_n z} \begin{pmatrix} \lambda e^{z/a_{n+n-k-1}} \\ n-k \end{pmatrix} \begin{pmatrix} \lambda+n-k-1 \\ n-k \end{pmatrix}^{-1}$ is uniformly bounded for all $n, k \in \mathbb{N}$ with $0 \le k < n$.

Proof: We first prove the assertion for k = 0. Note that

$$e^{-a_n z} \binom{\lambda e^{z/a_n} + n - 1}{n} \binom{\lambda + n - 1}{n}^{-1} = e^{-a_n z} e^{z/a_n} \prod_{j=1}^{n-1} \left(1 + \frac{\lambda}{\lambda + j} \left(e^{z/a_n} - 1\right)\right).$$

After taking logarithms it suffices to prove

$$\lim_{n \to \infty} \left\{ z/a_n - a_n z + \sum_{j=1}^{n-1} \log \left(1 + \frac{\lambda}{\lambda+j} \left(e^{z/a_n} - 1 \right) \right) \right\} = \frac{1}{2} z^2$$

Expanding first the logarithms and then the exponential functions into Taylor polynomials, we obtain

$$\sum_{j=0}^{n} \log \left(1 + \frac{\lambda}{\lambda+j} \left(e^{z/a_n} - 1 \right) \right) \right) =$$

$$\sum_{j=0}^{n} \frac{\lambda}{\lambda+j} \left(e^{z/a_n} - 1 \right) \right) - \sum_{j=0}^{n} \frac{\frac{1}{2}}{(1+\theta_{n,j})^2} \left(\frac{\lambda}{\lambda+j} \right)^2 \left(e^{z/a_n} - 1 \right)^2,$$

with $0 < \theta_{n,j} < \frac{\lambda}{\lambda+j} \left(e^{z/a_n} - 1 \right)$ for $n \to \infty$. The last term vanishes as $n \to \infty$, because $\sum_{j=0}^{\infty} \left(\frac{\lambda}{\lambda+j}\right)^2$ converges and $\lim_{n\to\infty} z/a_n = 0$. Now we expand the first term as

$$\left(z/a_n + \frac{1}{2} \left(z/a_n\right)^2 + o\left(a_n^{-2}\right)\right) \sum_{j=0}^n \frac{\lambda}{\lambda+j}.$$

Since $\sum_{i=0}^{n} \frac{\lambda}{\lambda+j} = a_n^2 + O(1)$, it follows that

$$\lim_{n \to \infty} \left[\left(z/a_n + \frac{1}{2} (z/a_n)^2 + o(a_n^{-2}) \right) \sum_{j=0}^n \frac{\lambda}{\lambda+j} \right] - a_n z = \frac{1}{2} z^2.$$

This completes the proof for k = 0. Because

$$\binom{\lambda e^{z/a_n} + n - k - 1}{n - k} \binom{\lambda + n - k - 1}{n - k}^{-1} = \frac{\lambda + n - k}{e^{z/a_n} + n - (k - 1) - 1} \binom{\lambda + n - (k - 1) - 1}{n - (k - 1)} \binom{\lambda + n - (k - 1) - 1}{n - (k - 1) - 1}^{-1},$$

induction on k yields the first assertion.

For the second assertion note that $e^{z/a_n} \leq e^{z/a_{n-k}}$, $e^{-a_n z} \leq e^{-a_{n-k} z}$, and that $\binom{x+n-1}{n} < \binom{y+n-1}{n}$ for 0 < x < y. Hence, $e^{-a_n z} \binom{\lambda e^{z/a_n} + n - k - 1}{n-k} \leq e^{-a_{n-k} z} \binom{\lambda e^{z/a_n} + n - k - 1}{n-k}$

Lemma 5.1.3 For all $\lambda \in \mathbb{C} \setminus \{-1, -2, ...\}$, we have $\lim_{n\to\infty} {\binom{\lambda+n-1}{n}} n^{1-\lambda} =$ 1 $\overline{\Gamma(\lambda)}$.

Proof: Since $\binom{\lambda+n-1}{n} = \frac{\Gamma(n+\lambda)}{\Gamma(n+1)\Gamma(\lambda)}$ for all $\lambda \in \mathbb{C} \setminus \{-1, -2, \dots\}$, the result follows from [68, sect. 27]

Lemma 5.1.4 If $\lambda > 0$ and $(\lambda - 1)\alpha > -1$, then

$$\sum_{j=0}^{m} {\binom{\lambda+j-}{j}}^{\alpha} \le C \, m^{1+(1-\lambda)\alpha},$$

where C depends on λ and α .

Proof: Since $\lambda > 0$ and $(\lambda - 1)\alpha > -1$, Lemma 5.1.3 yields $\sum_{j=0}^{m} {\binom{\lambda+j-}{j}}^{\alpha} = 1 + \sum_{j=0}^{m} {\binom{\lambda+j-}{j}}^{\alpha} \le 1 + C_1 \sum_{j=0}^{m} j^{(\lambda-1)\alpha}$. If $(\lambda - 1)\alpha \ge 0$, then

$$1 + C_1 \sum_{j=0}^m j^{(\lambda-1)\alpha} \le 1 + C_1 \int_0^m t^{(\lambda-1)\alpha} dt \le C_2 m^{1+(\lambda-1)\alpha}$$

If $-1 < (\lambda - 1)\alpha < 0$, then

$$1 + C_1 \sum_{j=0}^m j^{(\lambda-1)\alpha} \le 1 + C_1 \int_1^{m-1} t^{(\lambda-1)\alpha} dt \le C_3 m^{1+(\lambda-1)\alpha}.$$

We conclude this section with a useful lemma on convergence of moment generating functions.

Lemma 5.1.5 Let $a, b \in \mathbb{R}$ be arbitrary with a < b. If F_n $(n \in \mathbb{N})$ and F are probability distribution functions on the real line such that

$$\lim_{n \to \infty} \int_{-\infty}^{\infty} e^{zx} \, dF_n(x) = \int_{-\infty}^{\infty} e^{zx} \, dF(x)$$

for all $z \in (a, b)$, then F_n converges weakly to F as $n \to \infty$.

Proof: If a < 0 < b, then the result for arbitrary distribution functions follows from the proof of [71, Theorem 3]. Suppose 0 is not an interior point of (a,b). We will reduce this case to the case a < 0 < b. Choose an arbitrary $\xi \in (a,b)$. Define measures dG_n by $dG_n(x) := e^{\xi x} dF_n(x)$ for all $n \in \mathbb{N}$ and define dG by $dG(x) := e^{\xi x} dF(x)$. Then $\lim_{n\to\infty} \int_{-\infty}^{\infty} e^{vx} dG_n(x) = \lim_{n\to\infty} \int_{-\infty}^{\infty} e^{(v+\xi)x} dF_n(x) = \int_{-\infty}^{\infty} e^{(v+\xi)x} dF(x) = \int_{-\infty}^{\infty} e^{vx} dG(x)$ for all $v \in (a-\xi,b-\xi)$. Since $a-\xi < 0 < b-\xi$, it follows that $\lim_{n\to\infty} G_n(x) = G(x)$ for all continuity points x of G. Since $e^{\xi x}$ is continuous, it follows that $\lim_{n\to\infty} F_n(x) = F(x)$ for all continuity points x of F. Because F_n and F are probability distribution functions, it follows that F_n converges weakly to F as $n \to \infty$.

5.2 Asymptotics when g converges on its circle of convergence

Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$. In this section we study the asymptotic behaviour of $q_n(x)/q_n(1)$ as $n \to \infty$ in case $\sum_{n=0}^{\infty} g_n z^n$ converges absolutely on its circle of convergence. If $g_n \ge 0$ for all $n \in \mathbb{N}$, then the polynomials q_n have non-negative coefficients by Lemma 2.1.5, since $q_n(x) = \sum_{k=0}^{n} g_n^{k*} \frac{x^k}{k!}$ by Theorem

2.1.8. Hence, $q_n(x)/q_n(1)$ is the probability generating function of a discrete random variable. Stam [222, Theorem 4] has shown that in this case the only possible limit distribution without centering and scaling is a Poisson distribution shifted 1 to the right. We will extend this result to the case where the numbers g_n need not be non-negative. Of course, in this case $q_n(x)/q_n(1)$ need not be a probability generating function. The Banach algebra approach to subexponential distributions of [61] will be used and extended.

We start with stating and extending the results from [61] needed for the sequel. Recall that $\mathbb{N} = \{0, 1, ...\}$.

Definition 5.2.1 Let $(\mu_n)_{n \in \mathbb{N}}$ be a sequence of complex numbers such that

- 1. $\lim_{n\to\infty} \mu_n^{2*}/\mu_n = c$ exists and is finite
- 2. $\lim_{n\to\infty} \mu_{n+1}/\mu_n = 1/r$ exists and is positive
- 3. $\mu_n > 0$ for all $n \in \mathbb{N}$
- 4. $\sum_{n=0}^{\infty} \mu_n r^n < \infty.$

Define

$$\mathcal{U}_L := \{ (\nu_n)_{n \in \mathbb{N}} \subset \mathbb{C} \mid \lim_{n \to \infty} \nu_n / \mu_n \text{ exists} \}$$

and

$$\mathcal{U}_0 := \{ (\nu_n)_{n \in \mathbb{N}} \subset \mathbb{C} \mid \lim_{n \to \infty} \nu_n / \mu_n = 0 \}.$$

Condition 3) of definition 5.2.1 is missing in [61]. However, the proofs in [61] are not valid unless condition 3) is added. It is not clear how to remove condition 3).

For more information on sequences satisfying the conditions of Definition 5.2.1, see [86, sect. 2]. For example, it can be shown that $c = 2 \sum_{n=0}^{\infty} \mu_n r^n$ (see [86, Theorem 2.8]).

Parts a, b and c of the following theorem are taken from [61]; parts d and e are new.

- **Theorem 5.2.2** a) \mathcal{U}_L and \mathcal{U}_0 are Banach algebras when equipped with coordinatewise addition, convolution as multiplication and norm $\|\nu\| :=$ $M \sup_{n \in \mathbb{N}} \nu_n / \mu_n \mid$, where $M := u \, \mu_n^{2*} / \mu_n$. The sequence $1, 0, 0, \ldots$ is the unit element u of both \mathcal{U}_L and \mathcal{U}_0 .
 - b) If Λ is a complex homomorphism of \mathcal{U}_0 , then there exists $\lambda \in \{z \in \mathbb{C} : |z| \leq r\}$ such that $\Lambda(\nu) = \sum_{n=0}^{\infty} \nu_n \lambda^n$ for all $\nu \in \mathcal{U}_0$.
 - c) If Λ is a complex homomorphism of \mathcal{U}_L , then there exists $\lambda \in \{z \in \mathbb{C} : |z| \leq r\}$ such that $\Lambda(\nu) = \sum_{n=0}^{\infty} \nu_n \lambda^n$ for all $\nu \in \mathcal{U}_L$.
 - d) inv $\mathcal{U}_0 = exp \ \mathcal{U}_0$.

e) inv $\mathcal{U}_L = exp \ \mathcal{U}_L$.

Proof: a) See [61, Lemma 1].

b) See [61, Lemma 2].

c) See [61, Lemma 3].

d) The inclusion $\exp \mathcal{U}_0 \subset \operatorname{inv} \mathcal{U}_0$ follows from Remark 4.1.3a. Let $\nu \in \operatorname{inv} \mathcal{U}_0$ be arbitrary. Define $F : [0, 1] \to \operatorname{inv} \mathcal{U}_0$ by $F(t) := (t^n \nu_n)_{n \in \mathbb{N}}$. It follows from b) and Theorem 4.1.2 that $F(t) \in \operatorname{inv} \mathcal{U}_0$ for all $t \in [0, 1]$. We now show that F is continuous. If $s, t \in [0, 1]$, then $||F(s) - F(t)|| = M \sup_{n \in \mathbb{N}} |(s^n - t^n) \nu_n / \mu_n|$. Since $\lim_{n \to \infty} \nu_n / \mu_n = 0$, it follows that $\lim_{n \to \infty} ||F(s) - F(t)|| = 0$. Hence, $\nu \in \exp \mathcal{U}_0$ by Theorem 4.1.6 since $\nu_0 \neq 0$.

e) The inclusion $\exp \mathcal{U}_L \subset \operatorname{inv} \mathcal{U}_L$ follows from Remark 4.1.3a. Let $\nu \in \operatorname{inv} \mathcal{U}_L$ be arbitrary. By c) and theorem 4.1.2, $\sum_{n=0}^{\infty} \nu_n z^n \neq 0$ for all $|z| \leq r$. In particular, $\nu_0 \neq 0$. Thus, the Gelfand transform $\hat{\nu}$ of ν belongs to inv $\mathcal{C}(\mathcal{M})$. Since \mathcal{M} is homeomorphic to $\{z \in \mathbb{C} : |z| \leq r\}$, it follows from Theorem 4.2.5 that $\hat{\nu} \in \exp \mathcal{C}(\mathcal{M})$. By theorem 4.1.10, $\nu \in \exp \mathcal{U}_L$.

Remark 5.2.3 If $\nu \in \text{inv } \mathcal{U}_L$ and $\lim_{n\to\infty} \nu_n/\mu_n \neq 0$, then $F:[0,1] \to \text{inv } \mathcal{U}_0$, defined by $F(t) := (t^n \nu_n)_{n\in\mathbb{N}}$, is continuous for $0 \leq t < 1$ but discontinuous at t = 1. Thus the method of proof for Theorem 5.2.2d does not work for Theorem 5.2.2e.

The following theorem gives sufficient conditions for the convergence as $n \to \infty$ of $q_n(x)/q_n(1)$. If the polynomials q_n have non-negative coefficients, then $q_n(x)/q_n(1)$ is the probability generating function of a discrete random variable Y_n . Convergence of $q_n(x)/q_n(1)$ for all $x \in (0, 1]$ implies convergence in distribution of the random variables Y_n by the continuity theorem for probability generating functions (see [92, sect. XI.6]).

Theorem 5.2.4 was proved in [222] for q_n with non-negative coefficients (cf. Remark 5.2.5a).

Theorem 5.2.4 Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$ such that \mathcal{R}_g , the radius of convergence of $\sum_{n=0}^{\infty} g_n z^n$, is finite and positive. If an $x_0 \neq 0$ exists such that:

- 1. $q_n(x_0) > 0$ for all $n \in \mathbb{N}$
- 2. $\lim_{n \to \infty} q_n(2x_0)/q_n(x_0)$ exists and is finite
- 3. $\lim_{n \to \infty} q_{n+1}(x_0)/q_n(x_0) = 1/\mathcal{R}_g$

4.
$$\sum_{n=0}^{\infty} q_n(x_0) \mathcal{R}_g^n < \infty$$

5.
$$\sum_{n=0}^{\infty} q_n(x_0) z^n \neq 0 \text{ for } |z| = \mathcal{R}_g,$$

5.2 Asymptotics when g converges on its circle of convergence 115

then
$$\sum_{n=0}^{\infty} |g_n| \mathcal{R}_g^n < \infty$$
 and $\lim_{n \to \infty} q_n(x)/q_n(1) = x e^{(x-1)g(\mathcal{R}_g)}$ for all $x \in \mathbb{C}$.

Proof: By definition of convolution type, the sequence $(q_n(2x_0))_{n\in\mathbb{N}}$ is the twofold convolution of $(q_n(x_0))_{n\in\mathbb{N}}$. Define $\mu_n := q_n(x_0)$. It follows from Theorem 5.2.2 that \mathcal{U}_L and \mathcal{U}_0 are Banach algebras. It follows from Theorem 4.4.1 with $\alpha_n = \mathcal{R}_g^n$ that $\sum_{n=0}^{\infty} \mu_n z^n \neq 0$ for $|z| \leq \mathcal{R}_g$. Hence, $(\mu_n)_{n\in\mathbb{N}} \in inv \mathcal{U}_L$ by Theorem 5.2.2c and $(\mu_n)_{n\in\mathbb{N}} \in exp \mathcal{U}_L$ by Theorem 5.2.2e. It follows from Lemma 4.2.1 and Theorem 4.4.1 that $(g_n)_{n\in\mathbb{N}} \in \mathcal{U}_L$. Obviously, also $(x g_n)_{n\in\mathbb{N}} \in \mathcal{U}_L$ for all $x \in \mathbb{C}$ and therefore $(q_n(x))_{n\in\mathbb{N}} \in \mathcal{U}_L$ for all $x \in \mathbb{C}$. Now [61, Lemma 5] implies that $\lim_{n\to\infty} q_n(k)/q_n(1) = k e^{(k-1)g(\mathcal{R}_g)}$ for all $k \in \mathbb{N}$. Hence,

$$\lim_{n \to \infty} \frac{q_n(k/m)}{q_n(1)} = \lim_{n \to \infty} \frac{q_n(k/m)}{q_n(1/m)} \frac{q_n(1/m)}{q_n(1)} = \frac{k}{m} e^{((k/m) - 1) g(\mathcal{R}_g)}$$

for all $k, m \in \mathbb{N}$. By continuity, $\lim_{n \to \infty} q_n(x)/q_n(1) = x e^{(x-1)g(\mathcal{R}_g)}$ for all $x \ge 0$. Applying the theorem to $(q_n(ax))_{n \in \mathbb{N}}$ for suitable a with |a| = 1, we obtain $\lim_{n \to \infty} q_n(x)/q_n(1) = x e^{(x-1)g(\mathcal{R}_g)}$ for all $x \in \mathbb{C}$.

Remark 5.2.5 a) Let us compare Theorem 5.2.4 with [222, Theorem 4]. The conditions in the Stam theorem are: $g_n \ge 0$ for all $n \in \mathbb{N}, g_1 \ne 0$ (since $(q_n)_{n \in \mathbb{N}}$ is a basic sequence; cf. Theorems 2.2.17 and 2.1.12b), $\mathcal{R}_g < \infty, \sum_{n=0}^{\infty} g_n \mathcal{R}_g^n < \infty$ and the existence of a nonzero limit of $q_n(x)/q_n(1)$ for $0 \le x < 1$. In particular, these conditions imply 1), 2), 4) and 5) of Theorem 5.2.4 for $x_0 = \frac{1}{2}$. It follows from [86, Theorem 2.8 and Lemma 2.10] that 3) is also satisfied for $x_0 = \frac{1}{2}$. Hence, Theorem 5.2.4 is more general than the Stam theorem.

b) It is shown in [222, Theorem 3] that if $g_n \ge 0$ for all $n \in \mathbb{N}$, then $g(\mathcal{R}_g) = \infty$ implies $\liminf_{n \in \mathbb{N}} q_n(x)/q_n(1) = 0$ for $0 \le x < 1$. Thus, $g(\mathcal{R}_g) < \infty$ is necessary for the existence of a nonzero limit for $q_n(x)/q_n(1)$. Since the proof of Stam uses non-negativity in an essential way, it is not clear whether the above also holds in the general case.

c) It is possible to avoid the continuity argument at the end of the proof of Theorem 5.2.4 when $g_n > 0$ for $n \ge 1$. In order to do so, first note that $\lim_{n\to\infty} q_n(x)/g_n \ne 0$ for all $x \ne 0$ (cf. the proof of Theorem 5.5.4). This allows us to write $\lim_{n\to\infty} q_n(x)/q_n(1) = \lim_{n\to\infty} q_n(x)/g_n \lim_{n\to\infty} g_n/q_n(1)$. It follows from [86, Theorem 2.9iv] that $g_n^{2*} \sim 2g_n \ (n \to \infty)$. Moreover, $\lim_{n\to\infty} g_n/g_{n+1} = \mathcal{R}_g$ by [86, Theorem 2.8 and Lemma 2.10]. Now consider the Banach algebra \mathcal{U}_L with $\mu_n = g_n$. The theorem now follows from [61, Formula (2)] with $\varphi(z) = e^{xz}$ (cf. [61, Remark 2]).

Example 5.2.6 We now apply Theorem 5.2.4 to the Abel polynomials $x (x - an)^{n-1}/n!$ with a < 0. It follows from Remark 2.1.10c that $g_n = (-an)^{n-1}/n!$. Thus

$$\mathcal{R}_g = \lim_{n \to \infty} \frac{g_n}{g_{n+1}} = \lim_{n \to \infty} \left(\frac{|an|^{n-1}}{n!} \right)^{-1/n} = (|a|e)^{-1}$$

 and

$$\sum_{n=0}^{\infty} |g_n| \, (|a|e)^{-n} < \infty.$$

Moreover, a simple computation yields

$$\lim_{n \to \infty} q_n(x)/q_n(1) = x e^{-(x-1) a^{-1}}.$$

Together with Theorem 5.2.4 this yields $g(\mathcal{R}_g) = -a^{-1}$. Thus we obtain $\sum_{n=0}^{\infty} \frac{n^{n-1}}{n! e^n} = 1.$

5.3 Renewal theory

In the previous section we considered limit behaviour without centering or scaling of random variables with probability generating function $q_n(x)/q_n(1)$. We used the representation $q_n(x) = \sum_{k=0}^n g_n^{k*} \frac{x^k}{k!}$ and studied the behaviour of $\sum_{n=0}^{\infty} g_n z^n$. In [224] Stam introduced the idea to use the representation $q_n(x) = \sum_{k=0}^n f_n^{k*} \binom{x+k-1}{k}$ (cf. Theorem 2.3.10 and Example 2.2.16c) for studying limit behaviour with centering and scaling. The polynomials $\binom{x+n-1}{n}$ have interesting properties. Firstly, $\binom{x+n-1}{n}$ is the probability generating function of the number of cycles in a random permutation of $\{1, ..., n\}$ and satisfies a central limit theorem (see [92, Chapter X.6b], [214, Chapter 5, Theorem 1.1] or apply Lemmas 5.1.2 and 5.1.5). Secondly, the sequence $\binom{x+n-1}{n}_{n \in \mathbb{N}}$ is the unique sequence of polynomials of convolution type with $q_n(1) = 1$ for all $n \in \mathbb{N}$ (see Theorem 3.1.1). The purpose of Sections 5.3 and 5.4 is to extend the results of [224] to the case where f_n is not necessarily non-negative.

The following theorem shows the connection of the Stam approach with renewal theory.

Theorem 5.3.1 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n \in \mathbb{N}}$ and let $(f_n)_{n \in \mathbb{N}}$ be the unique sequence of complex numbers such that $q_n(x) = \sum_{k=0}^n f_n^{k*} \binom{x+k-1}{k}$. Let \mathcal{R}_f , \mathcal{R}_g be the radius of convergence of $f(z) = \sum_{n=0}^{\infty} f_n z^n$, $g(z) = \sum_{n=0}^{\infty} g_n z^n$ respectively. Then:

- a) $q_n(1) = \sum_{k=0}^n f_n^{k*}$
- b) $g_n = \sum_{k=1}^n f_n^{k*}$
- c) $f_0 = 0$; $f_n = -q_n(-1)$ for $n \ge 1$
- d) $f_n \ge 0$ for all $n \in \mathbb{N} \Rightarrow g_n \ge 0$ for all $n \in \mathbb{N}$
- e) the following formal generating function identity holds:

$$\sum_{n=0}^{\infty} q_n(x) z^n = (1 - f(z))^{-x}$$

- f) $\mathcal{R}_{q} = \min\{|z| : |z| \le \mathcal{R}_{f} \text{ and } f(z) = 1\}$
- g) $\sum_{n=0}^{\infty} |f_n| \mathcal{R}_f^n < \infty$ and $\sum_{n=0}^{\infty} f_n z^n \neq 1$ for all $|z| \leq \mathcal{R}_f$ if and only if $\mathcal{R}_f = \mathcal{R}_g$ and $\sum_{n=0}^{\infty} |g_n \mathcal{R}_g^n < \infty$.
- h) If there exists θ with $|\theta| < \mathcal{R}_f$ such that $\sum_{n=0}^{\infty} f_n \theta^n = 1$, then $\mathcal{R}_g < \mathcal{R}_f$ and $\sum_{n=0}^{\infty} |g_n \mathcal{R}_q^n = \infty$
- i) If $\sum_{n=0}^{\infty} |f_n| \mathcal{R}_f^n < \infty$, $\sum_{n=0}^{\infty} f_n z^n \neq 1$ for $|z| < \mathcal{R}_f$ and if $\sum_{n=0}^{\infty} f_n \theta^n = 1$ for some θ with $|\theta| = \mathcal{R}_f$, then $\mathcal{R}_g = \mathcal{R}_f$ and $\sum_{n=0}^{\infty} |g_n| \mathcal{R}_g^n = \infty$.

Proof: Recall that $\binom{\binom{x+n-1}{n}}{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type by Example 2.2.16c and Theorem 2.2.17. Thus existence and uniqueness of $(f_n)_{n \in \mathbb{N}}$ follows from Theorem 2.3.10.

- a) This follows from $q_n(x) = \sum_{k=0}^n f_n^{k*} \binom{x+k-1}{k}$ with x = 1.
- b) We have $g_n = (D q_n)(0) = \sum_{k=0}^n f_n^{k*} D^k \left(\binom{x+k-1}{k} \right) (0) = \sum_{k=1}^n f_n^{k*}$.

c) This follows from $\binom{k-2}{k} = 0$ for $k \ge 2(k \in \mathbb{N})$ and $\binom{k-2}{k} = -1$ for k = 1. d) This follows directly from b).

e) This follows from $\sum_{n=0}^{\infty} {\binom{x+n-1}{n}} z^n = \sum_{n=0}^{\infty} {\binom{-x}{n}} (-z)^n = (1-z)^{-x}$. f) First suppose $\mathcal{R}_f = 0$. If $\mathcal{R}_g > 0$, then c) and Theorem 4.4.10 imply that $\mathcal{R}_f > 0$. Hence, $\mathcal{R}_f = 0$ implies $\mathcal{R}_g = 0$. Now suppose $\mathcal{R}_f > 0$. Note that $f(z) = 1 - e^{-g(z)}$ for $|z| < \min\{\mathcal{R}_f, \mathcal{R}_g\}$. Hence, $\mathcal{R}_g \le \min\{|z| : |z| \le \mathcal{R}_f$ and $f(z) = 1\}$. If $f(z) \ne 1$ for $|z| \le r < \mathcal{R}_f$, then by Theorem 4.2.10 there exists a unique analytic function G on $|z| \leq r$ such that G(0) = 0 and $f(z) = 1 - e^{-G(z)}$. It follows from Lemma 4.2.1 with $\mathcal{K} = \{z \in \mathbb{C} : |z| \le r\}$ and a = 0 that G = g. Thus, $\mathcal{R}_g \ge r$. Since r was arbitrary, if follows that $\mathcal{R}_g \ge \min\{|z|: |z| \le \mathcal{R}_f \text{ and } f(z) = 1\}.$

g) ' \Rightarrow ' Part f) implies that $\mathcal{R}_f = \mathcal{R}_g$. It follows from theorem 3.2.2 with $\alpha_n = \mathcal{R}_f^n$ that $(\delta_{0n} - f_n)_{n \in \mathbb{N}} \in \exp \ell_1(\alpha)$. Since $f = 1 - e^{-g}$, lemma 3.2.1 implies that $(g_n)_{n \in \mathbb{N}} \in \ell_1(\alpha)$.

' \Leftarrow ' This follows from $f = 1 - e^{-g}$.

h) It follows from part f) that $\mathcal{R}_f < \mathcal{R}_g$ and that $\sum_{n=0}^{\infty} f_n \theta^n = 1$ for some θ with $|\theta| = \mathcal{R}_g$. Suppose $\sum_{n=0}^{\infty} |g_n| \mathcal{R}_g^n < \infty$. Then $\lim_{z \to \theta, |z| < \theta} f(z) = 0$ $1 - e^{-g(\theta)} \neq 1$, which contradicts $f(\theta) = 1$.

i) It follows from part f) that $\mathcal{R}_f = \mathcal{R}_g$. Suppose $\sum_{n=0}^{\infty} |g_n| \mathcal{R}_g^n < \infty$. Since $\sum_{n=0}^{\infty} |f_n| \mathcal{R}_f^n < \infty, \text{ we have } \lim_{z \to \theta, |z| < \theta} \sum_{n=0}^{\infty} f_n \theta^n = 1 - e^{-g(\theta)} \neq 1, \text{ which contradicts } \sum_{n=0}^{\infty} f_n \theta^n = 1.$

Remark 5.3.2 a) The converse of Theorem 5.3.1d is not true: consider e.g. $q_n(x) = \frac{x^n}{n!}$. Thus the set of sequences of polynomials of convolution type with $f_n \geq 0$ for all $n \in \mathbb{N}$ is a proper subset of the set of sequences of polynomials of convolution type with $g_n \geq 0$ for all $n \in \mathbb{N}$. This corresponds to the fact in probability theory that the class of compound geometric distributions is a proper subclass of the class of compound Poisson distributions.

b) From theorem 5.4.1 we see that if $f_n \ge 0$ for all $n \in \mathbb{N}$, then $(q_n(1))_{n \in \mathbb{N}}$ is an extended renewal sequence (see [Sta3, p. 185]).

c) If $f_n \ge 0$ for all $n \in \mathbb{N}$, then: $f(\mathcal{R}_f) < 1 \Rightarrow \mathcal{R}_g = \mathcal{R}_f$ and $g(\mathcal{R}_g) < \infty'$; $f(\mathcal{R}_f) = 1 \Rightarrow \mathcal{R}_g = \mathcal{R}_f$ and $g(\mathcal{R}_g) = \infty'$ and $f(\mathcal{R}_f) > 1$ or $\mathcal{R}_f = \infty \Rightarrow \mathcal{R}_g < \mathcal{R}_f$ and $g(\mathcal{R}_g) = \infty'$ (cf. Theorems 5.3.1g and 5.3.1h).

In [224] the usual renewal theory conditions are assumed. We now show that these conditions on $(f_n)_{n\in\mathbb{N}}$ imply that $\sum_{n=0}^{\infty} g_n z^n$ has a dominant logarithmic singularity on its circle of convergence. This explains why the centering and scaling constants of the central limit theorem in [224] do not depend on $(f_n)_{n\in\mathbb{N}}$ (cf. [224, p. 191, last paragraph]). An extension of the central limit theorem in [224] will be given in Section 5.4.

Theorem 5.3.3 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n \in \mathbb{N}}$ and let $(f_n)_{n \in \mathbb{N}}$ be the unique sequence of complex numbers such that $q_n(x) = \sum_{k=0}^n f_n^{k*} \binom{x+k-1}{k}$. Let \mathcal{R}_f , \mathcal{R}_g be the radius of convergence of $\sum_{n=0}^{\infty} f_n z^n$, $\sum_{n=0}^{\infty} g_n z^n$ respectively. Suppose that:

1.
$$0 < \mathcal{R}_g < \infty$$
 and $\lim_{r \uparrow \mathcal{R}_g} Re \ g(r) = +\infty$

2.
$$\sum_{n=0}^{\infty} n |q_n(-1)| \mathcal{R}_g^n < \infty$$

3.
$$\sum_{n=0}^{\infty} -nq_n(-1) \mathcal{R}_g^n \neq 0$$

4.
$$\sum_{n=0}^{\infty} -q_n(-1) z^n \neq 1 \text{ for } |z| = 1, z \neq 1.$$

Then there exists a sequence of polynomials $(r_n)_{n\in\mathbb{N}}$ of convolution type with coefficient sequence $(h_n)_{n\in\mathbb{N}}$ such that

a)
$$\mathcal{R}_{g}^{n}q_{n}(x) = \sum_{k=0}^{n} r_{k}(x) \binom{x+n-k-1}{n-k}$$

b) $\sum_{n=0}^{\infty} |r_{n}(x)| < \infty$ for all $x \in \mathbb{C}$
c) $\sum_{n=0}^{\infty} |h_{n}| < \infty$
d) $g(\mathcal{R}_{g}z) = -\mathcal{L}og(1-z) + h(z)$ for $|z| < 1$.

Moreover, $\lim_{n \to \infty} q_n(1) \mathcal{R}_g^n = \left(\sum_{n=0}^{\infty} -n q_n(-1) \mathcal{R}_g^n\right)^{-1}$ and $\sum_{n=0}^{\infty} |g_n \mathcal{R}_g^n - n^{-1}| < \infty$.

 $\mathbf{\alpha}$

Proof: First note that 2) implies $\sum_{n=0}^{\infty} |q_n(-1)| \mathcal{R}_g^n < \infty$. Hence,

$$\sum_{n=0}^{\infty} q_n(-1) \mathcal{R}_g^n = \lim_{r \uparrow \mathcal{R}_g} e^{-g(r)} = \lim_{r \uparrow \mathcal{R}_g} e^{-\operatorname{Re} g(r)} = 0.$$

Define the sequence $(\beta_k)_{k\in\mathbb{N}}$ by $\beta_k := \sum_{i=k+1}^{\infty} -q_i(-1) \mathcal{R}_g^i$. Thus $\beta_0 = 1$ and it follows from 2) that $(\beta_k)_{k\in\mathbb{N}} \in \ell_1$. We now show that $\sum_{k=0}^{\infty} \beta_k z^k \neq 0$ for $|z| \leq 1$. If $|z| \leq 1$ and $z \neq 1$, then

$$\sum_{k=0}^{\infty} \beta_k z^k = \sum_{k=0}^{\infty} z^k \sum_{i=k+1}^{\infty} -q_i(-1) \mathcal{R}_g^i =$$
$$\sum_{i=1}^{\infty} -q_i(-1) \mathcal{R}_g^i \sum_{k=0}^{i-1} z^k = \sum_{i=1}^{\infty} -q_i(-1) \mathcal{R}_g^i \frac{z^i - 1}{z - 1} =$$
$$\frac{1}{1 - z} \sum_{i=1}^{\infty} q_i(-1) \mathcal{R}_g^i (z^i - 1) = \frac{1}{1 - z} \left(1 - \sum_{i=0}^{\infty} q_i(-1) (\mathcal{R}_g z)^i \right),$$

since $\sum_{i=0}^{\infty} -q_i(-1)\mathcal{R}_g^i = 0$ and $q_0 = 1$. Thus 4) implies $\sum_{k=0}^{\infty} \beta_k z^k \neq 0$ for $|z| = 1, z \neq 1$. Moreover, since $\sum_{i=0}^{\infty} q_i(-1)(\mathcal{R}_g z)^i = e^{-g(\mathcal{R}_g z)}$ for |z| < 1, we have $\sum_{k=0}^{\infty} \beta_k z^k \neq 0$ for |z| < 1. Finally, we get from 3) that

$$\sum_{k=0}^{\infty} \beta_k = \sum_{k=0}^{\infty} \sum_{i=k+1}^{\infty} -q_i(-1) \mathcal{R}_g^i =$$
$$\sum_{i=1}^{\infty} \sum_{k=0}^{i-1} -q_i(-1) \mathcal{R}_g^i = \sum_{i=1}^{\infty} -q_i(-1) \mathcal{R}_g^i i = \left(1 - \sum_{i=0}^{\infty} -q_i(-1) \mathcal{R}_g^i i\right) \neq 0.$$

We conclude that $\sum_{k=0}^{\infty} \beta_k z^k \neq 0$ for $|z| \leq 1$. Since $\beta_0 = 1$, it follows from Theorem 4.2.2 with $\alpha_n = 1$ that there exists a sequence $(\gamma_n)_{n \in \mathbb{N}} \in \ell_1$ such that $\gamma_0 = 0$ and $\sum_{k=0}^{\infty} \beta_k z^k = \exp(\sum_{n=0}^{\infty} \gamma_n z^n)$ for $|z| \leq 1$. Define h_n by $h_n := -\gamma_n$ for all $n \in \mathbb{N}$ and write $h(z) = \sum_{n=0}^{\infty} h_n z^n$. Let $(r_n)_{n \in \mathbb{N}}$ be the sequence of polynomials of convolution type with coefficient sequence $(h_n)_{n \in \mathbb{N}}$, i.e. $r_n(x) = \sum_{k=0}^n h_n^{k*} \frac{x^k}{k!}$. It follows from $\sum_{k=0}^{\infty} \beta_k z^k = (1-z)^{-1} e^{-g(\mathcal{R}_g z)}$ that $e^{g(\mathcal{R}_g z)} = (1-z)^{-1} e^{h(z)}$. Since $g_0 = 0$, we have $g(\mathcal{R}_g z) = -\log(1-z) + h(z)$ for |z| < 1. Hence, $\mathcal{R}_g^n q_n(x) = \sum_{k=0}^n r_k(x) \binom{x+n-k-1}{n-k}$. This proves a) through d).

For the remaining statements, observe that $\mathcal{R}_g^n q_n(1) = \sum_{k=0}^n r_k(1)$. Since $\sum_{k=0}^{\infty} \beta_k z^k = \sum_{i=0}^{\infty} -q_i(-1) \mathcal{R}_g^i i$, it follows that

$$\lim_{n \to \infty} \mathcal{R}_g^n q_n(1) = \sum_{k=0}^{\infty} r_k(1) = e^{-\gamma(1)} = \left(\sum_{n=0}^{\infty} -n q_n(-1) \mathcal{R}_g^n\right)^{-1}.$$

Finally, it follows from $g(\mathcal{R}_g z) = -\log(1-z) + h(z)$ for |z| < 1 and $(h_n)_{n \in \mathbb{N}} \in \ell_1$ that $\sum_{n=0}^{\infty} |g_n \mathcal{R}_g^n - n^{-1}| < \infty$.

Remark 5.3.4 a) If $(f_n)_{n\in\mathbb{N}}$ is a probability distribution and g.c.d $\{n \ge 1 : f_n \ne 0\} = 1$, then $\sum_{n=0}^{\infty} f_n z^n \ne 1$ for $|z| = 1, z \ne 1$ (see [131, Theorem 3.6.1]). Thus if $(f_n)_{n\in\mathbb{N}}$ is a probability distribution, then condition 4) of Theorem 5.3.3 is implied by aperiodicity of $(f_n)_{n\in\mathbb{N}}$, since $f_n = -q_n(-1)$ for $n \ge 1$ by Theorem 5.3.1c (cf. [224]).

b) The statement $\lim_{n\to\infty} q_n(1) \mathcal{R}_g^n = \left(\sum_{n=0}^{\infty} -n q_n(-1) \mathcal{R}_g^n\right)^{-1}$ is an extension of the Discrete Renewal Theorem (cf. [93, Chapter 11]).

We now apply the previous theorems to obtain a theorem similar to [224, Theorem 4] (recall that $f_n = -q_n(-1)$ for $n \ge 1$ by Theorem 5.3.1c).

Theorem 5.3.5 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n \in \mathbb{N}}$. Suppose that:

- 1. the power series $g(z) = \sum_{n=1}^{\infty} g_n z^n$ has a positive, finite radius of convergence \mathcal{R}_g and $\lim_{r \uparrow \mathcal{R}_g} \operatorname{Re} g(r) = +\infty$
- 2. $\sum_{n=0}^{\infty} n |q_n(-1)| \mathcal{R}_g^n < \infty$ 3. $\sum_{n=0}^{\infty} -n q_n(-1) \mathcal{R}_g^n \neq 0$ 4. $\sum_{n=0}^{\infty} -q_n(-1) z^n \neq 1 \text{ for } |z| = 1, z \neq 1.$

Then $\lim_{n\to\infty} q_n(x)/q_n(1) = 0$ for |x| < 1.

Proof: It follows from Theorem 5.3.3 that $\mathcal{R}_g^n q_n(x) = \sum_{k=0}^n r_k(x) \binom{x+n-k-1}{n-k}$ with $(r_n(x))_{n\in\mathbb{N}} \in \text{inv } \ell_1$ for all $x \in \mathbb{C}$. If |x| < 1, then $\lim_{n\to\infty} \binom{x+n-1}{n} = 0$ by Lemma 5.1.3. It follows from dominated convergence that

$$\lim_{n\to\infty}\sum_{k=0}^n r_k(x) \begin{pmatrix} x+n-k-1\\ n-k \end{pmatrix} = 0.$$

Moreover, $\lim_{n\to\infty} \mathcal{R}_g^n q_n(1) = \sum_{n=0}^{\infty} r_n(1) \neq 0$ since $(r_n(1))_{n\in\mathbb{N}} \in \text{inv } \ell_1$. Hence, $\lim_{n\to\infty} q_n(x)/q_n(1) = 0$ for |x| < 1.

5.4 Logarithmic singularities

In this section we prove a central limit theorem for random variables $Y_n^{(\lambda)}$ with probability generating function $q_n(\lambda x)/q_n(\lambda)$, where $(q_n)_{n\in\mathbb{N}}$ is a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$ such that $\sum_{n=0}^{\infty} g_n z^n$ has a dominant logarithmic singularity on its circle of convergence. We derive our central limit theorem from the asymptotic behaviour of q_n . Similar results have been obtained by Flajolet and Soria (see [96, 97]). Contrary to Flajolet and Soria, we do not use contour integration. Instead, our method relies on simple estimates and a Banach algebra theorem from Chapter 4. Consequently, our conditions on g are different (probably incomparable) from those in [96, 97]. Note that the function R in [96, Definition on p. 169] should satisfy $R(z) = K + o\left(\left(\log(1-z/\rho)\right)^{-1}\right)$ instead of R(z) = K + o(1) (see [97, p. 11]).

We start with determining the asymptotic behaviour of $q_n(x)/q_n(1)$. This asymptotic behaviour will be used in Theorem 5.4.3 to obtain a central limit theorem.

Theorem 5.4.1 Let $(q_n)_{n\in\mathbb{N}}$ be a sequence of polynomials of convolution type with coefficient sequence $(g_n)_{n\in\mathbb{N}}$. Suppose that the \mathcal{R}_g , the radius of convergence of $g(z) = \sum_{n=0}^{\infty} g_n z^n$ is positive and finite. Define $(h_n)_{n\in\mathbb{N}}$ by $h_n :=$ $\mathcal{R}_g^n g_n - n^{-1}$ and let $(r_n)_{n\in\mathbb{N}}$ be the sequence of polynomials of convolution type with coefficient sequence $(h_n)_{n\in\mathbb{N}}$.

$$\begin{split} &If\sum_{n=0}^{\infty}|r_n(x)|<\infty \text{ for all }x>0, \text{ then }\mathcal{R}_g^n\,q_n(x)\sim \frac{n^{x-1}}{\Gamma(x)}\sum_{k=0}^{\infty}r_k(x) \text{ as }n\to\infty\\ &for \text{ fixed }x\geq 1. \quad \text{If moreover }r_n(x)=O(n^{-1}) \text{ as }n\to\infty \text{ for a fixed }x \text{ with}\\ &0< x<1, \text{ then }\mathcal{R}_g^n\,q_n(x)\sim \frac{n^{x-1}}{\Gamma(x)}\sum_{k=0}^{\infty}r_k(x) \text{ as }n\to\infty. \end{split}$$

Proof: The definition of h_n implies that $\mathcal{R}_g^n q_n(x) = \sum_{k=0}^n r_k(x) \binom{x+n-k-1}{n-k}$ for all $n \in \mathbb{N}$.

I. $x \ge 1$

By Lemma 5.1.3, $\lim_{n\to\infty} \binom{x+n-k-1}{n-k} n^{1-x} = 1/\Gamma(x)$ for fixed k with $0 \le k \le n$. Since $x \ge 1$, $\binom{x+n-k-1}{n-k} n^{1-x} \le \binom{x+n-1}{n} n^{1-x} \le C_1$ by Lemmas 5.1.1a and 5.1.3. Since $\sum_{n=0}^{\infty} |r_n(x)| < \infty$, dominated convergence yields $\mathcal{R}_g^n q_n(x) \sim n^{x-1}/\Gamma(x) \sum_{k=0}^{\infty} r_k(x)$ as $n \to \infty$.

II. 0 < x < 1

We first evaluate $\lim_{n\to\infty} n^{1-x} \sum_{k=0}^{\lfloor n/2 \rfloor} {\binom{x+n-k-1}{n-k}}$. If $k \leq \lfloor n/2 \rfloor$, then $n^{1-x} \leq 2^{1-x} (n-k)^{1-x}$ and $n^{1-x} {\binom{x+n-k-1}{n-k}}$ is uniformly bounded by Lemma 5.1.3. Hence, $\lim_{n\to\infty} \sum_{k=0}^{\lfloor n/2 \rfloor} r_k(x) n^{1-x} {\binom{x+n-k-1}{n-k}} = \sum_{k=0}^{\infty} r_k(x)/\Gamma(x)$ by dominated convergence. We are done if we prove that

$$\lim_{n \to \infty} n^{1-x} \sum_{k=[n/2]+1}^{n} r_k(x) \begin{pmatrix} x+n-k-1 \\ n-k \end{pmatrix} = 0.$$

Fix an α with $1 < \alpha < (1-x)^{-1}$. Applying Lemma 5.1.4 and Hölder's inequality, we obtain

$$\begin{split} n^{1-x} \sum_{k=[n/2]+1}^{n} \left| r_k(x) \begin{pmatrix} x+n-k-1\\n-k \end{pmatrix} \right| &\leq \\ n^{1-x} \left(\sum_{k=[n/2]+1}^{n} |r_k(x)|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{[n/2]} \begin{pmatrix} x+j-1\\j \end{pmatrix}^{\alpha} \right)^{1/\alpha} &\leq \\ C_1 n^{1-x} \left(\sum_{k=[n/2]+1}^{n} |r_k(x)|^{1/(\alpha-1)} |r_k(x)| \right)^{1-1/\alpha} [n/2]^{x-1+1/\alpha} &\leq \\ C_2 \left(\max_{k\ge [n/2]} |r_k(x)|^{1/\alpha} \right) \left(\sum_{k=[n/2]+1}^{n} |r_k(x)| \right)^{1-1/\alpha} n^{1/\alpha} &\leq \\ C_3 \left(\sum_{k=[n/2]+1}^{n} |r_k(x)| \right)^{1-1/\alpha} &= o(1) \end{split}$$

as $n \to \infty$.

If in Theorem 5.4.1, $(r_n)_{n \in \mathbb{N}}$ satisfies additional conditions as positivity or monotonicity, then Theorem 5.4.1 can be obtained from Tauber theorems (cf. [93, Chapter 8.5]).

The next theorem is a central limit theorem for a sequence of random variables $(Y_n^{(\lambda)})_{n \in \mathbb{N}}$, where $Y_n^{(\lambda)}$ has probability generating function $q_n(\lambda x)/q_n(\lambda)$. For an interpretation of $Y_n^{(\lambda)}$ in terms of a compound Poisson process, see [222]. For examples of combinatorial interpretations of $Y_n^{(1)}$, see Examples 5.4.5. We first need a lemma.

Lemma 5.4.2 If $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type such that $\sum_{n=0}^{\infty} |q_n(x)| < \infty$ for all x > 0, then $\lim_{t \to s} |q_n(t) - q_n(s)| = 0$ for all s > 0.

Proof: Consider the separable Banach algebra ℓ_1 with convolution as multiplication. Define $f: (0, \infty) \to \ell_1$ by $f(t) := (q_n(t))_{n \in \mathbb{N}}$. Since the polynomials q_n are of convolution type, we have f(u+v) = f(u) * f(v) for all u, v > 0. If $y = (y_n)_{n \in \mathbb{N}} \in (\ell_1)^* = \ell_\infty$, then $\langle ft \rangle, y \rangle = \sum_{n=0}^{\infty} q_n(t) y_n$. Thus, f is weakly measurable in ℓ_1 according to [118, Definition 3.5.4] and strongly measurable by [118, Corollary 2, p. 73]. The theorem now follows from [118, Theorem 9.3.1].

Theorem 5.4.3 Let $(q_n)_{n \in \mathbb{N}}$ be a sequence of polynomials of convolution type with non-negative coefficients and with coefficient sequence $(g_n)_{n \in \mathbb{N}}$. Define

 $(h_n)_{n\in\mathbb{N}}$ by $h_n := \mathcal{R}_g^n g_n - n^{-1}$ and let $(r_n)_{n\in\mathbb{N}}$ be the sequence of polynomials of convolution type with coefficient sequence $(h_n)_{n\in\mathbb{N}}$. Suppose that \mathcal{R}_g , the radius of convergence of $g(z) = \sum_{n=0}^{\infty} g_n z^n$, is finite and positive and suppose that $\sum_{n=0}^{\infty} |r_n(x)| < \infty$ for all x > 0. Let the random variable $Y_n^{(\lambda)}$ have probability generating function $q_n(\lambda x)/q_n(\lambda)$ for each $n \in \mathbb{N}$. Then the distribution of $(Y_n^{(\lambda)} - \lambda \log n) (\lambda \log n)^{-1/2}$ converges to the standard normal law for all $\lambda \geq 1$. If $0 < \lambda < 1$ and $r_n(x) = O(n^{-1})$ uniformly in a real neighbourhood of λ as $n \to \infty$, then the distribution of $(Y_n^{(\lambda)} - \lambda \log n) (\lambda \log n)^{-1/2}$ converges to the standard normal law.

Proof: Write $a_n := (\lambda \log n)^{1/2}$. Because $Y_n^{(\lambda)}$ has probability generating function $q_n(\lambda x)/q_n(\lambda)$, $(Y_n^{(\lambda)} - \lambda \log n) (\lambda \log n)^{-1/2}$ has moment generating function $q_n\left(\lambda e^{z/a_n}\right) e^{-a_n z}/q_n(\lambda)$. By Lemma 5.1.5, it suffices to prove $\lim_{n\to\infty} q_n\left(\lambda e^{z/a_n}\right) e^{-a_n z}/q_n(\lambda) = e^{\frac{1}{2}z^2}$ for all z > 0.

Recall that in both cases $(\lambda \ge 1 \text{ and } 0 < \lambda < 1)$ we have

$$\lim_{n \to \infty} n^{1-\lambda} \mathcal{R}_g^n q_n(\lambda) = \sum_{k=0}^{\infty} r_k(\lambda)$$

by Theorem 5.4.1. Thus it suffices to prove

$$\lim_{n\to\infty} n^{1-\lambda} \mathcal{R}_g^n q_n \left(\lambda e^{z/a_n}\right) e^{-a_n z} = e^{\frac{1}{2}z^2} \frac{1}{\Gamma(\lambda)} \sum_{k=0}^{\infty} r_k(\lambda).$$

Since $\mathcal{R}_{g}^{n} q_{n}(x) = \sum_{k=0}^{n} r_{k}(x) \binom{x+n-k-1}{n-k}$, we may write

$$n^{1-\lambda} \mathcal{R}_g^n q_n \left(\lambda e^{z/a_n} \right) e^{-a_n z} = n^{1-\lambda} \sum_{k=0}^n r_k (\lambda e^{z/a_n}) \begin{pmatrix} \lambda + n - k - 1 \\ n - k \end{pmatrix} \varphi_{nk}(z),$$

where

$$\varphi_{nk}(z) := e^{-a_n z} \left(\frac{\lambda e^{z/a_n} + n - k - 1}{n-k} \right) \left(\frac{\lambda + n - k - 1}{n-k} \right)^{-1}.$$

I. $\lambda \geq 1$

We now write

$$n^{1-\lambda} \sum_{k=0}^{n} r_k \left(\lambda e^{z/a_n} \right) \begin{pmatrix} \lambda + n - k - 1 \\ n - k \end{pmatrix} \varphi_{nk}(z) = T_1(n) + T_2(n),$$

where

$$T_1(n) := n^{1-\lambda} \sum_{k=0}^n \left\{ r_k \left(\lambda e^{z/a_n} \right) - r_k(\lambda) \right\} \, \binom{\lambda + n - k - 1}{n - k} \varphi_{nk}(z)$$

 and

$$T_2(n) := n^{1-\lambda} \sum_{k=0}^n r_k(\lambda) \begin{pmatrix} \lambda + n - k - 1 \\ n - k \end{pmatrix} \varphi_{nk}(z).$$

By Lemma 5.1.2, we have $\lim_{n\to\infty} \varphi_{nk}(z) = e^{\frac{1}{2}z^2}$ for fixed k with $0 \le k \le n$. For $\lambda \ge 1$ and $0 \le k \le n$, $n^{1-\lambda} \le (n-k)^{1-\lambda}$, thus $n^{1-\lambda} \binom{\lambda+n-k-1}{n-k}$ is uniformly bounded in n and k with $0 \le k \le n$ by Lemma 5.1.3. Applying Lemma 5.1.2 and the dominated convergence theorem, we obtain $\lim_{n\to\infty} T_2(n) = e^{\frac{1}{2}z^2} \frac{1}{\Gamma(\lambda)} \sum_{k=0}^{\infty} r_k(\lambda)$. Since $\varphi_{nk}(z)$ is uniformly bounded by Lemma 5.1.2, we have

$$\lim_{n \to \infty} |T_1(n)| \le C \lim_{n \to \infty} \sum_{k=0}^{\infty} \left| r_k \left(\lambda e^{z/a_n} \right) - r_k(\lambda) \right| = 0$$

by Theorem 5.4.2.

II. $0 < \lambda < 1$

We first evaluate $\lim_{n\to\infty} n^{1-\lambda} \sum_{k=0}^{\lfloor n/2 \rfloor} r_k(\lambda e^{z/a_n}) {\binom{\lambda+n-k-1}{n-k}} \varphi_{nk}(z)$. If $k \leq \lfloor n/2 \rfloor$, then $n^{1-\lambda} \leq 2^{1-\lambda} (n-k)^{1-\lambda}$, thus $n^{1-\lambda} {\binom{\lambda+n-k-1}{n-k}}$ is uniformly bounded by Lemma 5.1.3. By Lemma 5.1.2, $\varphi_{nk}(z)$ is uniformly bounded. Hence,

$$\lim_{n \to \infty} \sum_{k=0}^{\lfloor n/2 \rfloor} \left\{ r_k (\lambda e^{z/a_n}) - r_k(\lambda) \right\} n^{1-\lambda} \binom{\lambda+n-k-1}{n-k} \varphi_{nk}(z) \le \lim_{n \to \infty} \sum_{k=0}^{\lfloor n/2 \rfloor} \left| r_k \left(\lambda e^{z/a_n} \right) - r_k(\lambda) \right| = 0$$

by Theorem 5.4.2 and

$$\lim_{n \to \infty} \sum_{k=0}^{\lfloor n/2 \rfloor} r_k(\lambda) n^{1-\lambda} \binom{\lambda+n-k-1}{n-k} \varphi_{nk}(z) = \sum_{k=0}^{\infty} r_k(\lambda) \frac{1}{\Gamma(\lambda)} e^{\frac{1}{2} z^2}$$

by dominated convergence. We are done if we prove that

$$\lim_{n \to \infty} n^{1-\lambda} \sum_{k=\lfloor n/2 \rfloor+1}^{n} r_k \left(\lambda e^{z/a_n} \right) \, \binom{\lambda+n-k-1}{n-k} \varphi_{nk}(z) = 0.$$

Fix an α with $1 < \alpha < (1 - \lambda)^{-1}$. Applying Lemma 5.1.4 and Hölder's inequal-

ity, we obtain

$$n^{1-\lambda} \sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_k \left(\lambda e^{z/a_n} \right) \begin{pmatrix} \lambda+n-k-1\\ n-k \end{pmatrix} \right| \leq n^{1-\lambda} \left(\sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor} \begin{pmatrix} \lambda+j-1\\ j \end{pmatrix}^{\alpha} \right)^{1/\alpha} \leq n^{1-\lambda} \left(\sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(\sum_{j=0}^{\lfloor n/2 \rfloor+1} \left| r_k \left(\lambda e^{z/a_n} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left(r_k \left(\lambda e^{z/a_n} \right) \right)^{1/\alpha} \right)^{1/\alpha} \right)^{1/\alpha}$$

$$C_{1} n^{1-\lambda} \left(\sum_{k=[n/2]+1}^{n} \left| r_{k} \left(\lambda e^{z/a_{n}} \right) \right|^{1+1/(\alpha-1)} \right)^{1-1/\alpha} \left[\frac{n}{2} \right]^{\lambda-1+1/\alpha} \leq \frac{1-1}{\alpha}$$

$$C_{2} \max_{k \ge \lfloor n/2 \rfloor} \left| r_{k} \left(\lambda e^{z/a_{n}} \right) \right|^{1/\alpha} \left(\sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_{k} \left(\lambda e^{z/a_{n}} \right) \right| \right)^{1-1/\alpha} n^{1/\alpha} \le C_{3} \left(\sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_{k} \left(\lambda e^{z/a_{n}} \right) \right| \right)^{1-1/\alpha} .$$

Using Theorem 5.4.2, we see that

$$\lim_{n \to \infty} \sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_k \left(\lambda e^{z/a_n} \right) \right| \leq \\ \lim_{n \to \infty} \sum_{k=\lfloor n/2 \rfloor+1}^{n} \left| r_k \left(\lambda e^{z/a_n} \right) - r_k(\lambda) \right| + \lim_{n \to \infty} \sum_{k=\lfloor n/2 \rfloor+1}^{n} |r_k(\lambda)| = 0.$$

Remark 5.4.4 If in Theorem 5.4.1 or 5.4.3 we have
$$(h_n)_{n\in\mathbb{N}} \in \ell_1$$
 and $h_n = O(n^{-1})$, then $r_n(x) = O(n^{-1})$ uniformly in an interval around λ ($\lambda > 0$) as $n \to \infty$ as the following proof shows. Consider the algebra O of all sequences $a \in \ell_1$ such that $|a_n| = O(n^{-1})$ with componentwise addition and convolution as multiplication. Equipped with norm $||a|| := ||a||_1 + u n|a_n|$, this algebra becomes a Banach algebra. Hence, $r_n(x) = O(n^{-1})$ for all $x \in \mathbb{C}$. We now set out to prove the uniform $O(n^{-1})$ property. Define $(b_n)_{n\in\mathbb{N}}$ by $b_n := |h_n|$ and let $(v_n)_{n\in\mathbb{N}}$ be the unique sequence of polynomials of convolution type with coefficient sequence $(b_n)_{n\in\mathbb{N}}$. Hence, if $\mu > \lambda$ and $0 < x \leq \mu$, then $r_n(x) \leq v_n(x) \leq v_n(\mu) \leq C\mu n^{-1}$.

Examples 5.4.5 a) It follows from Remark 5.4.4 that the following sequences of polynomials satisfy the conditions of Theorems 5.4.1 and 5.4.3 $((g_n)_{n \in \mathbb{N}})$ is the coefficient sequence of $(q_n)_{n \in \mathbb{N}}$ and $g(z) = \sum_{n=0}^{\infty} g_n z^n$:

1. the derangement polynomials with $g(z) = -\log(1-z) + z$ (this solves the open problem of [47, p. 20]). These polynomials count the number of cycles in derangements, i.e. permutations without cycles of length one. Thus $P(Y_n^{(1)} = k)$ is the probability that a random derangement of $\{1, \ldots, n\}$ has k cycles.

2. the polynomials with $g(z) = -\log(1-z) + z + \frac{1}{2}z^2$. These polynomials count the number of connected components in 2-regular graphs. Thus $P(Y_n^{(1)} = k)$ is the probability that a random 2-regular graph with n points has k components (cf. [96, pp. 174-175]).

The other examples given in [96, pp. 173-175] also satisfy the conditions of Theorems 5.4.1 and 5.4.3.

b) Consider the Mittag-Leffler polynomials of Example 3.1.3b with $g(z) = -\log(1-z) + \log(1+z)$ (see also [202, p. 75]). Here $P(Y_n^{(\frac{1}{2})} = k)$ is the probability that a random permutation of $\{1, \ldots, n\}$ without cycles of even length has k cycles. Note that the polynomials of convolution type associated to $\log(1+z)$ are the polynomials $\binom{x}{n}$. It follows from Raabe's convergence test that $\sum_{n=0}^{\infty} |\binom{x}{n}| < \infty$ for x > 0. In the terminology of Theorem 5.4.3, the Mittag-Leffler polynomials are an example of a sequence of polynomials such that $\sum_{n=0}^{\infty} |r_n(x)| < \infty$ for all x > 0 and $\sum_{n=0}^{\infty} |h_n| = \infty$ (cf. Remark 5.4.4). The uniform $O(n^{-1})$ condition necessary for Theorem 5.4.3 follows from $n\binom{x}{n} = x \left| \frac{(x-1) \dots (x-(n-1))}{1 \dots n^{-1}} \right| \le x$ for $0 \le x < 1$.

c) Consider the polynomials q_n with $g(z) = z - \log(1 - z^2)$. We will show that the asymptotic behaviour of q_n is different for even n and odd n. We have

$$q_{2n}(x) = \sum_{k=0}^{n} {\binom{-x}{k}} (-1)^k \frac{x^{2n-2k}}{(2n-2k)!}$$

and

$$q_{2n+1}(x) = \sum_{k=0}^{n} {\binom{-x}{k}} (-1)^k \frac{x^{2n+1-2k}}{(2n+1-2k)!}.$$

Since $\sum_{n=0}^{\infty} \frac{x^{2n}}{(2n)!} = \frac{1}{2} (e^x + e^{-x})$ and $\sum_{n=0}^{\infty} \frac{x^{2n+1}}{(2n+1)!} = \frac{1}{2} (e^x - e^{-x})$, [86, Lemma 2.2.] and Lemma 5.1.3 yield $q_{2n}(x) \sim \frac{n^{x-1}}{\Gamma(x)} \frac{1}{2} (e^x + e^{-x})$ and $q_{2n+1}(x) \sim \frac{1}{2} (e^x - e^{-x})$. In spite of the different asymptotic behaviour, there exists a central limit theorem (same proof as Theorem 5.3.2).

5.5 Infinitely divisible probability measures on \mathbb{N}

In this section we show that using the Banach algebra theory developed in Chapter 4 and Section 5.2, it is possible to give a more transparent proof of the main result of [87]. The proofs in [87] use Banach algebra results from [61].

Definition 5.5.1 A probability generating function P is said to be infinitely divisible if for all $k \ge 1$ there exists a probability generating function P_k such that $P = (P_k)^k$.

For more information on infinitely divisible probability measures, see [93, Chapter 17] and [230, 233].

We now consider infinitely divisible probability measures on N. It follows from the Lévy-Hinčin representation (see [93, Chapter 17, Section 2] or [151, Theorem 5.5.1]; for a proof using Choquet theory see [127]) that if μ is a probability measure on N with infinitely divisible probability generating function, then there exists a measure ν on N (the **Lévy-measure**) such that

$$\sum_{n=0}^{\infty} \mu_n \, z^n = \exp\left\{-\lambda + \sum_{k=1}^{\infty} \nu_k \, z^k\right\}$$
(5.1)

for all $|z| \leq 1$, $\mu_n := \mu\{n\}$, $\nu_n := \nu\{n\}$ and $\lambda := \sum_{k=1}^{\infty} \nu_k$.

As an illustration of the Banach algebra theory of Chapter 4, we now prove the Lévy-Hinčin theorem for infinitely divisible probability generating functions. For a simple real analysis proof of this theorem, see [92, Section 12.2].

Lemma 5.5.2 Let $P(z) = \sum_{n=0}^{\infty} p_n z^n$ be an infinitely divisible probability generating function. Then $\sum_{n=0}^{\infty} p_n z^n \neq 0$ for $|z| \leq 1$.

Proof: It follows from [151, th. 5.3.1] that $\sum_{n=0}^{\infty} p_n z^n \neq 0$ for |z| = 1. Since P is infinitely divisible, Lemma 4.3.2 yields that ind P = 0. Hence, the Argument Principle ([45, Corollary 5.86]) yields that $\sum_{n=0}^{\infty} p_n z^n \neq 0$ for $|z| \leq 1$. \Box

For another proof of Lemma 5.5.2, combine [151, th. 5.3.1] and [151, th. 8.4.1] (cf. [232, p. 5]).

Theorem 5.5.3 Let $P(z) = \sum_{n=0}^{\infty} p_n z^n$ be an infinitely divisible probability generating function. Then there exists a sequence $(\nu_n)_{n \in \mathbb{N}}$ in \mathbb{R} such that $\nu_n \ge 0$ for all $n \ge 1$ and such that $\sum_{n=0}^{\infty} \nu_n < \infty$ and $P(z) = \exp \{\sum_{n=0}^{\infty} \nu_n z^n\}$ for $|z| \le 1$.

Proof: It follows from Lemma 5.5.2 that $\sum_{n=0}^{\infty} p_n z^n \neq 0$ for all $|z| \leq 1$. It follows from Theorem 4.2.2 with $\alpha_n = 1$ that there exists a sequence $(\nu_n)_{n \in \mathbb{N}} \in \ell_1$ such that $P(z) = \exp \{\sum_{n=0}^{\infty} \nu_n z^n\}$. We now set out to prove that $\nu_n \geq 0$ for $n \geq 1$. Let $(q_n)_{n \in \mathbb{N}}$ be the sequence of polynomials of convolution type with coefficient sequence $(0, \nu_1, \nu_2, \ldots)$. Thus, $q_n(1) = e^{-\nu_0} p_n$ for all $n \in \mathbb{N}$. Since P is infinitely divisible, there exists for each integer $k \geq 2$ a sequence $(a_n)_{n \in \mathbb{N}}$ of non-negative numbers such that $a_n^{k*} = p_n$. It easily follows by induction on k that each $(a_n)_{n \in \mathbb{N}}$ is unique except for a_0 . Thus infinite divisibility of P implies that $q_n(1/k) \geq 0$ for all $k, n \in \mathbb{N}$. Since $(q_n)_{n \in \mathbb{N}}$ is a sequence of polynomials of convolution type, Theorem 2.1.8 or Remark 2.1.10 fimplies that $\nu_n \geq 0$ for all $n \in \mathbb{N}$.

We now give a proof of the Embrechts-Hawkes result on tails of infinitely divisible probability measures on N.

Theorem 5.5.4 ([87, Theorem 1]) Let p be a probability measure on \mathbb{N} with infinitely divisible probability generating function and Lévy measure ν . Define $\alpha_0 := -1$ and $\alpha_k := \nu_k / \lambda$ $(k \ge 1)$, where $\lambda := \sum_{k=1}^{\infty} \nu_k$. Suppose that $p_1 \ne 0$ and $\alpha_n \ne 0$ for n large enough. Then the following are equivalent:

- (i) $\alpha_n^{2*} \sim 2 \alpha_n$ and $\alpha_{n+1} \sim \alpha_n \ (n \to \infty)$
- (ii) $p_n^{2*} \sim 2 p_n$ and $p_{n+1} \sim p_n \ (n \to \infty)$
- (iii) $p_n \sim \lambda \alpha_n$ and $\alpha_{n+1} \sim \alpha_n \ (n \to \infty)$.

Proof: Note that by the Lévy-Hinčin representation (5.1) and the choice $\alpha_0 =$ $-\lambda/\lambda = -1$, we have $p = e^{\lambda \alpha}$, where $p = (p_n)_{n \in \mathbb{N}}$ and $\alpha = (\alpha_n)_{n \in \mathbb{N}}$.

 $(ii) \Rightarrow (i)$ We use the Banach algebra \mathcal{U}_L of Definition 5.2.1 with $\mu_n = p_n$. First note that $p_n \neq 0$ for all $n \in \mathbb{N}$ by applying [231, Corollary on p. 813] or by using Lemma 2.1.5b to show that for $n \ge 1$ we have $e^{\lambda} p_n = \sum_{k=1}^n \alpha_n^{k*} \ge 1$ $\alpha_1^{n*} = (\alpha_1)^n = e^{n\lambda} (p_1)^n / n! > 0.$

 $\alpha_1^{n} = (\alpha_1)^n = e^{\alpha_1} (p_1)^n / n! > 0.$ It follows from Lemma 5.5.2 that $\sum_{n=0}^{\infty} p_n z^n \neq 0$ for all $|z| \leq 1$. Thus $(p_n)_{n \in \mathbb{N}} \in \exp \mathcal{U}_L$ by Theorem 5.2.2e. Hence, $(\alpha_n)_{n \in \mathbb{N}} \in \mathcal{U}_L$. In particular, $\lim_{n\to\infty} \alpha_n/p_n = L$ exists. We now show that $L \neq 0$. If L = 0, then the fact that \mathcal{U}_0 is a Banach algebra implies that $\lim_{n\to\infty} \alpha_n^{k*}/p_n = 0$ for all $k \in \mathbb{N}$. By continuity, $\lim_{n\to\infty} p_n/p_n = 0$ because $e^{\lambda \alpha} = p$. Since this is absurd, we conclude that $L \neq 0$. Thus,

$$\lim_{n \to \infty} \frac{\alpha_{n+1}}{\alpha_n} = \lim_{n \to \infty} \frac{\alpha_{n+1}}{\alpha_n} \lim_{n \to \infty} \frac{p_{n+1}}{p_n} \lim_{n \to \infty} \frac{p_n}{\alpha_n} = 1.$$

It remains to prove that $\alpha_n^{2*} \sim 2\alpha_n \ (n \to \infty)$. This follows from [86, Theorem 2.9iv].

 $(i) \Rightarrow (iii)$ Using mathematical induction on k, it follows that $\alpha_n^{k*} \sim k \alpha_n$ $(n \to \infty)$ (see [61, Lemma 5]). It follows from [87, Lemma 2] or [86, Theorem 2.9iii] that for each c > 1 there exists a positive constant A such that $\alpha_n^{k*} \leq A c^k \alpha_n$ for all $k, n \in \mathbb{N}$. Thus we may apply the dominated convergence theorem to $p_n/\alpha_n = e^{-\lambda} \sum_{k=1}^{\infty} \frac{\alpha_n^{k*}}{\alpha_n} \frac{\lambda^k}{k!}$ (this equality follows from (5.1)). We conclude that $p_n \sim \lambda \alpha_n \ (n \to \infty)$.

 $(iii) \Rightarrow (ii)$ Use the real analysis proof of [87, Theorem 1].

Remark 5.5.5 a) The proof of $(i) \Rightarrow (iii)$ in [87] contains some misprints, especially Formula (12).

b) For a version of Theorem 5.5.4 on |z| < r, see [87, Theorem 2].

c) The Banach algebra method works well for probability measures on N, since the maximal ideal space of the Banach algebra involved has a simple structure (see Theorem 5.2.2c). It is possible to derive analogues of Theorem 5.5.4 for probability measures on \mathbb{Z} (cf. [61, pp. 267-268]).

The Banach algebra of all complex Borel measures on \mathbb{R} is much more complicated (see however [95]).

Bibliography

- J Aczél. Lectures on Functional Equations and Their Applications. Academic Press, New York, 1966.
- J. Aczél. Functions of binomial type mapping groupoids into rings. Math. Z., 154:115-124, 1977. (MR 55#13115).
- [3] J. Aczél and J. Dhombres. Functional Equations in Several Variables, volume 31 of Encyclopedia of mathematics and its applications. Cambridge University Press, 1989.
- [4] J. Aczél and G. Vranceanu. Equations fonctionelles liées aux groupes linéaires commutatifs. Coll. Math., 26:371–383, 1972. (MR 49#11084).
- [5] W. Al-Salam. Characterization theorems for orthogonal polynomials. In P. Nevai, editor, Orthogonal polynomials: Theory and practice, pages 1– 24. Kluwer Academic Publishers, Dordrecht, Netherlands, 1990.
- [6] W.A. Al-Salam. On a characterization of Meixner's polynomials. Quart. J. Math. Oxford, 17, 1966. (MR 32#7804).
- [7] W.A. Al-Salam and A. Verma. Generalized Sheffer polynomials. Duke Math. J., 37:361-365, 1970. (MR 41#7175).
- [8] W.R. Allaway. Extensions of Sheffer polynomial sets. SIAM J. Math. Anal., 10:38-48, 1979. (MR 80e:33008).
- [9] G.E. Andrews. On the foundations of combinatorial theory V, Eulerian differential operators. Stud. App. Math., 50:345-375, 1971. (MR 46#8845).
- [10] P. Appell. Sur une classe de polynômes. Ann. Sci. Ecole Norm. Sup, (2) 9:119-144, 1880.
- [11] R. Arens. The group of invertible elements of a commutative Banach algebra. Studia Math. (Seria specjalna), 1, 1963. (MR 26#4198).
- [12] F. Avram and M.S. Taqqu. Noncentral limit theorems and appell polynomials. Ann. Probab., 15:767-775, 1987. (MR 88i:60058.

- [13] A.K. Avramjonok. The theory of operators (n-dimensional case) in combinatorial analysis (Russian). In Combinatorial analysis and asymptotic analysis no. 2, pages 103-113. Krasnojarsk Gos. Univ., Krasnojarsk, 1977. (MR 80c:05017).
- [14] S.K. Bar-Lev, D. Bshouty, P. Enis, G. Letac, I.-L. Lu, and D. Richards. The diagonal multivariate natural exponential families and their classification. J. Theoret. Probab., 7:883-929, 1994. (MR 96b:60030).
- [15] M. Barnabei. Lagrange inversion in infinitely many variables. J. Math. Anal. Appl., 108:198-210, 1985. (MR 86j:05023).
- [16] M. Barnabei, A. Brini, and G. Nicoletti. Polynomial sequences of integral type. J. Math. Anal. Appl., 78:598–617, 1980. (MR 82c:05016).
- [17] M. Barnabei, A. Brini, and G. Nicoletti. Recursive matrices and umbral calculus. J. Algebra, 75:546–573, 1982. (MR 84i:05020).
- [18] M. Barnabei, A. Brini, and G. Nicoletti. A general umbral calculus in infinitely many variables. Adv. Math., 50:49-93, 1983. (MR 85g:05025).
- [19] M. Barnabei, A. Brini, and G. Nicoletti. A general umbral calculus. Adv. Math., Suppl. Stud, 10:221-244, 1986. (Zbl. 612.05009).
- [20] P.D. Barry and D.J. Hurley. Generating functions for relatives of classical polynomials. Proc. Amer. Math. Soc., 103:839-846, 1988. (MR 89f:33025).
- [21] G. Baxter. Polynomials defined by a difference system. J. Math. Anal. Appl., 2:223-263, 1961. (MR 23#A3421).
- [22] B. Beauzamy. Introduction to Banach Spaces and Their Geometry. North Holland, 2nd. edition, 1985.
- [23] E.T. Bell. The history of Blissard's symbolic calculus, with a sketch of the inventor's life. Amer. Math. Monthly, 45:414-421, 1938. (Zbl. 19, 389).
- [24] C. Berg and G. Forst. Potential Theory on Locally Compact Abelian Groups. Springer, Berlin, 1975.
- [25] L.C. Biedenharn, R.A. Gustafson, M.A. Lohe, J.D. Louck, and S.C. Milne. Special functions and group theory in theoretical physics. In Special functions: group theoretical aspects and applications, Math. Appl., pages 129–162. Reidel, Dordrecht, 1984. (MR 86h:22034).
- [26] L.C. Biedenharn, R.A. Gustafson, and S.C. Milne. An umbral calculus for polynomials characterizing U(n) tensor products. *Adv. Math.*, 51:36–90, 1984. (MR 86m:05016).
- [27] R.P. Boas. The Stieltjes moment problem for functions of bounded variation. Bull. Amer. Math. Soc., 45:399-404, 1939. (Zbl. 21, 307).

- [28] R.P. Boas and R.C. Buck. Polynomials defined by generating relations. Amer. Math. Monthly, 63:626-632, 1956. (MR 18, 300).
- [29] R.P. Boas and R.C. Buck. *Polynomial expansions of analytic functions*. Springer, Berlin, second edition, 1964. (MR 29#218).
- [30] F. Bonetti, G.-C. Rota, and D. Senato. On the foundation of combinatorial theory. X. A categorical setting for symmetric functions. *Stud. Appl. Math.*, 86:1-29, 1992. (MR 93h:05167).
- [31] K. Borsuk. Theory of shape. PWN, Warsaw, 1975.
- [32] N. Bourbaki. Eléments de mathématique. Fonctions d'une variable réelle. Herman, Paris, 1976.
- [33] W.C. Brenke. On generating functions of polynomial systems. Amer. Math. Monthly, 52:297-301, 1945. (MR 7, 64).
- [34] A. Brini. Higher dimensional recursive matrices and diagonal delta sets of series. J. Comb. Th. Ser. A, 36:315-331, 1984. (MR 86d:05013).
- [35] A. Browder. Function Algebras. Benjamin, New York, 1969.
- [36] J.W. Brown. Generalized Appell connection sequences. II. J. Math. Anal. Appl., 50:458-464, 1975. (MR 51#10721).
- [37] J.W. Brown. On multivariable Sheffer sequences. J. Math. Anal. Appl., 69:398-410, 1979. (MR 80j:05007).
- [38] J.W. Brown. A property of Steffensen sequences. Glasnik Math. Ser. III, 24 (44)(1):31-34, 1989. (MR 91a:05009).
- [39] J.W. Brown and J.L. Goldberg. A note on generalized Appell polynomials. Amer. Math. Monthly, 75:169–170, 1968. (MR 37#1662).
- [40] J.W. Brown and J.L. Goldberg. Generalized Appell connection sequences. J. Math. Anal. Appl., 46:242-248, 1974. (MR 49#7489).
- [41] J.W. Brown and S. Roman. Inverse relations for certain Sheffer sequences. SIAM J. Math. Anal., 12:186–195, 1981. (MR 82b:33015).
- [42] R.F. Brown. Elementary consequences of the noncontractibility of the circle. Amer. Math. Monthly, 81:247-252, 1974. (MR 48#9629).
- [43] N. Bruschlinsky. Stetige Abbildungen und Bettische Gruppen der Dimensionszahlen 1 und 3. Math. Ann., 103:525-537, 1934. (Zbl. 8, 373).
- [44] V.M. Bukhshtaber and A.N. Kholodov. Boas-Buck structures on sequences of polyomials. *Funct. Anal. Appl.*, 23(4):266-276, 1990. (MR 91d:26017).
- [45] R.B. Burckel. An Introduction to Classical Complex Analysis, volume 1. Academic Press, New York, 1979.

- [46] A. Calderón, F. Spitzer, and H. Widom. Inversion of Toeplitz matrices. *Illinois J. Math.*, 3:490-498, 1959. (MR 22#12386).
- [47] E.R. Canfield. Asymptotic normality in binomial type enumeration. PhD thesis, University of California, San Diego, 1975.
- [48] E.R. Canfield. Central and local limit theorems for the coefficients of polynomials of binomial type. J. Comb. Th. Ser. A, 23:275–290, 1977. (MR56#8375).
- [49] B.C. Carlson. Polynomials satisfying a binomial theorem. J. Math. Anal. Appl., 32:543-558, 1970. (MR 42#6288).
- [50] M. Cerasoli. Enumerazione binomiale e processi stocastici di Poisson composti. Bollettino U.M.I., (5) 16-A:310-315, 1979. (MR 80k:05008).
- [51] L. Cerlienco, G. Nicoletti, and F. Piras. Coalgebra and umbral calculus. Rend. Sem. Mat. Fis. Milano, 54:79-100, 1984. (MR 88k:05019).
- [52] L. Cerlienco, G. Nicoletti, and F. Piras. Polynomial sequences associated with a class of incidence coalgebras. Ann. Discr. Math., 30:159–169, 1986. (MR 88b:05018).
- [53] L. Cerlienco and F. Piras. Coalgebraic aspects of the umbral calculus. Rend. Sem. Mat. Brescia, 7:205-217, 1984. (MR 86a:05010).
- [54] L. Cerlienco and F. Piras. G-R-sequences and incidence coalgebras of posets of full binomial type. J. Math. Anal. Appl., 115:46–56, 1986. (MR 87k:05018).
- [55] Ch. A. Charalambides and A. Kyriakoussis. An asymptotic formula for the exponential polynomials and a central limit theorem for their coefficients. *Discr. Math.*, 54:259–270, 1985. (MR 86f:05009).
- [56] T.S. Chihara. Orthogonal polynomials with Brenke type generating functions. Duke Math. J., 35:505–517, 1968. (MR 37#3072).
- [57] T.S. Chihara. Orthogonality relations for a class of Brenke polynomials. Duke Math. J., 38:599–603, 1971. (MR 43#6476).
- [58] T.S. Chihara. An Introduction to Orthogonal Polynomials. Gordon and Breach, 1978.
- [59] Y. Chikuse. Multivariate Meixner classes of invariant distributions. Lin. Alg. Appl., 82, 1986.
- [60] F.M. Cholewinski. The Finite Calculus Associated with Bessel Functions, volume 75 of Contemporary Mathematics. Amer. Math. Soc., 1988. (MR 89m:05013).
- [61] J. Chover, P. Ney, and S. Wainger. Functions of probability measures. J. d'Anal. Math., 26:255–302, 1973. (MR 50#891).

- [62] K.L. Chung. A Course in Probability Theory. Academic Press, New York, 2nd edition, 1974.
- [63] J. Cigler. Some remarks on Rota's umbral calculus. Indag. Math., 40:27–42, 1978. (MR 57#12939).
- [64] P.C. Consul. A simple urn model dependent on predetermined strategy. Sankhyā, B36:391-399, 1974. (MR 54#3909).
- [65] P.C. Consul. Some new characterizations of discrete Lagrangian distributions. In G.P. Patil, S. Kotz, and J.K. Ord, editors, *Statistical distributions in scientific work*, volume 3, pages 279–290. Reidel, 1975.
- [66] P.C. Consul and S.P. Mittal. A new urn with predetermined strategy. Biom. Z., 17:67-75, 1975. (MR 52#9438).
- [67] P.C. Consul and S.P. Mittal. Some discrete multinomial probability models with predetermined strategy. *Biom. Z.*, 19:163–171, 1977. (MR 57#10875).
- [68] E.T. Copson. Asymptotic Expansions. Cambridge University Press, Cambridge, 1965.
- [69] H.H. Crapo and G.-C. Rota. On the foundations of combinatorial theory II. Combinatorial geometries. *Stud. Appl. Math.*, 49:109–133, 1970. (MR 44#3882).
- [70] H.B. Curry. Abstract differential operators and interpolation formulas. *Portugal. Math.*, 10:135–162, 1951. (MR 13, 632).
- [71] J.H. Curtiss. A note on the theory of moment generating functions. Ann. Math. Stat., 13:430-433, 1942. (MR 4, 163).
- [72] H.T. Davis. *The theory of linear operators*. Principia Press, Bloomington, Indiana, 1936. (bibliography on Appell polynomials on p. 25 etc.).
- [73] A. Di Bucchianico. Banach algebras, logarithms, and polynomials of convolution type. J. Math. Anal. Appl., 156:253-273, 1991. (MR 92d:46123).
- [74] A. Di Bucchianico. *Polynomials of convolution type*. PhD thesis, University of Groningen, The Netherlands, 1991.
- [75] A. Di Bucchianico. Representations of Sheffer polynomials. Stud. Appl. Math., 93:1-14, 1994.
- [76] A. Di Bucchianico and D.E. Loeb. Operator expansion in the derivative and multiplication by x. Integr. Transf. Spec. Fun., 4:49–68, 1996.
- [77] A. Di Bucchianico and D.E. Loeb. Natural exponential families and umbral calculus. In B. Sagan, editor, *Rota Festschrift*. Birkhäuser, 1997. to appear.

- [78] A.J. Dobson. An introduction to generalized linear models. Chapman and Hall, 1994.
- [79] H.H. Domingues. Some applications of umbral algebra to combinatorics. *Rev. Mat. Estat.*, 3:39-44, 1985. (MR 89h#05010).
- [80] P. Doubilet. On the foundations of combinatorial theory VII. Symmetric functions through the theory of distribution and occupancy. *Stud. Appl. Math.*, 51(4):377-396, 1972. (MR 55#2589).
- [81] P. Doubilet, G.-C. Rota, and R.P. Stanley. On the foundations of combinatorial theory VI. The idea of generating function. In 6th Berkeley Symp. Math. Stat. Prob. vol. 2, pages 267-318, 1972. (MR 58#16376).
- [82] P. Doubilet, G.-C. Rota, and J. Stein. On the foundations of combinatorial theory IX. On the algebra of subspaces. *Stud. Appl. Math.*, 53:185-216, 1974. (MR 58#16736).
- [83] R.G. Douglas. Banach Algebra Techniques in Operator Theory. Academic Press, New York, 1972.
- [84] J. Dugundji. Topology. Allyn and Bacon, Boston, 1966.
- [85] S. Eilenberg. Transformations continues en circonférence et la topologie du plan. *Fund. Math.*, 26:61–112, 1936. (Zbl. 13, 420).
- [86] P. Embrechts. The asymptotic behaviour of power series with positive coefficients. Academia Analecta (= Med. Konink. Akad. Weten. België), 45:41-61, 1983. (MR 85e:40001).
- [87] P. Embrechts and J. Hawkes. A limit theorem for the tails of discrete infinitely divisible laws with applications to fluctuation theory. J. Austral. Math. Soc., 32:412-422, 1982. (MR 83m:60093).
- [88] P. Feinsilver. Operator calculus. Pac. J. Math., 78:95–116, 1978. (MR 80c:60093b).
- [89] P. Feinsilver. Special functions, probability semigroups and Hamiltionian flows, volume 696 of Lect. Notes in Math. Springer, 1978. (MR 80c:60093a).
- [90] P. Feinsilver and R. Schott. Appell systems on Lie groups. J. Theoret. Probab., 5:251-281, 1992. (MR 93i:60014).
- [91] P. Feinsilver and R. Schott. Algebraic Structures and Operator Calculus. Volume I: Representation Theory. Kluwer, 1993.
- [92] W. Feller. An Introduction to Probability Theory and its Applications, volume 1. Wiley, 3rd edition, 1968.
- [93] W. Feller. An Introduction to Probability Theory and Its Applications, volume 2. Wiley, 2nd. edition, 1971.

- [94] J.P. Fillmore and S.G. Williamson. A linear algebra setting for the Rota-Mullin theory of polynomials of binomial type. *Lin. and Multilin. Alg.*, 1:67–80, 1973. (MR 47#9321b).
- [95] M.J. Fisher. The embeddability of an invertible measure. Semigroup Forum, 5:340-353, 1973. (MR 52#6321).
- [96] P. Flajolet and M. Soria. Gaussian limiting distributions for the number of components in combinatorial structures. J. Comb. Th. Ser. A, 53:165– 182, 1990. (MR 91c:05012).
- [97] P. Flajolet and M. Soria. General combinatorial schemes: Gaussian limiting distributions and exponential tails. *Discr. Math.*, 114:159–180, 1993. (MR 94e:05021).
- [98] O. Forster. Lectures on Riemann Surfaces. Springer, Berlin, 1981.
- [99] J.M. Freeman. New solutions to the Rota-Mullin problem of connection constants. Congr. Numer., 21:301-305, 1978. (MR 80c:05019).
- [100] J.M. Freeman. Orthogonality via transforms. Stud. Appl. Math., 77:119– 127, 1987. (MR 90g:42046).
- [101] T. Gamelin. Uniform Algebras. Prentice Hall, Englewood Cliffs, N.J., 1969.
- [102] A.M. Garsia and S. Joni. A new expression for umbral operators and power series inversion. Proc. Amer. Math. Soc., 64:179–185, 1977. (MR 56#2838).
- [103] A.M. Garsia and S. Joni. Higher dimensional polynomials of binomial type and formal power series inversion. *Comm. Algebra*, 6:1187-1211, 1978. (MR 58#10484).
- [104] A.M. Garsia and S. Joni. Composition sequences. Comm. Alg., 8:1195– 1266, 1980. (MR 82e:05008).
- [105] I.M. Gelfand, D. Raikov, and G.E. Shilov. Commutative Normed Rings. Chelsea, New York, 1964.
- [106] L. Giraitis and D. Surgailis. Multivariate Appell polynomials and the central limit theorem. In *Dependence in probability and statistics*, volume 11 of *Progress Prob. Stat.*, pages 21–71. Birkhäuser, 1986. (MR 89c:60024).
- [107] J.L. Goldberg. A note on polynomials generated by $a(t) \ \psi[xh(t)]$. Duke Math. J., 32:643–651, 1965. (MR 32#2628).
- [108] J.L. Goldberg. On the Sheffer A-type of polynomials generated by $a(t) \psi[xb(t)]$. Proc. Amer. Math. Soc., 17:170–173, 1966. (MR 32#4297).

- [109] J. Goldman and G.-C. Rota. On the foundations of combinatorial theory IV: Finite vector spaces and Eulerian generating functions. *Stud. Appl. Math.*, 49:239-258, 1970. (MR 45#6632).
- [110] Z. Govindarajulu and R.T. Leslie. Elementary characterizations of discrete distributions. In G.P. Patil, editor, *Random counts in scientific* work, volume 1, pages 77–96. Pennsylvania State Univ. Press, 1970.
- [111] S. Grabiner. Convergent expansions and bounded operators in the umbral calculus. Adv. Math., 72:132–167, 1988. (MR 90c:05015).
- [112] S. Grabiner. Using Banach algebras to do analysis with the umbral calculus. In *Conference on Automatic Continuity and Banach Algebras*, volume 21, pages 170–185. Proc. Centre Math. Anal. Austral. Nat. Univ, 1989. (MR 91j:46097).
- [113] A. Guinand. The umbral method: A survey of elementary mnemonic and manipulative uses. Amer. Math. Monthly, 86:187–195, 1979. (MR 80e:05001).
- [114] H. Gzyl. Interpretacion combinatorica de polinomios de tipa binomial. Acta Cient. Venezolana, 27:244-246, 1976. (MR 55#118).
- [115] H. Gzyl. Canonical transformations, umbral calculus and orthogonal theory. J. Math. Anal. Appl., 111:547–558, 1985. (MR 87e:05019).
- [116] H. Gzyl. Umbral calculus via integral transforms. J. Math. Anal. Appl., 129:315–325, 1988. (MR 89a:05022).
- [117] H. Gzyl. Hamilton Flows and Evolution Semigroups, volume 239 of Research Notes in Mathematics. Pitman, 1990.
- [118] E. Hille and R.S. Phillips. Functional Analysis and Semi-groups. Number 31 in American Mathematical Society Collected Publications. American Mathematical Society, 1957.
- [119] J. Hofbauer. Beiträge zu Rota's Theorie der Folgen von Binomialtyp. Sitzungber. Abt. II Österr. Akad. Wiss. Math. Naturw. Kl, 187:437-489, 1978. (MR 82j:05013).
- [120] S.-T. Hu. Homotopy Theory. Academic Press, New York, 1959.
- [121] W.N. Huff. The type of the polynomials generated by $f(xt) \varphi(t)$. Duke Math. J., 14:1091–1104, 1947. (MR 9, 282).
- [122] W.N. Huff and E.D. Rainville. On the Sheffer A-type of polynomials generated by $\varphi(t) f(xt)$. Proc. Amer. Math. Soc., 3:296–299, 1952. (MR 13, 841).
- [123] M.E.H. Ismail. Polynomials of binomial type and approximation theory. J. Approx. Th., 23:177–186, 1978. (MR 81a:41033).

- [124] M.E.H. Ismail and C.P. May. On a family of approximation operators. J. Math. Anal. Appl., 63:446-462, 1978. (MR 80a:41017).
- [125] K.G. Janardan. Characterizations of certain discrete distributions. In G.P. Patil, S. Kotz, and J.K. Ord, editors, *Statistical Distributions in Scientific Work*, volume 3, pages 359–364. Reidel, Dordrecht, 1975.
- [126] L. Jánossy, A. Rényi, and J. Aczél. On composed Poisson distributions I. Acta Math., 1(2-4):210-224, 1950. (MR 13, 363).
- [127] S. Johansen. An application of extreme point methods to the representation of infinitely divisible distributions. Z. Warsch. verw. Geb., 5:304-316, 1966. (MR 35#3613).
- [128] N.L. Johnson and S. Kotz. Discrete Distributions. Houghton Mifflin Company, Boston, 1969.
- [129] S.A. Joni. Multivariate exponential operators. Stud. Appl. Math., 62:175– 182, 1980. (MR 81c:41050).
- [130] S.A. Joni. Umbralized umbral operators. Eur. J. Comb., 2:41–53, 1981. (MR 82e:05015).
- [131] T. Kawata. Fourier Analysis in Probability Theory. Academic Press, New York, 1972.
- [132] A.N. Kholodov. The umbral calculus and orthogonal polynomials. Acta Appl. Math., 19:1–54, 1990. (MR 92b:33022).
- [133] A.N. Kholodov. The umbral calculus on logarithmic algebras. Acta Appl. Math., 19:55-76, 1990. (MR 91k:05014).
- [134] L.M. Koganov. Pseudogenerable two-index sequences (Russian). Nedra, Moscow, 1989.
- [135] T. Kreid. Combinatorial operators. Comment. Math. Prace Math., 29:243-249, 1990. (MR 92d:05014).
- [136] T. Kreid. Combinatorial sequences of polynomials. Comment. Math. Prace Math., 29:233-242, 1990. (MR 92h:05012).
- [137] G. Kreweras. The number of more or less 'regular' permutations. Fibonacci Quart., 18:226-229, 1980. (MR 82c:05011).
- [138] S.G. Kurbanov. Some integral representations for the exponentials of divided difference operators. Voprosy Vychisl. Prikl. Mat. (Tashkent), 80(139):105-111, 1986. (MR89h:39006).
- [139] S.G. Kurbanov and V.M. Maksimov. Mutual expansions of differential operators and divided difference operators. *Dokl. Akad. Nauk UzSSR*, 4:8–9, 1986. (MR 87k:05021).

- [140] G. Labelle. Sur l'inversion et l'itération continue des séries formelles. Eur. J. Comb., 1:113–138, 1980. (MR 82a:05003).
- [141] H.O. Lancaster. Joint probability distributions in the Meixner classes. J. Roy. Stat. Soc. B, 37:434-443, 1975. (MR 52#15770).
- [142] R. Larsen. Banach Algebras. Marcel Dekker, New York, 1973.
- [143] G. Letac. Le problème de la classification des familles exponentielles naturelles de \mathbf{R}^d ayant une fonction variance quadratique. In H. Heyer, editor, *Probability measures on groups IX*, volume 1379 of *Lect. Notes in Math.*, pages 192–216. Springer, Berlin, 1989. (MR 92a:60041).
- [144] G. Letac and M. Mora. Natural real exponential families with cubic variance functions. Ann. Stat., 18:1–37, 1990. (MR 91b:62032).
- [145] D.E. Loeb. The iterated logarithmic algebra. PhD thesis, MIT, 1989.
- [146] D.E. Loeb. Sequences of symmetric functions of binomial type. Stud. Appl. Math., 83:1-30, 1990. (MR 92e:05012).
- [147] D.E. Loeb. The iterated logarithmic algebra. Adv. Math., 86:155-234, 1991. (MR 92g:05022).
- [148] D.E. Loeb. The iterated logarithmic algebra II: Sheffer sequences. J. Math. Anal. Appl., 156:172-183, 1991. MR 92d:05013.
- [149] D.E. Loeb. Series with general exponents. J. Math. Anal. Appl., 156:184– 208, 1991. (MR 92e:05126).
- [150] D.E. Loeb and G.-C. Rota. Formal power series of logarithmic type. Adv. Math., 75:1–118, 1988. (MR 90f:05014).
- [151] E. Lukacs. Characteristic Functions. Griffin, London, 2nd. edition, 1970.
- [152] A. Lupaş. Dobinski-type formula for binomial polynomials. Studia Univ. Babeş-Bolyai Math., 33(2):40-44, 1988. (MR 90i:05009).
- [153] C. Markett, M. Rosenblum, and J. Rovnyak. A Plancherel theory for Newton spaces. Integr. Eq. Oper. Th., 9:831–862, 1986. (MR 89a:33008).
- [154] G. Markowsky. Differential operators and the theory of binomial enumeration. J. Math. Anal. Appl., 63:145-155, 1978. (MR 58#21666).
- [155] W.S. Massey. Algebraic Topology, an Introduction. Harcourt, Brace & World, New York, 1967.
- [156] C.P. May. Saturation and inverse theorems for combinations of a class of exponential-type operators. *Can. J. Math.*, 28:1224–1250, 1976. (MR 55#8640).

- [157] J. Meixner. Orthogonale Polynomsysteme mit einer besonderen Gestalt der erzeugenden Funktion ,(German) orthogonal polynomial systems with a generating function of a special form. J. London Math. Soc., 9:6-13, 1934. (Zbl. 7, 307).
- [158] P. Michor. Contributions to finite operator calculus in several variables. J. Combin. Inform. System Sci., 4:39-65, 1979. (MR 81b:05013).
- [159] G. Moldovan. Algebraic properties of a class of positive convolution operators. Studia Univ. Babes-Bolyai Math., 26:9-14, 1981. (MR 83i:41029).
- [160] C.N. Morris. Natural exponential families with quadratic variance functions. Ann. Stat., 10:65-80, 1982. (MR 83a:62037).
- [161] R.A. Morris, editor. Umbral calculus and Hopf algebras, volume 6 of Contemporary Mathematics. Amer. Math. Soc., 1982. (MR 83a:05001).
- [162] R. Mullin and G.-C. Rota. On the foundations of combinatorial theory III. Theory of binomial enumeration. In Harris, editor, *Graph theory and its applications*, pages 167–213. Academic Press, 1970. (MR 43#65).
- [163] W. Nichols and M.E. Sweedler. Hopf algebras and combinatorics. In Umbral calculus and Hopf algebras, volume 6 of Contemporary Mathematics, pages 49-84. Amer. Math. Soc., Providence, 1982. (MR 83g:16019).
- [164] H. Niederhausen. Methoden zur Berechnung exakter Verteilungen vom Kolmogorov-Smirnov Typ. Technical Report 99, Technical Univ. Graz, Austria, 1978.
- [165] H. Niederhausen. Linear recurrences under side conditions. Eur. J. Combin., 1:353-368, 1980. (MR 83c:05009).
- [166] H. Niederhausen. Sheffer polynomials and linear recurrences. Congr. Num., 29:689-698, 1980. (MR 82m:05012).
- [167] H. Niederhausen. Sheffer polynomials in path enumeration. Congr. Num., 26:281-294, 1980. (MR 82d:05015).
- [168] H. Niederhausen. Sheffer polynomials for computing exact Kolmogorov-Smirnov and Rényi type distributions. Ann. Statist., 9:923-944, 1981. (MR 84b:62067).
- [169] H. Niederhausen. How many paths cross at least l given lattice points. Congr. Num., 36:161–173, 1982. (MR 85b:05014).
- [170] H. Niederhausen. Sheffer polynomials for computing Takács's goodnessof-fit distributions. Ann. Statist., 11:600–606, 1983. (MR 84h:62077).
- [171] H. Niederhausen. A formula for explicit solutions of certain linear recursions on polynomial sequences. Congr. Num., 49:87–98, 1985. (MR 87j:11018).

- [172] I Niven. Formal power series. Amer. Math. Monthly, 76:871–889, 1969.
 (MR 40#5606).
- [173] G Olive. Some functions that count. Austral. Math. Soc. Gaz., 10(1):2–13, 1983. (MR 85f:05004).
- [174] G. Olive. Catalan numbers revisited. J. Math. Anal. Appl., 111:201-235, 1985. (MR 87e:05011).
- [175] G. Olive. The ballot problem revisited. Stud. Appl. Math., 78:21–30, 1988. (MR 90f:05014).
- [176] F.J. Palas. The polynomials generated by f(t) exp[p(x)u(t)]. PhD thesis, Oklahoma University, 1955.
- [177] C. Parrish. Multivariate umbral calculus. PhD thesis, University of California at San Diego, La Jolla, 1974.
- [178] C. Parrish. Multivariate umbral calculus. J. Linear and Multilinear Algebra, 6:93-109, 1978. (MR 58#10487).
- [179] S. Pincherle. Mémoire sur le calcul fonctionnel distributif. Math. Ann., 49:325–382, 1897.
- [180] S. Pincherle and U. Amaldi. Le operazioni distributive e le loro applicazioni all'analisi. N. Zanichelli, Bologna, 1901.
- [181] G. Pólya. Sur l'indétermination d'un problème voisin du problème des moments. C.R. Acad. Sci. Paris, 207:708-711, 1938. (Zbl. 20, 42).
- [182] T. Popoviciu. Remarques sur les polynomes binomiaux. Bul. Soc. Şti. Cluj, 6:146-148, 1931. Zbl. 2, 398.
- [183] T.R. Prabhakar, M. Chopra, and S. Gupta. An Appell cross-sequence suggested by Hermite polynomials. *Indian J. Pure Appl. Math.*, 9:194– 199, 1978. (MR 57#16746).
- [184] E. Rainville. Special Functions. Chelsea, New York, 1960.
- [185] E.D. Rainville. Some symbolic relations among classical polynomials. Amer. Math. Monthly, 53:299–305, 1946. (MR 7, 440).
- [186] R. Rasala. The Rodrigues formula and polynomial differential operators. J. Math. Anal. Appl., 84:443-482, 1981. (MR 83g:33009).
- [187] N. Ray. Extensions of umbral calculus, penumbral coalgebras and generalized Bernoulli numbers. Adv. Math., 61:49-100, 1986. (MR 88b:05019).
- [188] N. Ray. Symbolic calculus: a 19th century approach to MU and BP. In Homotopy theory (Durham 1985), volume 117 of London Math. Soc. Lect. Notes Series, pages 195–238. Cambridge University Press, 1987. (MR 89k:55007).

- [189] N. Ray. Umbral calculus, binomial enumeration and chromatic polynomials. Trans. Amer. Math. Soc., 309:191-213, 1988. (MR 89k:05014).
- [190] N. Ray. Loops on the 3-sphere and umbral calculus, volume 96 of Cont. Math., pages 297-302. Amer.Math. Soc., 1989. (MR 90i:55006).
- [191] N. Ray. Stirling and Bernoulli numbers for complex oriented homology theory, volume 1370 of Lect. Notes in Math., pages 362–373. Springer, Berlin, 1989. (MR 90f:55010).
- [192] N. Ray. Tutte algebras of graphs and formal group theory. Proc. London Math. Soc., 65:23-45, 1992. (MR 93f:05042).
- [193] N. Ray and C. Wright. Colourings and partition types: a generalised chromatic polynomial. Ars Combin., 25 B:277-286, 1988. (MR 89e:05092).
- [194] M. Razpet. An application of the umbral calculus. J. Math. Anal. Appl., 149:1-16, 1990. (MR 91i:05018).
- [195] M. Razpet. A new class of polynomials with applications. J. Math. Anal. Appl., 150:85-99, 1990. (MR 91i:05020).
- [196] D.L. Reiner. Multivariate sequences of binomial type. Stud. Appl. Math., 57 (2):119–133, 1977. (MR 58#21668).
- [197] D.L. Reiner. Sequences of polynomials of fractional binomial type. Lin. Multilin. Alg., 5:175-179, 1977. (MR 56#11809).
- [198] D.L. Reiner. The combinatorics of polynomial sequences. Stud. Appl. Math., 58:95-117, 1978. (MR 58#260).
- [199] J. Riordan. An Introduction to Combinatorial Analysis. Wiley, New York, 1958.
- [200] J. Riordan. Combinatorial Identities. Wiley, New York, 1968.
- [201] S.M. Roman. The theory of umbral calculus I. J. Math. Anal. Appl., 87:58-115, 1982. (MR 84c:05008a).
- [202] S.M. Roman. The Umbral Calculus. Academic Press, 1984. (MR 87c:05015 = Zbl. 536.33001).
- [203] S.M. Roman. More on the umbral calculus, with emphasis on the q-umbral calculus. J. Math. Anal. Appl., 107:222-254, 1985. (MR 86h:05024).
- [204] S.M. Roman. The harmonic logarithms and the binomial formula. J. Combin. Theory, 63:143-163, 1993. (MR 94i:05007).
- [205] S.M. Roman, P.N. De Land, R.C. Shiflett, and H.S. Schultz. Inverse relations and the umbral calculus. J. Comb. Inf. Syst. Sci., 8:185–198, 1983. (MR 86h:05023).

- [206] S.M. Roman, P.N. De Land, R.C. Shiflett, and H.S. Schultz. The umbral calculus and the solution to certain recurrence relations. J. Comb. Inf. Syst. Sci., 8:235-240, 1983. (MR 87b:05026).
- [207] S.M. Roman and G.-C. Rota. The umbral calculus. Adv. Math., 27:95– 188, 1978. (MR 58#5256).
- [208] G.-C. Rota. The number of partitions of a set. Amer. Math. Monthly, 71:498-504, 1964. (MR 28#5009).
- [209] G.-C. Rota. Finite operator calculus. Academic Press, New York, 1975. (MR 53#7796).
- [210] G.-C. Rota, D. Kahaner, and A. Odlyzko. On the foundations of combinatorial theory VII. Finite operator calculus. J. Math. Anal. Appl., 42:684-760, 1973. (MR 49#10556).
- [211] H. Royden. Function algebras. Bull. Amer. Math. Soc, 69:281–298, 1963. (MR 26, 6817).
- [212] W. Rudin. Functional Analysis. McGraw-Hill, 1973.
- [213] W. Rudin. Real and Complex Analysis. McGraw Hill, 2nd. edition, 1974.
- [214] V.N. Sačkov. Probabilistic Methods in Combinatorial Analysis (in Russian). Nauka, Moscow, 1978. MR 80g:05002.
- [215] C. Scaravelli. Su i polinomi di Appell. Riv. Mat. Univ. Parma, 6 (2):103– 116, 1965. (MR 36#440).
- [216] E. Schmutz. Asymptotic expansions of the coefficients of $e^{P(z)}$. Bull. London Math. Soc., 21:482–486, 1989. (MR 90j:41060).
- [217] I.M. Sheffer. A simplified solution of the equation $\Delta y(x) = f(x)$. Bull. Amer. Math. Soc., 43:283–287, 1937. (Zbl. 16, 307).
- [218] I.M. Sheffer. Note on Appell polynomials. Bull. Amer. Math. Soc., 51:739-744, 1945. (MR 7, 64).
- [219] E.S.W. Shiu. Proofs of central-difference interpolation formulas. J. Approx. Theory, 35:177–180, 1982. (MR 84i:41004).
- [220] E.S.W. Shiu. Steffensen's poweroids. Scand. Actuar. J., 2:123–128, 1982. (MR 83m:62167).
- [221] S.D. Silvey. Statistical inference, volume 7 of Monographs on Statistics and Applied Probability. Chapman and Hall, 1991.
- [222] A J. Stam. Polynomials of binomial type and renewal sequences. Stud. Appl. Math., 77:183-193, 1987. (MR 90m:60097).
- [223] A J. Stam. Two identities in the theory of polynomials of binomial type. J. Math. Anal. Appl., 122:439-443, 1987. (MR 88b:05015).
- [224] A J Stam. Polynomials of binomial type and compound Poisson processes. J. Math. Anal. Appl., 130:493-508, 1988. (MR 89d:60134).
- [225] A J. Stam. Lagrange's theorem, polynomials of convolution type and probability distributions. Technical Report W-9011, University of Groningen, September 1990.
- [226] J. F. Steffensen. On a special type of polynomials. Mat. Tidsskr. B., pages 6-9, 1950. (MR 12, 409).
- [227] J.F Steffensen. The poweroid, an extension of the mathematical notion of power. Acta Math., 73:333-366, 1941. (MR 3, 326).
- [228] J.F. Steffensen. On a class of polynomials. Mat. Tidsskr. B, 1945:10–14, 1945. (MR 7, 157).
- [229] J.F. Steffensen. On the polynomials $R_{\nu}^{[\lambda]}(x)$, $N_{\nu}^{[\lambda]}(x)$ and $M_{\nu}^{[\lambda]}(x)$. Acta Math., 78:291–314, 1946. (MR 8, 155).
- [230] F.W. Steutel. Preservation of Infinite Divisibility under Mixing, and Related Topics, volume 33 of Math. Centre Tracts. Mathematical Centre, Amsterdam, 1970.
- [231] F.W. Steutel. On the zeros of infinitely divisible densities. Ann. Math. Stat., 42:812-815, 1971. (Zbl. 218.60026).
- [232] F.W. Steutel. A Lagrange type identity related to random walk theory. Technical report, Technical University Twente, 1972.
- [233] F.W. Steutel. Some recent results in infinite divisibility. Stoch. Proc. Appl., 1:125-143, 1973. (MR 51#9152).
- [234] E.G. Tashes. Application of the Liouville-Steklov method to orthogonal Appell polynomials in two variables. In Application of functional analysis in approximation theory, pages 76–82. Gos. univ. Kalinin, 1988. (MR 90b:33026).
- [235] J.L. Teugels. Bivariate sequences. Bull. Soc. Math. Belg. Sér. B, 42:1–30, 1990. (MR 91h:05017).
- [236] C.J. Thorne. A property of Appell sets. Amer. Math. Monthly, 52:191– 193, 1945. (MR 6, 217).
- [237] F. Topsøe. Banach algebra methods in prediction theory. Manuscripta Math., 23:19–55, 1977. (MR 57#10800).
- [238] K. Ueno. Umbral calculus and special functions. Adv Math., 67:174-229, 1988. (MR 88m:05012).
- [239] K. Ueno. General power umbral calculus in several variables. J. Pure Appl. Algebra, 59:299–308, 1989. (MR 90m:05014).

- [240] K. Ueno. Hypergeometric series formulas through operator calculus. Funkcialaj Ekvacioj, 33:493–518, 1990. (MR 92b:33004).
- [241] Univ. "Babeş-Bolyai". C. Manole, Approximation operators of binomial type, number 9 in Seminar on Numerical and Statistical Calculus, Cluj-Napoca, 1987, pages 93–98. (MR 89c:00027).
- [242] O.V. Viskov. Operator characterization of generalized Appell polynomials. Sov. Math. Dokl., 16:1521–1524, 1975. (MR 52#14416).
- [243] O.V. Viskov. On bases in the space of polynomials. Sov. Math. Dokl., 19:250-253, 1978. (MR 58#10854).
- [244] O.V. Viskov. On a class of linear operators. In V.S. Vladimirov, editor, Generalized functions and their applications in mathematical physics, Moscow, pages 110-120, 1980. (Zbl. 518.34025).
- [245] C.T.C. Wall. A Geometric Introduction to Topology. Addison Wesley, Reading, Massachusetts, 1972.
- [246] M. Ward. A calculus of sequences. Amer. J. Math., 58:255-266, 1936. (Zbl. 14, 56).
- [247] T. Watanabe. On a dual relation for addition formulas of additive groups: I. Nagoya Math. J., 94:171-191, 1984. (MR 86f:05020).
- [248] T. Watanabe. On a dual relation for addition formulas of additive groups: II. Nagoya Math. J., 97:95-135, 1985. (MR 86i:05023).
- [249] T. Watanabe. On a generalization of polynomials in the ballot problem.
 J. Statist. Planning & Inference, 14:143-152, 1986. (MR 87j:05024).
- [250] T. Watanabe. On a determinant sequence in the lattice path counting. J. Math. Anal. Appl., 123:401-414, 1987. (MR 88g:05015).
- [251] J.H. Williamson, editor. Algebras in Analysis. Academic Press, New York, 1975.
- [252] B.G Wilson and F.J. Rogers. Umbral calculus and the theory of multispecies nonideal gases. *Phys. A*, 139:359–386, 1986. (MR 88d:82024).
- [253] K.W. Yang. Integration in the umbral calculus. J. Math. Anal. Appl., 74:200-211, 1980. (MR 82i:05008).

Index

 $\mathcal{A}, 91$ $\mathcal{A}(a,b), 105$ $\overline{\mathcal{A}}(a,b), 105$ Abel operator, 19, 23, 30 polynomials, 23, 53, 57, 115 actuarial polynomials, 46 algebra Banach, 82 Wiener, 94 algebraic topology, 9 analysis harmonic, 21 Appell polynomials, 36 approximation operator, 57 theory, 9, 77 $\mathcal{A}_r, 91$ Arens-Royden Theorem, 87 Argument Principle, 96, 127 backward difference operator, 19, 23, 30 Banach algebra, 82 Banach-Stone Theorem, 95 basic sequence, 22 Bernoulli operator, 19, 20 polynomials, 36, 42, 46 Bernstein function, 65 Bessel function of the first kind, 103 polynomials, 54 binomial distribution, 55, 59, 79 formula, 24 Abel generalization, 24, 54

type, 7 Blissard Calculus, 9, 31 Borel measure invertible, 68 Brownian semigroup, 70 $\mathcal{C}(\mathcal{K}), 89$ Calculus Blissard, 31 Symbolic, 31 Umbral, 19 Catalan numbers, 54 coefficient connection, 33 sequence, 17 combinatorics, 9 complex homomorphism, 82 component, 83, 84 compound geometric distribution, 117 Poisson distribution, 117 conditioning, 14 connected topological space, 83 connection coefficients, 33 contractible, 89 contraction, 89 convolution of sequences, 13 operator, 12 semigroup, 64 weakly continuous, 64 weakly measurable, 64 type function, 13 two-sided sequence, 102 cross sequence, 43 cumulant, 74

 $\mathcal{D}, 87$ $\overline{\mathcal{D}}, 87$ degree of polynomial, 12 delta operator, 20 derangement polynomials, 125 disc algebra, 91 Discrete Renewal Theorem, 120 distribution binomial, 55, 59, 79 compound geometric, 117 compound Poisson, 117 gamma, 79 hyperbolic, 79 hypergeometric, 55, 59 negative binomial, 79 normal, 42, 79 Poisson, 75 Pólya-Eggenberger, 55, 59 quasi-binomial, 56, 60 uniform, 42 Dobinski Formula, 69 duplication formula, 34

 E^a , 19 Euler polynomials, 46 Expansion Theorem Operator, 25, 40 Polynomial, 25, 40 exp \mathcal{B} , 83 exponential family natural, 74 polynomials, 69

 φ , 103 factorial lower, 23 upper, 23, 51, 54 Favard's Theorem, 60 Formula Binomial, 24 Dobinski, 69 duplication, 34 Newton Interpolation, 26 Rodrigues, 29

for Sheffer polynomials, 41 Taylor, 20 Vandermonde, 24 forward difference operator, 19, 23, 30 function Bernstein, 65 Bessel, 103 convolution type, 13 moment generating, 42 negative definite, 65 functional on \mathcal{P} , 26 $\mathcal{G}_1, 84$ gamma distribution, 79 Gamma-semigroup, 69 Gelfand topology, 83, 85 transform, 85 generator infinitesimal, 44 Gould polynomials, 53, 54, 57 group first Čech cohomology group, 86 first cohomotopy, 86 strongly continuous, 68 harmonic analysis, 21 Hermite polynomials, 36, 42, 46, 79 homomorphism complex, 82 homotopy, 86 hyperbolic distribution, 79 hypergeometric distribution, 55, 59 identity, 82 ind, 93 index, 93 infinite divisibility, 68, 81 infinitesimal generator, 44 inv \mathcal{B} , 82 invertible Borel measure, 68 operator, 38 Isomorphism Theorem, 27

146

Krawtchouk polynomials, 79 $\mathcal{L}(a, b), 107$ $\ell_1(\alpha), 88$ Lagrange inversion, 51 Laguerre operator, 19, 20, 30, 71 polynomials, 30, 36, 47 Lévy-measure, 127 link function, 75 lower factorial, 23 $\mathcal{M}, 83$ maximal ideal space, 83 Maximum Modulus Theorem, 99 measure invertible Borel, 68 Lévy, 127 Mittag-Leffler polynomials, 51, 126 moment generating function, 42 system, 66 natural exponential family, 74 associated Sheffer polynomials, 76 negative binomial distribution, 79 negative definite function, 65 Newton Interpolation Formula, 26 normal distribution, 42, 79 $\mathcal{N}_t, 101$ $\nu_t, 101$ operator Abel, 19, 23, 30 approximation, 57 backward difference, 19, 23, 30 Bernoulli, 19, 20 delta, 20 forward difference, 19, 23, 30 invertible, 38 Laguerre, 19, 20, 30, 71 shift, 19 shift-invariant, 19 Toeplitz, 26 umbral, 31, 64, 68 Weierstrass, 71

Operator Expansion Theorem, 25, 40 orthogonal Sheffer sequence, 60 $\mathcal{P}, 12$ path, 84 Pincherle derivative, 28, 61 Poisson distribution, 75, 117 process, 12, 13 semigroup, 69 Poisson-Charlier polynomials, 46, 79 Pólya-Eggenberger distribution, 55, 59Polynomial Expansion Theorem, 25, 40 polynomials Abel, 23, 53, 57, 115 actuarial, 46 Appell, 36 Bernoulli, 36, 42, 46 Bessel, 54 binomial type, 7 convolution type, 13 generating function, 17 cross, 43 degree, 12 derangement, 125 Euler, 46 exponential, 69 Gould, 53, 54, 57 Hermite, 36, 42, 46, 79 Krawtchouk, 79 Laguerre, 30, 36, 47 Mittag-Leffler, 51, 126 Poisson-Charlier, 46, 79 set of type zero, 8 Sheffer, 35 Wick, 41 ψ , 96 quantum mechanics, 41 quasi-binomial distribution, 56, 60 renewal theory, 116 renewal property, 51 $\mathcal{R}_g, 101$

Rodrigues Formula, 29 for Sheffer polynomials, 41 Roman shift, 25, 30 $\rho_t, 101$ semigroup Brownian, 70 Gamma, 69 of operators, 43 Poisson, 69 sequence basic, 22 coefficient, 17 cross, 43Sheffer, 35 Steffensen, 46 Sheffer polynomials, 35 sequence, 35 orthogonal, 60 strict sense, 35 wide sense, 35 shift operator, 19 shift-invariant operator, 19 statistics, 9 Steffensen sequence, 46 Stirling numbers first kind, 53 second kind, 53, 58 signless, 53 strongly continuous group, 68 subordination, 68 subspace translation-invariant, 21 Symbolic Calculus, 31

T, 92

TA, 99
Taylor's Formula, 20
Theorem
Arens-Royden, 87
Banach-Stone, 95
Borsuk, 91
Discrete Renewal, 120
Favard, 60
Isomorphism, 27

Maximum Modulus, 99 noncentral limit, 41 Operator Expansion, 25, 40 Polynomial Expansion, 25, 40 three-term recurrence relation, 60 Toeplitz operator, 26 topology algebraic, 9 Gelfand, 85 transition probability, 12 translation-invariant subspace, 21 u, 82 $U_0, \, 113$ $\mathcal{U}_L,\,113$ umbral calculus, 9, 19 electronic survey, 9 operator, 31, 64, 68 uniform distribution, 42 unit element, 82 upper factorial, 23, 51, 54 Vandermonde Formula, 24

 $\mathcal{W}, 94$ $\mathcal{W}(a,b), 105$ Weierstrass operator, 71 Wick polynomials, 41 Wiener algebra, 94

148

CWI TRACTS

- 1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984.
- 2 J.J. Dijkstra. Fake topological Hilbert spaces and characterizations of dimension in terms of negligibility. 1984.
- 3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.
- 4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.
- 5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.
- 6 A.P.W. Böhm. Dataflow computation. 1984
- 7 A. Blokhuis. Few-distance sets. 1984.
- 8 M.H. van Hoorn. Algorithms and approximations for queueing systems. 1984.
- 9 C.P.J. Koymans. *Models of the lambda calculus.* 1984.
- 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.
- N.M. van Dijk. Controlled Markov processes; timediscretization. 1984.
- 12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.
- 13 D. Grune. On the design of ALEPH. 1985.
- 14 J.G.F. Thiemann. Analytic spaces and dynamic programming: a measure theoretic approach. 1985.
- 15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.
- 16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral operators. 1985.
- 17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.
- 18 A.D.M. Kester. Some large deviation results in statistics. 1985.
- 19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.
- 20 B.F. Schriever. Order dependence. 1986.
- 21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.
- 22 J.C.S.P. van der Woude. Topological dynamix. 1986.
- 23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.
- 24 J.C.M. Baeten. Filters and ultrafilters over definable subsets of admissible ordinals. 1986.
- 25 A.W.J. Kolen. Tree network and planar rectilinear location theory. 1986.
- 26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.
- 27 A.J.M. van Engelen. Homogeneous zerodimensional absolute Borel sets. 1986.
- 28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.
- 29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.
- 30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

- 31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987.
- 32 J.H.J. Einmahl. Multivariate empirical processes. 1987.
 33 O.J. Vrieze. Stochastic games with finite state and
- action spaces. 1987. 34 P.H M. Kersten. Infinitesimal symmetries: a com-
- putational approach. 1987. 35 M.L. Eaton. Lectures on topics in probability inequalities. 1987.
- 36 A.H.P. van der Burgh, R.M.M. Mattheij (eds.). Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.
- 37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.
- 38 J.B.G. Frenk. On Banach algebras, renewal measures and regenerative processes. 1987.
- 39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.
- 40 J.L. Geluk, L. de Haan. Regular variation, extensions and Tauberian theorems. 1987.
- 41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.
- 42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, formalisms, and tools. 1987.
- 43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.
- 44 A.W. van der Vaart. Statistical estimation in large parameter spaces. 1988.
- 45 S.A. van de Geer. Regression analysis and empirical processes. 1988.
- 46 S.P. Spekreijse. Multigrid solution of the steady Euler equations. 1988.
- 47 J.B. Dijkstra. Analysis of means in some nonstandard situations. 1988.
- 46 F.C. Drost. Asymptotics for generalized chi-square goodness-of-fit tests. 1988.
 49 F.W. Wubs. Numerical solution of the shallow-
- water equations. 1988.
- 50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.
- 51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.
- 52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.
- 53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic programming in function space. 1988.
- 54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.
- 55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.
- 56 J.C. Ebergen. Translating programs into delayinsensitive circuits. 1989.
- 57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.
- 58 M.C.M. de Gunst. A random model for plant cell population growth. 1989.
- 59 D. van Dulst. Characterizations of Banach spaces not containing l¹. 1989.
- 60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.

- 61 P. de Jong. Central limit theorems for generalized multilinear forms. 1989.
- 62 V.J. de Jong. A specification system for statistical software. 1989.
- 63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.
- 64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1080
- 65 B.M.M. de Weger. Algorithms for diophantine equations, 1989.
- 66 A. Jung. Cartesian closed categories of domains. 1989
- 67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?, 1989.
- 68 H.J. Woerdeman. Matrix and operator extensions. 1989.
- 69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989. 70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.).
- Twenty-five years of operations research in the Netherlands: Papers dedicated to Gijs de Leve. 1990
- 71 P.J.C. Spreij. Counting process systems. laentification and stochastic realization. 1990.
- 72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990.
- 73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.
- 74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991
- 75 M.W.P. Savelsbergh. Computer aided routing 1992.
- 76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991. 77 A.J. van Es. Aspects of nonparametric density es-
- timation. 1991. 78 G.A.P. Kindervater. Exercises in parallel combina-
- torial computing. 1992. 79 J.J. Lodder. Towards a symmetrical theory of gen-
- eralized functions. 1991. 80 S.A. Smulders. Control of freeway traffic flow.
- 1996 81 P.H.M. America, J.J.M.M. Rutten. A parallel
- object-oriented language: design and semantic foundations. 1992. 82
- F. Thuijsman. Optimality and equilibria in stochastic games. 1992. 83 R.J. Kooman. Convergence properties of recurrence
- sequences. 1992 84 A.M. Cohen (ed.). Computational aspects of Lie
- group representations and related topics. Proceedings of the 1990 Computational Algebra Seminar at CWI, Amsterdam. 1991. V. de Valk. One-dependent processes. 1994.
- J.A. Baars, J.A.M. de Groot. On topological and 86 linear equivalence of certain function spaces. 1992. 87 A.F. Monna. The way of mathematics and mathe-
- maticians, 1992. E.D. de Goede. Numerical methods for the three-
- dimensional shallow water equations. 1993. 89 M. Zwaan. Moment problems in Hilbert space with
- applications to magnetic resonance imaging. 1993. 90 C. Vuik. The solution of a one-dimensional Stefan
- problem. 1993. 91 E.R. Verheul. Multimedians in metric and normed
- spaces. 1993. 92 J.L.M. Maubach. Iterative methods for non-linear partial differential equations, 1994.

- 93 A.W. Ambergen. Statistical uncertainties in posterior probabilities. 1993.
- Moving-grid methods for time-P.A. Zegeling. dependent partial differential equations. 1993.
- M.J.C. van Pul. Statistical analysis of software re-95 liability models. 1993.
- 96 J.K. Scholma. A Lie algebraic study of some integrable systems associated with root systems. 1993. J.L. van den Berg. Sojourn times in feedback and
- processor sharing queues. 1993.
- 98 A.J. Koning. Stochastic integrals and goodness-offit tests. 1993.
- 99 B.P. Sommeijer. Parallelism in the numerical integration of initial value problems. 1993.
- 100 J. Molenaar. Multigrid methods for semiconductor device simulation. 1993.
- 101 H.J.C. Huijberts. Dynamic feedback in nonlinear synthesis problems. 1994.
- 102 J.A.M. van der Weide. Stochastic processes and point processes of excursions. 1994
- 103 P.W. Hemker, P. Wesseling (eds.). Contributions to multigrid. 1994.
- 104 I.J.B.F. Adan. A compensation approach for queueing problems. 1994.
- 105 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 1. 1994.
- 106 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 2. 1994.
- 107 R.A. Trompert. Local uniform grid refinement for time-dependent partial differential equations. 1995.
- 108 M.N.M. van Lieshout. Stochastic geometry models in image analysis and spatial statistics. 1995.
- 109 R.J. van Glabbeek. Comparative concurrency semantics and refinement of actions. 1996.
- 110 W. Vervaat, H. Holwerda (ed.). Probability and lattices. 1997.
- 111 I. Helsloot. Covariant formal group theory and some applications, 1995.
- 112 R.N. Bol. Loop checking in logic programming. 1995
- 113 G.J.M. Koole. Stochastic scheduling and dynamic programming. 1995. 114 M.J. van der Laan. Efficient and inefficient estima-
- tion in semiparametric models. 1995.
- 115 S.C. Borst. Polling models. 1996. 116 G.D. Otten. Statistical test limits in quality control.
- 1996. 117 K.G. Langendoen. Graph reduction on shared-
- memory multiprocessors. 1996. 118 W.C.A. Maas. Nonlinear \mathcal{H}_{∞} control: the singular
- case. 1996. 119 A. Di Bucchianico. Probabilistic and analytical as-
- pects of the umbral calculus. 1997. 120 M. van Loon. Numerical methods in smog predic-
- tion. 1997. 121 B.J. Wijers. Nonparametric estimation for a win-
- dowed line-segment process. 1997.
- 122 W.K. Klein Haneveld, O.J. Vrieze, L.C.M. Kallenberg (editors). Ten years LNMB - Ph.D. research and graduate courses of the Dutch Network of Operations Research. 1997
- 123 R.W. van den Hofstad. One-dimensional random polymers. 1998.
- 124 W.J.H. Stortelder. Parameter estimation in nonlinear dynamical systems. 1998.
- 125 M.H. Wegkamp. Entropy methods in statistical estimation, 1998.

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963.

2 A.R. Bloemena. Sampling from a graph. 1964.

3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

part 1: model and melnoa. 1964.
4 G. de Leve. Generalized Markovian decision processes, part 11: probabilistic background. 1964.
5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.
6 M.A. Maurice. Compact ordered spaces. 1964.
7 W.R. van Zwet. Convex transformations of random variables. 1964.

8 J.A. Zonneveld. Automatic numerical integration. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964. 11 A.B. Paalman-de Miranda. Topological semigroups. 1964.

12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965.

13 H.A. Lauwerier. *Asymptotic expansions.* 1966, out of print: replaced by MCT 54.

14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

R. Doornbos. Slippage tests. 1966.
 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60.
 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1, 1968.

23 T.J. Decker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968. 24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part 1. 1968

27 European Meeting 1968. Selected statistical papers, part II. 1968

28 J. Oosterhoff. Combination of one-sided statistical tests. 1969.

29 J. Verhoeff. Error detecting decimal codes. 1969.

30 H. Brandt Corstius. Exercises in computational linguistics. 1970.

31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.

32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.

33 F.W. Steutel. Preservations of infinite divisibility under mix-

I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971.

36 J. Grasman. On the birth of boundary layers. 1971.

37 J.W. de Bakker, G.A. Blaauw, A.J.W. Duijvestijn, E.W. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium. 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972. 42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

44 H. Bart. Meromorphic operator valued functions. 1973.

45 A.A. Balkema. Monotone transformations and limit laws. 1973.

46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.

47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973.

48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

49 H. Kok, Connected orderable spaces. 1974.
50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A.
Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G.
Fisker (eds.), Revised report on the algorithmic language
ALGOL 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory. 1974.

52 P.C. Baayen (ed.). Topological structures. 1974.

53 M.J. Faber. Metrizability in generalized ordered spaces.

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

54 H.A. Lauwerter. Asymptotic analysis, part 1. 1974.
55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974.
56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry. 1974.
57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.
58 W Albers Asymptotic expansions and the deficiency con-

58 W. Albers. Asymptotic expansions and the deficiency concept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975.

63 J.W. de Bakker (ed.). Foundations of computer science. 1975.

64 W.J. de Schipper. Symmetric closed categories. 1975. 65 J. de Vries. Topological transformation groups, 1: a categor-ical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977. 70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

and proof theory, 1910. 71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976. 72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveenides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979.

74 H.J.J. te Riele. A theoretical and computational study of generalized aliquo sequences. 1976. 75 A.E. Brouwer. Treelike spaces and related connected topo-logical spaces. 1977.

76 M. Rem. Associons and the closure statements. 1976.

77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.

79 M.C.A. van Zuijlen. Empirical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary problems. 1977.

81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976.

82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

Science 11, part 2, 1910.
83 L.S. van Benthem Jutting, Checking Landau's "Grandlagen" in the AUTOMATH system. 1979.
84 H.L.L. Busard, The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii:xii. 1977.

85 J. van Mill. Supercompactness and Wallmann spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977.

89 J.W. de Roever. Complex Fourier transformation and analytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen. Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteris-tic. 1979.

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977.

94 A. Bijlsma. Simultaneous approximations in transcendental number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978.

96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978.

(eds.). Interfaces between computer science and operations research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 1, 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces. 1978

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978.

104 J.M. van Wouwe. GO-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science 111, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part I: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part II: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random' walks with stationary increments and renewai theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

proviems with restricted alternatives. 1979. 114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979. 115 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 1. 1979.

116 P.C. Baayen, J. van Mill (eds.). Topological structures 11, part 2. 1979.

117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980. 120 W.P.M. de Ruyter. On the asymptotic analysis of large-scale ocean circulation. 1980.

121 W.H. Haemers. Eigenvalue techniques in design and graph theory. 1980.

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980.

123 I. Yuhász. Cardinal functions in topology - ten years later. 1980

124 R.D. Gill. Censoring and stochastic integrals. 1980.

125 R. Eising. 2-D systems, an algebraic approach. 1980.

126 G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980.

127 J.W. Klop. Combinatory reduction systems. 1980.

128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures*. 1980.

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.
 132 H.M. Mulder. The interval function of a graph. 1980.

133 C.A.J. Klaassen. Statistical performance of location esti-mators. 1981.

134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

tional conference on ALGOL 08. 1981. 135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.), Formal methods in the study of language, part I. 1981. 136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.), Formal methods in the study of language, part II. 1981.

137 J. Telgen. Redundancy and linear programs. 1981.

138 H.A. Lauwerier. Mathematical models of epidemics. 1981. 139 J. van der Wal. Stochastic dynamic programming, succes-sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis.

ехспа 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981. 142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order

structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms. 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1982.

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a sympo-sium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes 1982

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequencies. 1982.

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired ser-vices. 1982.

154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.

155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983. 160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.

162 J.J. Dik. Tests for preference. 1983.

163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983. 164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implemen-tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.